## Hochschild Cohomology, Modular Tensor Categories, and Mapping Class Groups

I. General Theory

Simon Lentner, Svea Nora Mierach, Christoph Schweigert, Yorck Sommerhäuser

#### Abstract

Given a finite modular tensor category, we associate with each compact surface with boundary a cochain complex in such a way that the mapping class group of the surface acts projectively on its cohomology groups. In degree zero, this action coincides with the known projective action of the mapping class group on the space of chiral conformal blocks. In the case that the surface is a torus and the category is the representation category of a factorizable ribbon Hopf algebra, we recover our previous result on the projective action of the modular group on the Hochschild cohomology groups of the Hopf algebra.

# Contents

Introduction						
1	Mapping class groups					
	1.1	The classification of surfaces	6			
	1.2	The fundamental group	9			
	1.3	Mapping class groups	12			
	1.4	Dehn twists	14			
	1.5	Braidings	15			
	1.6	The action on the fundamental group	17			
	1.7	Dehn twists for special curves	20			
	1.8	Dehn twists related to two boundary components	23			
	1.9	The capping homomorphism	24			
	1.10	The Birman sequence	25			
	1.11	Singular homology	27			
	1.12	The mapping class group of the torus	29			
	1.13	The mapping class group of the sphere	32			
<b>2</b>	Tensor categories					
	2.1	Finiteness	34			
	2.2	Factorizable Hopf algebras	37			
	2.3	Coends	38			
	2.4	Coends from Hopf algebras	41			
	2.5	The block spaces	43			

	2.6	Mapping class group representations	47		
	2.7	Modular functors	52		
	2.8	The case of the torus	54		
3	3 Derived functors				
	3.1	Projective resolutions	57		
	3.2	Derived block spaces	59		
	3.3	The case of the sphere	67		
	3.4	Hochschild cohomology	73		

### Introduction

It is by now well-understood that a semisimple modular tensor category gives rise to a topological field theory. In particular, such a topological field theory assigns finite-dimensional vector spaces to surfaces, possibly with boundary. These vector spaces are constructed from the homomorphism spaces of the category and are called the spaces of chiral conformal blocks, or briefly the block spaces. They depend functorially on the objects of the category assigned to the boundary components, so that it is more appropriate to say that a topological field theory assigns to a surface not only a space, but rather a functor. To diffeomorphisms of surfaces, the topological field theory assigns natural transformations between these functors, which gives rise to projective representations of mapping class groups. These data obey factorization constraints related to the cutting and gluing of surfaces, properties formalized in the notion of a modular functor.

Already early in the development of the theory, it was noted that such projective representations can be obtained even if the modular category is not semisimple. In this case, the arising functors are no longer exact, but they are still left exact, so that it is natural to study their derived functors. The main result of the present work is that the cohomology groups arising from these derived functors still carry a projective action of the mapping class group in such a way that the original action is recovered in degree zero. It is therefore appropriate to call these spaces derived block spaces.

This result can be applied to a special case: For the category, one can use the representation category of a factorizable ribbon Hopf algebra, and for the surface, one can use the torus. In this case, the mapping class group is isomorphic to the modular group, and the cohomology groups become the Hochschild cohomology groups of the Hopf algebra. In this way, we recover the main result of our previous article [LMSS1].

The approach used to reach our present result is inspired by the principle of 'propagation of vacua' (cf. [TUY, Par. 2.2, p. 476]). This technique introduces an additional boundary component on the surface that, if labeled with the monoidal unit of the category, leads to block spaces that are canonically isomorphic to the block spaces without the additional boundary component. In our construction, the new boundary component serves as the position where we insert a projective resolution of the unit object. By the functoriality of the block spaces, we then obtain a cochain complex on which the mapping class group of the surface with one additional boundary component acts. The main point of the argument will be that this additional boundary component can be closed again when passing to cohomology. To establish this, we proceed in two steps, using standard techniques from the theory of mapping class groups, namely the capping sequence in the first step and the Birman sequence in the second step.

The present work sets out the general theory needed to establish the result just described, but also prepares the ground for the computation of explicit examples of these mapping class group representations, which will be addressed in our forthcoming article [LMSS2]. A new aspect of our approach is that we consider an action of the mapping class group on the fundamental group by requiring that the base point of the fundamental group be kept fixed. In this way, it is possible to avoid the necessity to identify homomorphisms that are related by simultaneous conjugation that can be found in many of the other articles on topological or conformal field theory.

The material is organized as follows: Section 1 reviews surfaces, fundamental groups, and mapping class groups as well as the capping sequence and the Birman sequence. It also introduces the action of the mapping class group on the fundamental group just mentioned. Section 2 explains how representations of mapping class groups are assigned to certain tensor categories in topological field theory. For this, we use the framework created by V. Lyubashenko in his articles [L1] and [L2]; in particular, we use the approach to surfaces via nets and ribbon graphs described in his articles. In Section 3, we first state and prove our main result and then explain why this result generalizes our previous one from [LMSS1]. In fact, our present result was already mentioned in [LMSS1], and it was also described in [FS2]. Here, we are now supplying proofs for our claims.

Throughout the text, the word 'projective' is used frequently. It has two very different meanings: When speaking about projective modules and projective resolutions, the term is used in the sense of ring theory and homological algebra. When speaking about projective space and projective representations or actions, the term is used in the sense of projective geometry. In particular, for a vector space V, we denote the associated projective space, i.e., the set of its one-dimensional subspaces, by P(V), and the projectivity or homography induced by a bijective linear map f by P(f). The set of all projectivities from P(V) to itself forms the projective linear group PGL(V), which is isomorphic to the general linear group GL(V) modulo the scalar multiples of the identity transformation. By a projective representation or projective action, we mean a group homomorphism from a given group to the projective linear group. We will assume throughout that our base field K is perfect, i.e., that all of its algebraic extensions are separable (cf. [J, Chap. IV, § 1, p. 146]). In particular, algebraically closed fields are perfect, as are fields of characteristic zero.

In general, we compose mappings and morphisms in a category from right to left, i.e., we have  $(g \circ f)(x) = g(f(x))$ , so that f is applied first. In contrast, we concatenate paths in the fundamental group from left to right, i.e., in the concatenation  $\kappa \gamma$ , the path  $\kappa$  is traced out first, while the path  $\gamma$  is traced out afterwards.

We use the symbol  $\mathcal{V}$  for the category of finite-dimensional vector spaces and the symbol  $S_n$  for the symmetric group in n letters. Additional notation will be explained in the text.

While carrying out this research, the first and the third author were partially supported by the RTG 1670 'Mathematics inspired by String theory and Quantum Field Theory' and by the 'Deutsche Forschungsgemeinschaft' under Germany's Excellence Strategy EXC 2121 'Quantum Universe' 390833306, while the second and the fourth author were partially supported by NSERC grant RGPIN-2017-06543.

### 1 Mapping class groups

#### **1.1** The classification of surfaces

The principal result of the classification of surfaces asserts that every compact, connected, orientable, smooth 2-dimensional manifold is diffeomorphic to the connected sum  $\Sigma_g$  of a sphere with  $g \ge 0$  tori, where it is understood that a diffeomorphism is arbitrarily often differentiable, and smoothness is understood in the same way. The number g of the attached tori, which are often called handles, is uniquely determined by the surface and is called its genus. If we include manifolds with boundary, then the classification theorem asserts that a compact, connected, orientable, smooth 2-dimensional manifold with boundary is diffeomorphic to  $\Sigma_g$  with  $n \ge 0$  open disks removed, where the disks have to satisfy the requirement that their closures do not intersect. We will denote this surface by  $\Sigma_{g,n}$ , so that  $\Sigma_g = \Sigma_{g,0}$ . If the boundary is not empty, it is the finite disjoint union of n connected components, called the boundary components. We assume that on each of the boundary components, a point has been distinguished, or marked, as we also say. The Euler characteristic of  $\Sigma_{g,n}$  is

$$\chi(\Sigma_{g,n}) = 2 - 2g - n.$$

By assumption, the surfaces above are orientable, but when we refer to  $\Sigma_{g,n}$ , we will always assume that an orientation has been chosen. In the pictures of  $\Sigma_{g,n}$  that we will encounter below,  $\Sigma_{g,n}$  will be drawn embedded into 3-dimensional space, in which case we will assume that the orientation is represented by the outward-pointing normal vector field.

In the standard proof of the classification theorem, the surface is realized not as a connected sum, but rather as a quotient space of a polygon whose edges are labeled in a certain normal form. In the normal form for the closed surface  $\Sigma_g$ , the edges are labeled consecutively as

$$\alpha_1, \beta_1, \alpha_1^{-1}, \beta_1^{-1}, \alpha_2, \beta_2, \alpha_2^{-1}, \beta_2^{-1}, \dots, \alpha_g, \beta_g, \alpha_g^{-1}, \beta_g^{-1},$$

while for the surface  $\Sigma_{g,n}$  with boundary, the edges are labeled consecutively as

$$\alpha_1, \beta_1, \alpha_1^{-1}, \beta_1^{-1}, \dots, \alpha_g, \beta_g, \alpha_g^{-1}, \beta_g^{-1}, \xi_1, \rho_1, \xi_1^{-1}, \dots, \xi_n, \rho_n, \xi_n^{-1}.$$

In the case g = n = 3, this polygon has the form



where the inverse signs have been depicted by reversing the orientation of the corresponding arrow. In the case of a closed surface, the quotient space is then realized in such a way that all the vertices of the polygon become a single point x, while the edges labeled  $\alpha_i$  and  $\beta_i$  are, respectively, identified with their counterparts labeled  $\alpha_i^{-1}$  and  $\beta_i^{-1}$ , which are traced out in the opposite direction, as we use the general convention that  $\gamma^{-1}(t) := \gamma(1-t)$  for a path defined on the unit interval. In the case of the surface  $\Sigma_{g,n}$  with boundary, in addition an edge labeled  $\xi_j$  is identified in the same way with its counterpart labeled  $\xi_j^{-1}$ . The edges labeled  $\rho_j$  then become closed curves in the quotient that represent the *n* boundary components (cf. [ST, Kap. 6, § 40, p. 142ff], [Mas, Chap. I, Sec. 10, p. 37ff]). In this case, not all vertices of the polygon map to the same point *x*, but only those that are not the start point or the end point of one of the edges  $\rho_1, \ldots, \rho_n$ . The start point of  $\rho_j$  is instead identified with the end point of  $\rho_j$  and in this way yields the marked point on the *j*-th boundary component.

If we carry out only the second identification in the case depicted above, i.e., the identification of  $\xi_j$  with  $\xi_j^{-1}$ , we obtain the following picture of the intermediate stage:



When we carry out all the identifications, the edges labeled  $\alpha_i$  and  $\beta_i$  become in the quotient the closed curves that appear in the following picture:



This realization of the surface is called the polygon model of the surface, as opposed to the connected sum model that we briefly described at the beginning of the paragraph. Note that, in order to obtain the outward-pointing orientation of the surface mentioned above, the surface of the polygon needs to be oriented by a normal vector field that is orthogonal to the page and is pointing away from the reader, which is not the standard orientation of the plane used elsewhere in the text, for example in Paragraph 1.6. Furthermore, in this model the boundary curves  $\rho_j$  do not carry the orientation that is induced by the surface on its boundary, but rather the opposite orientation.

### 1.2 The fundamental group

In the polygon model of a closed surface, we use the point x in the quotient that all vertices of the polygon map to as the base point of the fundamental group  $\pi_1(\Sigma_g, x)$ . The elements of the fundamental group are the homotopy classes  $[\gamma]$  of paths  $\gamma$ defined on the unit interval, where homotopy is relative to this base point x. It turns out that the relative homotopy classes of the 2g elements  $\alpha_1, \beta_1, \ldots, \alpha_g, \beta_g$  just defined generate the fundamental group. In the connected sum model, each pair  $\alpha_i, \beta_i$  lies on one of the g tori that we attached to a sphere at the beginning of Paragraph 1.1. Two neighboring tori are connected by the path  $\mu_i := \alpha_{i+1}^{-1}\beta_i\alpha_i\beta_i^{-1}$ , where it is understood that  $\mu_g := \alpha_1^{-1}\beta_g\alpha_g\beta_g^{-1}$  if i = g. To see this, we first observe that the path  $\beta_i\alpha_i\beta_i^{-1}$  is homotopic to a kind of mirror image of  $\alpha_i$ :



If we now also concatenate with the curve  $\alpha_{i+1}^{-1}$ , we see that the curve  $\mu_i$  is homotopic to a curve that connects the *i*-th and (i + 1)-st handle:



These pictures in fact illustrate that the concatenated path  $\mu_g \mu_{g-1} \cdots \mu_1$  is homotopic to the constant path based at x, which means that

$$(\alpha_1^{-1}\beta_g\alpha_g\beta_g^{-1})(\alpha_g^{-1}\beta_{g-1}\alpha_{g-1}\beta_{g-1}^{-1})\cdots(\alpha_2^{-1}\beta_1\alpha_1\beta_1^{-1})$$

represents the unit element in the fundamental group. Alternatively, by conjugating with  $\alpha_1$  and inverting, we see that

$$\alpha_1\beta_1\alpha_1^{-1}\beta_1^{-1}\alpha_2\beta_2\alpha_2^{-1}\beta_2^{-1}\cdots\alpha_g\beta_g\alpha_g^{-1}\beta_g^{-1}$$

represents the unit element in the fundamental group. This fact is even easier to see in the polygon model, where this path exactly traces out the boundary of the polygon and can therefore be contracted to a point in the interior of the polygon. Using the Seifert-van Kampen theorem, one can show that this relation is a defining relation for the fundamental group (cf. [F, Prop. 17.6, p. 242]).

For the surfaces  $\Sigma_{g,n}$  with boundary, additional generators are needed for the fundamental group, which we denote by  $\delta_1, \ldots, \delta_n$ . Starting at the base point,  $\delta_j$  circles around the *j*-th boundary component. Together  $\delta_1, \ldots, \delta_n$  form a so-called bouquet of circles. In the polygon model,  $\delta_j$  arises by mapping the concatenation of  $\xi_j$ ,  $\rho_j$ , and  $\xi_i^{-1}$  to the quotient. We therefore see that in this case,

$$\alpha_1\beta_1\alpha_1^{-1}\beta_1^{-1}\alpha_2\beta_2\alpha_2^{-1}\beta_2^{-1}\cdots\alpha_g\beta_g\alpha_g^{-1}\beta_g^{-1}\delta_1\cdots\delta_n$$

is homotopic to the constant path based at x, and again this yields a defining relation for the fundamental group (cf. [AS, Chap. I, No. 43B, p. 100]). Because it is possible to solve in this relation for the last of the new generators, this implies that, in the presence of boundary components, the fundamental group of  $\Sigma_{g,n}$  is a free group on 2g+n-1 generators. We note that, by a slight movement, it is always possible to find a representative in a relative homotopy class that does not intersect the boundary. In the sequel, we will assume in particular that this has been done for the generators  $\delta_i$ .

#### **1.3** Mapping class groups

An important object associated with the surface  $\Sigma_{g,n}$  is its mapping class group  $\Gamma_{g,n}$ . There are several variants for the definition of this group in the literature; we will now explain which one is used in this article and how it compares to other variants.

By the boundary invariance theorem (cf. [D, Chap. IV, Prop. 3.9, p. 61]), a diffeomorphism of  $\Sigma_{g,n}$  will necessarily map a boundary component to a boundary component, but we require that the diffeomorphisms we consider also map the marked point on each boundary component to the corresponding marked point on the other boundary component. The group Diffeo<sup>+</sup>( $\Sigma_{g,n}$ ) of orientation-preserving diffeomorphisms of  $\Sigma_{g,n}$  that permute the marked points becomes a topological group when endowed with the compact-open topology (cf. [A, Thm. 4, p. 598]). By the fundamental property of the compact-open topology (cf. [Q, Satz 14.17, p. 167]), the path-component Diffeo<sup>+</sup><sub>0</sub>( $\Sigma_{g,n}$ ) of the identity mapping consists precisely of those orientation-preserving diffeomorphisms that are isotopic to the identity. It is a normal subgroup. Continuity implies that the elements of Diffeo<sup>+</sup><sub>0</sub>( $\Sigma_{g,n}$ ) cannot permute the marked points, but rather need to preserve them individually. We define the mapping class group as the corresponding quotient group:

**Definition 1.1.** The mapping class group  $\Gamma_{g,n}$  of  $\Sigma_{g,n}$  is the quotient group

$$\Gamma_{g,n} := \text{Diffeo}^+(\Sigma_{g,n}) / \text{Diffeo}^+_0(\Sigma_{g,n})$$

In accordance with our notation for surfaces in Paragraph 1.1, we will also write  $\Gamma_g$ if n = 0. The mapping class of a diffeomorphism  $\psi \in \text{Diffeo}^+(\Sigma_{g,n})$  will be denoted by  $[\psi]$ . In view of this definition, two diffeomorphisms represent the same element in the mapping class group if and only if they are contained in the same path-component of Diffeo<sup>+</sup>( $\Sigma_{g,n}$ ); i.e., if and only if they are isotopic. Because the path connecting them is contained in Diffeo<sup>+</sup>( $\Sigma_{g,n}$ ), the corresponding isotopy must permute the marked points at each time of the deformation process. By continuity, this is only possible if all diffeomorphisms occurring in this process, and in particular those at the beginning and the end, permute the marked points in the same way. By enumerating the marked points, we can therefore obtain a group homomorphism

$$p\colon \Gamma_{g,n}\to S_n$$

to the symmetric group  $S_n$ . We call the kernel of p the pure mapping class group and denote it by  $\Pr_{g,n}$ . An element in the pure mapping class group can be represented by a diffeomorphism that restricts to the identity on the entire boundary, not only on the marked points. As explained in [FM, Sec. 1.4, p. 42], the mapping class group can alternatively be defined by using homeomorphisms instead of diffeomorphisms.

For a subset U of  $\Sigma_{g,n}$ , we also consider the subgroup of Diffeo<sup>+</sup>( $\Sigma_{g,n}$ ) consisting of those diffeomorphisms that restrict to the identity on U. If we divide it by its path component of the identity mapping, the arising quotient group  $\Gamma_{g,n}(U)$  is called the relative mapping class group modulo U. It comes with a canonical group homomorphism

$$F_U \colon \Gamma_{q,n}(U) \to \Gamma_{q,n}$$

called the forgetful map, because it arises by forgetting the information about the restriction to U. This homomorphism is not necessarily injective, because it might happen that a diffeomorphism is isotopic to the identity, although the diffeomorphisms that occur during this deformation process cannot be chosen so that they restrict to the identity on U. If  $U = \{u\}$  consists of a single point u, we will also write  $\Gamma_{q,n}(u)$  and  $F_u$  instead of  $\Gamma_{q,n}(\{u\})$  and  $F_{\{u\}}$ .

The definition of mapping class groups used here is the one from [L1, Sec. 4, p. 485]. Often, other definitions are used, for example in [FM, Sec. 2.1, p. 44] or [Ko, Sec. 1, p. 101]. In these definitions, there are additional marked points in the interior, not on the boundary like in our definition. These points are frequently called punctures. In contrast to our definition, both the definition in [FM] and the definition in [Ko] require that the diffeomorphisms restrict to the identity on the boundary. In [FM], the punctures in the interior may be permuted by a diffeomorphism, while they are required to be preserved individually in [Ko].

#### 1.4 Dehn twists

An annulus can be defined as the Cartesian product  $A := S^1 \times [0, 2\pi]$  of the unit circle  $S^1 := \{z \in \mathbb{C} \mid |z| = 1\}$  and the interval  $[0, 2\pi] \subset \mathbb{R}$ . A is an orientable 2-manifold with boundary, which is diffeomorphic to  $\Sigma_{0,2}$  and therefore sometimes called a binion. We single out one of the two possible orientations by requiring that in the tangent space of A at the point  $(1,\pi)$  the tangent vector  $v_1$  to the curve  $t \mapsto (1,\pi+t)$  and the tangent vector  $v_2$  to the curve  $t \mapsto (e^{it},\pi)$  form a positively oriented basis  $v_1, v_2$ . Note that both curves start at the specified point for the parameter value t = 0. On A, we define the twist map

$$\mathfrak{d} \colon A \to A, \ (z,t) \mapsto (ze^{it},t)$$

Now suppose that  $\gamma: S^1 \to \Sigma_{g,n}$  is a simple closed curve that does not intersect the boundary. It follows from the existence of tubular neighborhoods (cf. [H, Chap. 4, Thm. 5.2, p. 110]) that there is an orientation-preserving embedding  $\phi: A \to \Sigma_{g,n}$  with the property that  $\phi(z, 0) = \gamma(z)$ .

**Definition 1.2.** We define  $\mathfrak{d}_{\gamma} \colon \Sigma_{g,n} \to \Sigma_{g,n}$  as the map

$$p \mapsto \begin{cases} (\phi \circ \mathfrak{d} \circ \phi^{-1})(p) & : p \in \phi(A) \\ p & : p \notin \phi(A) \end{cases}$$

Note that this map is a homeomorphism, not a diffeomorphism. But as we mentioned above, the mapping class group can also be defined using homeomorphisms, so that  $\mathfrak{d}_{\gamma}$  determines an element  $[\mathfrak{d}_{\gamma}]$  in the mapping class group, called the Dehn twist along  $\gamma$ . Since  $\mathfrak{d}(z,0) = (z,0)$ , we have  $\mathfrak{d}_{\gamma} \circ \gamma = \gamma$ .

It follows from the isotopy of tubular neighborhoods (cf. [H, Chap. 4, Thm. 5.3, p. 112]) that the mapping class of  $\mathfrak{d}_{\gamma}$  does not depend on  $\phi$ . Furthermore, any curve isotopic to  $\gamma$  yields the same Dehn twist in the mapping class group. Further details on Dehn twists and their geometric meaning can be found in [FM, Sec. 3.1, p. 64ff]. It is important to note that in some references the other orientation for the annulus A is used in the definition of a Dehn twist, for example in [L2, Par. 2.2, p. 316f] and [Ko, Sec. 2, p. 102]. Dehn twists defined using the other orientation are the inverses of the Dehn twists as they are defined here.

It is almost immediate from this definition that non-intersecting curves  $\gamma_1$  and  $\gamma_2$  give rise to commuting Dehn-twists  $\mathfrak{d}_{\gamma_1}$  and  $\mathfrak{d}_{\gamma_2}$ . This and other basic properties of Dehn-twists are discussed in [FM, Sec. 3.2, p. 73ff].

We will introduce some special notation for a number of Dehn twists along certain curves. The first of these are the following: In the construction of  $\Sigma_{g,n}$  from  $\Sigma_g$ , the removal of the *j*-th open disk from  $\Sigma_g$  leaves a connected component of the boundary that is diffeomorphic to the unit circle  $S^1$ . In the polygon model discussed in Paragraph 1.1, this curve is parametrized by  $\rho_j$ . By moving this curve slightly to the interior, for example by using a collar (cf. [H, Chap. 4, Thm. 6.1, p. 113]), we obtain a simple closed curve  $\partial_j$  that is freely homotopic to  $\rho_j$ , and therefore also to  $\delta_j$ , but does not intersect the boundary. We will use the notation  $\mathfrak{d}_j$  for the Dehn twist along  $\partial_j$ . As explained in [FM, Par. 4.2.5, p. 102],  $\mathfrak{d}_j$  is contained in the center of the mapping class group.

#### 1.5 Braidings

Besides Dehn twists, the second type of elements of the mapping class group that we will need are the braidings of the boundary components. To define them, we consider the surface B in  $\mathbb{R}^2$  that consists of the closed unit disk from which two open disks with radius 1/4 and centers (1/2, 0) and (-1/2, 0) on the x-axis, respectively, have been removed. B is an orientable 2-manifold with boundary, which is sometimes called a trinion, as it is diffeomorphic to  $\Sigma_{0,3}$ . We orient B by requiring that the canonical basis at the origin is positively oriented. On the boundary components, i.e., the unit circle and the two smaller circles, we choose the base points (0, 1), (1/2, 1/4), and (-1/2, 1/4), respectively. As discussed in [T, Sec. V.2.5, p. 251], there is an orientation-preserving diffeomorphism  $\mathfrak{b}: B \to B$ , which we call the braiding map, that is the identity on the unit circle, interchanges the two smaller circles as well as their base points while preserving their orientations, and transforms the line segments connecting the base point of the unit circle with the base points of the smaller circles as indicated in the picture



(cf. also [T, Fig. 2.9, p. 254]). As also stated in [T, loc. cit.], these conditions determine **b** up to isotopy. It should be noted that the braiding map we define here

is the inverse of the braiding map in [T, Sec. V.2.5, p. 251], but coincides with the map used in [FM, Par. 5.1.1, p. 119].

We will now use the braiding map to define the braiding of two boundary components of  $\Sigma_{q,n}$ . It should be noted that it is not possible to define such a braiding if the boundary components are distributed over the surface in a somewhat arbitrary fashion, as in our description of the surface in terms of the connected sum model, as this definition depends not only on these two boundary components alone, but also on their position relative to the other boundary components. Rather, it is necessary that the boundary components are not only numbered, but in fact arranged in a certain order, like in the polygon model described in Paragraph 1.1. We use the standard arrangement of the boundary components in this model as follows: For two indices satisfying  $1 \le i < j \le n$ , we choose an orientation-preserving embedding  $\varphi_{i,j}: B \to \Sigma_{g,n}$  of surfaces that maps the circle with center (-1/2, 0) to the *i*-th boundary component of  $\Sigma_{g,n}$  and the circle with center (1/2, 0) to the *j*-th boundary component of  $\Sigma_{q,n}$ , mapping base points to base points. The unit circle is mapped to a smooth curve  $\gamma_{i,j}$  on  $\Sigma_{g,n}$  that is defined as follows: Starting at x, we follow a path slightly right from  $\xi_i$  until we reach the curve  $\partial_i$ , which we follow until we need to return along  $\xi_i$ , which we now do on its left side, which is the right side for the reversed orientation of  $\xi_i$ . Shortly before reaching x, however, we turn right and follow a path slightly right from  $\xi_j$  until we again reach  $\partial_j$ , which we follow in the same way as before until we need to return along  $\xi_i$ , which we now do on its left side until we reach the base point x. The curve  $\gamma_{i,j}$  is illustrated in the following picture:



We now define the braiding of the *i*-th and the *j*-th boundary component in the same way as we defined Dehn twists in Paragraph 1.4:

**Definition 1.3.** We define  $\mathfrak{b}_{i,j}: \Sigma_{g,n} \to \Sigma_{g,n}$  as the map

$$p \mapsto \begin{cases} (\varphi_{i,j} \circ \mathfrak{b} \circ \varphi_{i,j}^{-1})(p) & : p \in \varphi_{i,j}(B) \\ p & : p \notin \varphi_{i,j}(B) \end{cases}$$

Again, this map is a homeomorphism, not a diffeomorphism, but as for Dehn twists, it nonetheless determines an element  $[\mathfrak{b}_{i,j}]$  in the mapping class group, called the braiding of the *i*-th and the *j*-th boundary component. It is obvious that under the projection  $p: \Gamma_{g,n} \to S_n$  to the symmetric group introduced in Paragraph 1.3,  $[\mathfrak{b}_{i,j}]$ maps to the transposition of *i* and *j*.

We note that a mapping class very similar to  $[\mathfrak{b}_{i,j}]$ , using punctures instead of boundary components, is called a half-twist in [FM, Sec. 9.4, p. 255]. There are several relations between half-twists and Dehn twists; one is explained right there, another one will be discussed in Paragraph 1.8.

#### **1.6** The action on the fundamental group

During certain steps of our argument, it will be necessary to cut out small disks from our surface or at least to avoid certain points. For a Dehn twist, it makes a difference how we avoid this point. Suppose that a homotopy class is represented by a smooth simple closed curve  $\gamma$  that starts and ends at a point x on our surface. We choose a neighborhood U of x that is diffeomorphic to the open unit disk in  $\mathbb{R}^2$  via an orientation-preserving diffeomorphism  $\phi$  that sends x to the origin. As before, we assume that the unit disk has the orientation in which the canonical basis is a positively oriented basis for the tangent space at the origin. By perhaps passing to a smaller neighborhood, we can assume that  $\gamma$  passes through the unit disk as indicated on the left of the picture below. Because  $\phi$  is orientation-preserving, it is meaningful to talk about pushing off  $\gamma$  to the left or to the right, as indicated in the middle and right picture below:



In this way, we obtain two curves  $\gamma'$  and  $\gamma''$  whose free homotopy classes in  $\Sigma_{g,n} \setminus \{x\}$  do not depend on our choices. Both curves avoid a small neighborhood of the point x

and can be chosen so that they are still simple closed curves. Although we have in general  $[\mathfrak{d}_{\gamma'}] \neq [\mathfrak{d}_{\gamma''}] \in \Gamma_{g,n}(x)$ , we clearly have

$$F_x([\mathfrak{d}_{\gamma'}]) = F_x([\mathfrak{d}_{\gamma''}]) = [\mathfrak{d}_{\gamma}]$$

for their images under the forgetful map  $F_x \colon \Gamma_{g,n}(x) \to \Gamma_{g,n}$ .

As explained in [FM, Par. 8.2.7, p. 235], we can choose for x the base point of the fundamental group to get an action of  $\Gamma_{g,n}(x)$  on  $\pi_1(\Sigma_{g,n}, x)$ . In particular, the Dehn twists  $\mathfrak{d}_{\gamma'}$  and  $\mathfrak{d}_{\gamma''}$  act on  $\pi_1(\Sigma_{q,n}, x)$ . We now give explicit formulas for this action in the case where the homotopy class acted upon can be represented by a curve  $\kappa$ that does not intersect  $\gamma$ , except at the base point x. It should be noted that, in the present situation, the curves  $\kappa$  and  $\gamma$ , and therefore also  $\gamma'$  and  $\gamma''$ , are oriented, although, according to our construction, Dehn twists like  $\mathfrak{d}_{\gamma'}$  do not depend on the orientation of  $\gamma'$ . In each homotopy class, we can choose smooth representatives (cf. [FM, Par. 1.2.2, p. 26]), which by definition of smoothness have nowhere a vanishing tangent vector. If we in addition can choose these representatives so that their tangent vectors  $\dot{\kappa}_x$  and  $\dot{\gamma}_x$  at x are linearly independent, we say that  $\kappa$  and  $\gamma$ intersect transversely at x. If  $\dot{\kappa}_x$  and  $\dot{\gamma}_x$  form a positively oriented basis of the tangent space at x, we define the algebraic intersection number  $i_A(\kappa, \gamma)$  as +1, and if they form a negatively oriented basis of the tangent space at x, we define the algebraic intersection number  $i_A(\kappa, \gamma)$  as -1. This definition can be extended to curves with finitely many intersection points by defining  $i_A(\kappa, \gamma)$  as the sum of the so-determined signs at all intersection points. This definition is illustrated in the picture



in which the standard orientation of the plane has been used, i.e., the orientation in which the canonical basis is positive and the normal vector points towards the reader.

The action of the two Dehn twists on the homotopy class  $[\kappa]$  of  $\kappa$  is summarized in the following table:

	$i_A(\kappa,\gamma) = 1$	$i_A(\kappa,\gamma) = -1$
$\mathfrak{d}_{\gamma'}([\kappa])$	$[\kappa\gamma]$	$[\gamma^{-1}\kappa]$
$\mathfrak{d}_{\gamma''}([\kappa])$	$[\gamma\kappa]$	$[\kappa \gamma^{-1}]$

Here, expressions like  $[\kappa\gamma] = [\kappa][\gamma]$  mean the product in the fundamental group: The concatenation of first  $\kappa$ , followed by  $\gamma$ . To understand the table, let us consider how the entry for  $\mathfrak{d}_{\gamma'}([\kappa])$  in the case  $i_A(\kappa, \gamma) = 1$  comes about: The curve starts at x with  $\kappa$ . In view of how  $\gamma'$  has been pushed off  $\gamma$  and how the curves intersect,  $\kappa$  is moving away from  $\gamma'$  and traces out  $\kappa$  almost completely before approaching the base point x upon returning. But before reaching x, it encounters  $\gamma'$  and at this point is forced to turn left in view of our definition of a Dehn twist. But because  $i_A(\kappa, \gamma) = 1$ , turning left is in this case the same as following  $\gamma$  in the direction of its orientation. The other entries arise from analogous considerations.

It is of course possible that we cannot choose representatives whose tangent vectors  $\dot{\kappa}_x$  and  $\dot{\gamma}_x$  at x are linearly independent. In this case, the two curves do not intersect transversely, and by pushing off  $\gamma$  either to the left or to the right, we can avoid intersection completely. Depending on the orientation of  $\gamma$ , the situation is described by one of the following two pictures:



In the first case, the actions of the Dehn twists are given by

$$\mathfrak{d}_{\gamma'}([\kappa]) = [\gamma^{-1}\kappa\gamma] \quad \text{and} \quad \mathfrak{d}_{\gamma''}([\kappa]) = [\kappa],$$

while in the second case they are given by

 $\mathfrak{d}_{\gamma'}([\kappa]) = [\kappa]$  and  $\mathfrak{d}_{\gamma''}([\kappa]) = [\gamma \kappa \gamma^{-1}].$ 

In both cases, these formulas are valid regardless of the orientation of  $\kappa$ .

#### 1.7 Dehn twists for special curves

In Paragraph 1.4, we have already introduced the notation  $\mathfrak{d}_j$  for the Dehn twist determined by the curve  $\partial_j$ . We now also assign names to the Dehn twists derived from the other curves stemming from the polygon model discussed in Paragraph 1.1 and define

$$\mathfrak{t}_i \coloneqq \mathfrak{d}_{lpha_i}, \qquad \qquad \mathfrak{r}_i \coloneqq \mathfrak{d}_{eta_i}, \qquad \qquad \mathfrak{n}_i \coloneqq \mathfrak{d}_{\mu_i}.$$

Because the curves  $\alpha_i$  and  $\beta_i$  intersect exactly once, the corresponding Dehn twists satisfy the braid relation  $[\mathbf{t}_i \mathbf{r}_i \mathbf{t}_i] = [\mathbf{r}_i \mathbf{t}_i \mathbf{r}_i]$  (cf. [FM, Par. 3.5.1, Prop. 3.11, p. 77]). For the elements  $\mathbf{s}_i := \mathbf{t}_i^{-1} \mathbf{r}_i^{-1} \mathbf{t}_i^{-1}$ , the braid relation implies that  $[\mathbf{s}_i \mathbf{t}_i \mathbf{s}_i^{-1}] = [\mathbf{r}_i]$ .

These special Dehn twists are important in view of the following result, which is known as the Dehn-Lickorish theorem:

**Theorem 1.4.** If  $g \ge 1$ , the Dehn twists  $[\mathfrak{t}_i]$ ,  $[\mathfrak{r}_i]$ , and  $[\mathfrak{n}_l]$ , for  $i = 1, \ldots, g$  and  $l = 1, \ldots, g - 1$ , which are called the Lickorish generators, generate the mapping class groups  $\Gamma_g = \Gamma_{g,0}$  and  $\Gamma_{g,1}$ .

A proof of this result can be found in [FM, Par. 4.4.4, p. 113ff]. We note that in the case g = 0, the groups  $\Gamma_g$  and  $\Gamma_{g,1}$  are trivial anyway as a consequence of the Alexander lemma (cf. [FM, Par. 2.2.1, p. 47ff]). The case g = 1 will be discussed in more detail in Paragraph 1.12.

In the case where there is more than one boundary component, we need additional Dehn twists around the curves  $\zeta_l$ , which we denote by  $\mathfrak{z}_l$  for  $l = 1, \ldots, n-1$ . The curve  $\zeta_l$  separates the *l*-th and *l*+1-st boundary component and connects to the *g*-th handle as shown in the picture for the case g = 3 and n = 2:



For this case, we have the following variant of the Dehn-Lickorish theorem (cf. [FM, Par. 4.4.4, p. 113f]):

**Theorem 1.5.** If  $g \ge 1$  and  $n \ge 2$ , the mapping classes  $[\mathfrak{t}_i]$ ,  $[\mathfrak{r}_i]$ ,  $[\mathfrak{n}_j]$ ,  $[\mathfrak{d}_k]$ ,  $[\mathfrak{z}_l]$ , and  $[\mathfrak{b}_{l,l+1}]$ , for  $i = 1, \ldots, g$ ,  $j = 1, \ldots, g - 1$ ,  $k = 1, \ldots, n$ , and  $l = 1, \ldots, n - 1$ , generate the mapping class group  $\Gamma_{g,n}$ .

We can assume without loss of generality that the curve  $\partial_j$  does not only avoid the *n* boundary components, but also the base point *x*. Then the corresponding Dehn twist  $\mathfrak{d}_j$  is contained in  $\Gamma_{g,n}(x)$  and therefore can act on  $\pi_1(\Sigma_{g,n}, x)$ , as explained in Paragraph 1.6. Since any homotopy class in  $\pi_1(\Sigma_{g,n}, x)$  can be represented by a curve that does not intersect  $\partial_j$ , this action is trivial. However, the curves  $\alpha_i$ ,  $\beta_i$ , and  $\mu_i$ start and end at the base point *x*, and as explained in Paragraph 1.6, it makes a difference for the action on  $\pi_1(\Sigma_{g,n}, x)$  whether the curves meet before or after the base point *x*. We will now make specific choices for these curves. For  $\mathfrak{t}_i = \mathfrak{d}_{\alpha_i}$ , we choose  $\mathfrak{t}'_i := \mathfrak{d}_{\alpha'_i}$ , for  $\mathfrak{r}_i = \mathfrak{d}_{\beta_i}$ , we choose  $\mathfrak{r}''_i := \mathfrak{d}_{\beta''_i}$ , and for  $\mathfrak{n}_i = \mathfrak{d}_{\mu_i}$ , we choose  $\mathfrak{n}''_i := \mathfrak{d}_{\mu''_i}$ . For  $\mathfrak{s}_i$ , we mix the choices and define  $\mathfrak{s}'_i := \mathfrak{t}'_i^{-1}\mathfrak{r}''_i^{-1}\mathfrak{t}'_i^{-1}$ . These maps act on the generators of the fundamental group as follows:

**Proposition 1.6.** Suppose that  $i \leq g$ .

- 1. For  $j \leq n$ , the Dehn twists  $\mathfrak{t}'_i([\delta_j]) = \mathfrak{r}''_i([\delta_j]) = [\delta_j]$ . If  $i \neq g$ , we also have  $\mathfrak{n}''_i([\delta_j]) = [\delta_j]$ .
- 2. We have  $\mathfrak{t}'_i([\beta_i]) = [\beta_i \alpha_i]$  and  $\mathfrak{t}'_i([\alpha_i]) = [\alpha_i]$ . For  $j \neq i$ , we have  $\mathfrak{t}'_i([\beta_j]) = [\beta_j]$  and  $\mathfrak{t}'_i([\alpha_j]) = [\alpha_j]$ .
- 3. We have  $\mathbf{r}''_i([\alpha_i]) = [\alpha_i \beta_i^{-1}]$  and  $\mathbf{r}''_i([\beta_i]) = [\beta_i]$ . For  $j \neq i$ , we have  $\mathbf{r}''_i([\alpha_j]) = [\alpha_j]$  and  $\mathbf{r}''_i([\beta_j]) = [\beta_j]$ .
- 4. For  $i = 1, \ldots, g 1$ , we have  $\mathfrak{n}''_i([\alpha_i]) = [\alpha_i]$  as well as

$$\mathfrak{n}_i''([\beta_i]) = [\mu_i \beta_i], \quad \mathfrak{n}_i''([\alpha_{i+1}]) = [\mu_i \alpha_{i+1} \mu_i^{-1}], \quad \mathfrak{n}_i''([\beta_{i+1}]) = [\beta_{i+1} \mu_i^{-1}].$$

For  $j \neq i$  and  $j \neq i+1$ , we have  $\mathfrak{n}''_i([\alpha_j]) = [\alpha_j]$  and  $\mathfrak{n}''_i([\beta_j]) = [\beta_j]$ .

5. We have  $\mathfrak{s}'_i([\alpha_i]) = [\alpha_i \beta_i \alpha_i^{-1}]$  and  $\mathfrak{s}'_i([\beta_i]) = [\alpha_i^{-1}]$ . For  $j \neq i$ , we have that  $\mathfrak{s}'_i([\alpha_j]) = [\alpha_j]$  and  $\mathfrak{s}'_i([\beta_j]) = [\beta_j]$ .

**Proof.** (1) We first note that  $\alpha_i$  and  $\beta_i$  are oriented so that  $\alpha'_i$  and  $\beta''_i$  arise by pushing  $\alpha_i$  and  $\beta_i$  off the base point x in the direction of the *i*-th attached torus, so that we could cut the *i*-th attached torus off again in such a way that the base point x on the one hand and  $\alpha'_i$  and  $\beta''_i$  on the other hand would lie on different connected components. If  $j \neq i$ , this implies that the curves  $\alpha'_i$  and  $\beta''_i$  do not intersect the curves  $\alpha_j$  and  $\beta_j$ . This implies that  $\mathfrak{t}'_i([\beta_j]) = [\beta_j]$  and  $\mathfrak{t}'_i([\alpha_j]) = [\alpha_j]$ as well as  $\mathfrak{r}''_i([\alpha_j]) = [\alpha_j]$  and  $\mathfrak{r}''_i([\beta_j]) = [\beta_j]$ . For the same reason,  $\mathfrak{t}'_i$  and  $\mathfrak{r}''_i$  preserve the homotopy class of  $\delta_j$ .

(2) We have  $i_A(\beta_i, \alpha_i) = -i_A(\alpha_i, \beta_i) = 1$ , so that it follows from the table in Paragraph 1.6 that  $\mathfrak{t}'_i([\beta_i]) = [\beta_i\alpha_i]$  and  $\mathfrak{r}''_i([\alpha_i]) = [\alpha_i\beta_i^{-1}]$ . Clearly, we have  $\mathfrak{t}'_i([\alpha_i]) = [\alpha_i]$ and  $\mathfrak{r}''_i([\beta_i]) = [\beta_i]$ . Therefore, the second and the third assertions are now completely proved.

(3) For the fourth assertion, we see from the discussion in Paragraph 1.2 that  $\mu_i$ is oriented in such a way that  $\mu''_i$  arises by pushing  $\mu_i$  off the base point x in the direction of the *i*-th and the i+1-st attached tori, so that we could cut these two tori off again in such a way that the base point x and  $\mu''_i$  would lie on different connected components. For  $j \neq i$  and  $j \neq i+1$ , this implies that  $\mu''_i$  does not intersect  $\alpha_j$ or  $\beta_j$ , so that  $\mathfrak{n}''_i([\alpha_j]) = [\alpha_j]$  and  $\mathfrak{n}''_i([\beta_j]) = [\beta_j]$ . This discussion also shows that  $\mu''_i$ does not intersect  $\alpha_i$ , so that  $\mathfrak{n}''_i([\alpha_i]) = [\alpha_i]$ . On the other hand,  $\mu_i$  intersects  $\beta_i$ and  $\beta_{i+1}$  exactly once, the intersection is transversal, and the intersection numbers are  $i_A(\beta_i, \mu_i) = 1$  and  $i_A(\beta_{i+1}, \mu_i) = -1$ , respectively. It therefore follows from the table in Paragraph 1.6 that

$$\mathfrak{n}_i''([\beta_i]) = [\mu_i \beta_i] \quad \text{and} \quad \mathfrak{n}_i''([\beta_{i+1}]) = [\beta_{i+1} \mu_i^{-1}].$$

In contrast, the curves  $\mu_i$  and  $\alpha_{i+1}$  do not intersect transversally, but rather as in the second picture in Paragraph 1.6, so that  $\mathfrak{n}''_i([\alpha_{i+1}]) = [\mu_i \alpha_{i+1} \mu_i^{-1}]$ . Since  $i \neq g$ , the curve  $\mu''_i$  does not intersect  $\delta_j$ , so that  $\mathfrak{n}''_i([\delta_j]) = [\delta_j]$ .

(4) It remains to show the fifth assertion. For  $j \neq i$ , the claim follows easily from the second and the third assertion. For the case j = i, we argue as follows: By inverting part of the third assertion, we have  $\mathbf{r}_i^{\prime\prime-1}([\alpha_i\beta_i^{-1}]) = [\alpha_i]$ , which implies that  $\mathbf{r}_i^{\prime\prime-1}([\alpha_i])[\beta_i]^{-1} = \mathbf{r}_i^{\prime\prime-1}([\alpha_i])\mathbf{r}_i^{\prime\prime-1}([\beta_i]^{-1}) = [\alpha_i]$ . This in turn yields

$$\mathfrak{s}'_{i}([\alpha_{i}]) = (\mathfrak{t}'^{-1}\mathfrak{r}''^{-1}\mathfrak{t}'^{-1}_{i})([\alpha_{i}]) = (\mathfrak{t}'^{-1}\mathfrak{r}''^{-1}_{i})([\alpha_{i}]) = \mathfrak{t}'^{-1}_{i}([\alpha_{i}\beta_{i}]) = \mathfrak{t}'^{-1}_{i}([\alpha_{i}])\mathfrak{t}'^{-1}_{i}([\beta_{i}]).$$

A second inversion yields  $\mathfrak{t}_i^{\prime-1}([\beta_i]) = [\beta_i \alpha_i^{-1}]$ , so that

$$\mathfrak{s}_i'([\alpha_i]) = [\alpha_i]\mathfrak{t}_i'^{-1}([\beta_i]) = [\alpha_i\beta_i\alpha_i^{-1}]$$

as asserted. The formula for  $\mathfrak{s}'_i([\beta_i])$  follows from a similar computation.

#### **1.8** Dehn twists related to two boundary components

According to our conventions in Paragraph 1.4, the curve  $\delta_i \delta_j$ , for i < j, cannot be used to define a Dehn twist for two reasons: On the one hand, it intersects the boundary, and on the other hand, it is not a simple closed curve. The first problem can be addressed by moving it slightly to the interior with the help of a collar, in the same way as in our treatment of  $\partial_j$  at the end of Paragraph 1.4. As a consequence of the orientation of  $\rho_i$  and  $\rho_j$ , a movement to the interior is a movement to the right of the given curve.

The second problem arises not only from the fact that the curve returns to the base point x at the time of concatenation, when  $\delta_i$  ends and  $\delta_j$  begins, but also from the fact that the paths  $\xi_i$  and  $\xi_j$  are both traced out twice, in opposite directions. To address this problem, we consider the curve  $\gamma_{i,j}$  defined in Paragraph 1.5, which does not intersect the boundary, starts and ends in x, and represents the same relative homotopy class as  $\delta_i \delta_j$  in the fundamental group  $\pi_1(\Sigma_{g,n}, x)$ . Passing to  $\gamma'_{i,j}$ as indicated in Paragraph 1.6, we in addition avoid the base point x. In contrast to  $\delta_i \delta_j$ , the modification  $\gamma'_{i,j}$  is a simple closed curve that can be used to define a Dehn twist. This curve is shown in the picture that appears in the proof below. We denote the Dehn twist along  $\gamma'_{i,j}$  by  $\mathfrak{d}_{i,j}$ .

As explained in [FM, Par. 5.1.1, p. 118f] or [PS, § 7, p. 63], the Dehn twist  $\mathfrak{d}_{i,j}$  is related to the braiding introduced in Paragraph 1.5: If we define the double braiding as  $\mathfrak{q}_{i,j} := \mathfrak{b}_{j,i}\mathfrak{b}_{i,j}$ , it is related to our Dehn twist via the formula

$$[\mathfrak{q}_{i,j}] = [\mathfrak{d}_{i,j}\mathfrak{d}_i^{-1}\mathfrak{d}_j^{-1}].$$

This formula holds because the Dehn twist  $\mathfrak{d}_{i,j}$  not only interchanges the *i*-th and the *j*-th boundary components twice, but also twists these boundary components themselves, while the double braiding moves the boundary components in a parallel fashion, without introducing a twist.

The action of  $q_{i,j}$  on the generators of the fundamental group is given as follows:

**Proposition 1.7.** Suppose that  $1 \le i < j \le n$ .

1. We have

$$\mathfrak{q}_{i,j}([\delta_i]) = [(\delta_i \delta_j)^{-1} \delta_i(\delta_i \delta_j)] = [\delta_j^{-1} \delta_i \delta_j] \quad \text{and} \quad \mathfrak{q}_{i,j}([\delta_j]) = [(\delta_i \delta_j)^{-1} \delta_j(\delta_i \delta_j)].$$

- 2. If  $l \neq i$  and  $l \neq j$ , we have  $\mathfrak{q}_{i,j}([\delta_l]) = [\delta_l]$ .
- 3. The double braiding  $\mathbf{q}_{i,j}$  acts trivially on the homotopy classes of  $\alpha_1, \ldots, \alpha_g$ and  $\beta_1, \ldots, \beta_g$  in  $\pi_1(\Sigma_{g,n}, x)$ .

**Proof.** We have already seen in Paragraph 1.7 that the Dehn twists  $\mathbf{d}_i$  and  $\mathbf{d}_j$  act trivially, so that the action of  $\mathbf{q}_{i,j}$  coincides with the action of  $\mathbf{d}_{i,j}$ . If l = i or l = j, then  $\gamma_{i,j}$  and  $\delta_l$  intersect only in x, and the intersection is described by the first picture at the end of Paragraph 1.6. Therefore, the discussion of the first case there yields that  $\mathbf{d}_{i,j}([\delta_l]) = [\gamma_{i,j}^{-1} \delta_l \gamma_{i,j}]$ . Because  $\gamma_{i,j}$  is homotopic to  $\delta_i \delta_j$  relative to the base point x, this proves the first assertion.

If l < i or l > j, the generator  $\delta_l$  also intersects  $\gamma_{i,j}$  only in x, but now the intersection is described by the second picture at the end of Paragraph 1.6, so that  $\gamma'_{i,j}$  does not intersect  $\delta_l$ , which implies the second assertion in these cases. The case i < l < j is more complicated and is illustrated by the picture



in which we have replaced the curves  $\delta_i$ ,  $\delta_l$ , and  $\delta_j$  by slight modifications within their relative homotopy class. It shows that the curve  $\mathfrak{d}_{i,j}(\delta_l)$  comes about as follows: It starts at x with  $\delta_l$ , but then soon meets  $\gamma'_{i,j}$ . At that point, it turns left and follows  $\gamma'_{i,j}$ against its orientation. Upon returning, it follows  $\delta_l$  again, but soon encounters  $\gamma'_{i,j}$ a second time. Again, it turns left, but now follows  $\gamma'_{i,j}$  in the direction of its orientation. Up to relative homotopy, these two encounters with  $\gamma'_{i,j}$  cancel each other. While continuing along  $\delta_l$ , the curve has two similar encounters with  $\gamma'_{i,j}$ , which again cancel each other. In summary, the curve is homotopic to  $\delta_l$ , which finishes the proof of the second assertion.

The curves  $\alpha_i$  and  $\beta_i$  intersect  $\gamma_{i,j}$  again only in x, and the intersection is again described by the second picture at the end of Paragraph 1.6, so that  $\gamma'_{i,j}$  does not intersect them at all. This yields the third assertion.

#### 1.9 The capping homomorphism

In the process of defining  $\Sigma_{g,n}$ , we have removed *n* open disks from  $\Sigma_g$ . If we assume that our surface is realized via the polygon model, the boundary components come

with a natural enumeration, and the *j*-th boundary component is parametrized by  $\rho_j$ . In the place of the *j*-th removed disk, we now glue a punctured disk back, i.e., a disk whose interior contains a marked point denoted by *y*. If a homeomorphism of  $\Sigma_{g,n}$ restricts to the identity on the image of  $\rho_j$ , we can extend it to  $\Sigma_{g,n-1}$  by requiring that the extension restricts to the identity on the newly inserted punctured disk. In this way, we obtain a group homeomorphism

$$C_j \colon \Gamma_{g,n}(\rho_j) \to \Gamma_{g,n-1}(y)$$

which we call the *j*-th capping homomorphism. Here, we have not only written  $\Gamma_{g,n-1}(y)$  for  $\Gamma_{g,n-1}(\{y\})$ , as indicated in Paragraph 1.3, but also briefly  $\Gamma_{g,n}(\rho_j)$  for  $\Gamma_{g,n}(\operatorname{Im}(\rho_j))$ .

It is a consequence of the Alexander lemma (cf. [FM, Sec. 2.2, Lem. 2.1, p. 47f]) that  $\mathfrak{d}_j$  is contained in the kernel of  $C_j$ . The following result, which can be found in [FM, Sec. 3.6, Prop. 3.19, p. 85] or [Ko, Sec. 3, p. 104f], states that  $\mathfrak{d}_j$  in fact generates the kernel:

**Proposition 1.8.** The sequence

$$1 \longrightarrow \langle \mathfrak{d}_j \rangle \longrightarrow \Gamma_{g,n}(\rho_j) \xrightarrow{C_j} \Gamma_{g,n-1}(y) \longrightarrow 1$$

is short exact.

Almost always,  $\mathfrak{d}_j \in \Gamma_{g,n}(\rho_j)$  is a nontrivial mapping class, in which case  $\langle \mathfrak{d}_j \rangle$  is isomorphic to  $\mathbb{Z}$ . The only exception is the case g = 0 and n = 1: Here, it is easy to see that  $\Gamma_{0,0}(y)$  is trivial (cf. [FM, p. 49]), and  $\Gamma_{0,1}$  is trivial by the Alexander lemma.

Clearly, we can compose  $C_j$  with the forgetful map  $F_y: \Gamma_{g,n-1}(y) \to \Gamma_{g,n-1}$  to obtain a homomorphism  $D_j := F_y \circ C_j$  from  $\Gamma_{g,n}(\rho_j)$  to  $\Gamma_{g,n-1}$  that caps off the boundary component not with a punctured, but rather with a full disk. The effect of the additional mapping  $F_y$  will be studied in the next paragraph.

#### 1.10 The Birman sequence

Because we can always modify a diffeomorphism by isotopy so that it fixes a point, the forgetful map

$$F_x \colon \Gamma_{g,n}(x) \to \Gamma_{g,n}$$

is surjective. If  $\phi \in \text{Diffeo}^+(\Sigma_{g,n})$  not only fixes x, but in addition represents a mapping class in the kernel of  $F_x$ , it can be connected to the identity in  $\text{Diffeo}^+(\Sigma_{g,n})$ , i.e., there is a path

$$[0,1] \to \text{Diffeo}^+(\Sigma_{g,n}), \ t \mapsto \phi_t$$

with  $\phi_0 = \text{id}$  and  $\phi_1 = \phi$ . Since  $\phi_t$  is not required to fix the point x, we get a closed path

$$\gamma \colon [0,1] \to \Sigma_{g,n}, \ t \mapsto \phi_t(x)$$

based at x, which represents a homotopy class in the fundamental group  $\pi_1(\Sigma_{q,n}, x)$ .

Using the long exact sequence of a fibration (cf. [Ro1, Thm. 11.48, p. 358]), one can show that the homotopy class of  $\gamma$  determines the mapping class of  $\phi$ . More precisely, there is group antihomomorphism

$$P_x \colon \pi_1(\Sigma_{g,n}, x) \to \Gamma_{g,n}(x)$$

that maps the class of  $\gamma$  to the class of  $\phi$ . This map is called the point-pushing map, or briefly the pushing map. The arising exact sequence

$$\pi_1(\Sigma_{g,n}, x) \xrightarrow{P_x} \Gamma_{g,n}(x) \xrightarrow{F_x} \Gamma_{g,n} \longrightarrow 1,$$

is called the Birman sequence. If the Euler characteristic of  $\Sigma_{g,n}$  is strictly negative, the Birman sequence is in fact short exact, i.e.,  $P_x$  is injective. The Birman sequence is discussed in greater detail in [FM, Sec. 4.2, p. 96ff].

As we have explained in Paragraph 1.6, the mapping class group  $\Gamma_{g,n}(x)$  acts on the fundamental group  $\pi_1(\Sigma_{g,n}, x)$ , and so in particular  $P_x([\gamma])$  acts on the fundamental group. If  $P_x([\gamma]) = [\phi]$  and  $[\beta] \in \pi_1(\Sigma_{g,n}, x)$ , it can be shown that  $\phi(\beta)$ , i.e., the closed path  $t \mapsto \phi(\beta(t))$ , is homotopic to  $\gamma^{-1}\beta\gamma$  relative to the basepoint x, so that

$$P_x([\gamma])([\beta]) = [\gamma^{-1}\beta\gamma].$$

In other words,  $P_x([\gamma])$  acts on the fundamental group by an inner automorphism. The Birman sequence therefore implies that we have a homomorphism from  $\Gamma_{g,n}$  to  $Out(\pi_1(\Sigma_{g,n}, x))$  that makes the following diagram commutative:

$$\begin{array}{ccc} \Gamma_{g,n}(x) & \longrightarrow & \operatorname{Aut}(\pi_1(\Sigma_{g,n}, x)) \\ F_x & & \downarrow \\ F_{g,n} & & \downarrow \\ & & & & \\ \end{array}$$

Further aspects of this diagram are discussed in [FM, Par. 8.2.7, p. 235].

The action of the mapping class group  $\Gamma_{g,n}(x)$  on the fundamental group  $\pi_1(\Sigma_{g,n}, x)$  is compatible with the pushing map in another way: Clearly,  $\Gamma_{g,n}(x)$  acts on the kernel of the group homomorphism  $F_x$  by conjugation. This action is compatible with the pushing map in the sense that

$$P_x([\psi(\gamma)]) = [\psi]P_x([\gamma])[\psi]^{-1}$$

for  $[\psi] \in \Gamma_{g,n}(x)$  (cf. [FM, Par. 4.2.2, Fact 4.8, p. 99]). This relation is a direct consequence of our description above.

If a homotopy class in the fundamental group can be represented by a simple closed curve  $\gamma$ , there is a formula for its image under the pushing map in terms of Dehn twists: Using the notation from Paragraph 1.6, we have

$$P_x([\gamma]) = [\mathfrak{d}_{\gamma'}\mathfrak{d}_{\gamma''}^{-1}]$$

(cf. [FM, Par. 4.2.2, Fact 4.7, p. 99]).

#### 1.11 Singular homology

If we think of a closed curve as being defined on the interval [0, 1], it can be considered as a singular cycle. In this way, we obtain a group homomorphism

$$\pi_1(\Sigma_{g,n}, x) \to H_1(\Sigma_{g,n}, \mathbb{Z})$$

from the fundamental group to the first singular homology group that is called the (first) Hurewicz homomorphism. Hurewicz' theorem asserts in this situation that the Hurewicz homomorphism is surjective and induces an isomorphism from the commutator factor group of the fundamental group to the first singular homology group (cf. [Ro1, Thm. 4.29, p. 83]).

It follows from its universal property that the commutator factor group of a finitely presented group is presented by the same relations, but now understood as a presentation of an abelian group. Therefore, Hurewicz' theorem and the presentations of the fundamental group discussed in Paragraph 1.2 together imply that  $H_1(\Sigma_g, \mathbb{Z})$  is a free abelian group of rank 2g with the homology classes of the curves  $\alpha_1, \beta_1, \alpha_2, \beta_2, \ldots, \alpha_g, \beta_g$  as a basis. In the presence of boundary components, Hurewicz' theorem yields that  $H_1(\Sigma_{g,n}, \mathbb{Z})$  is generated by the homology classes of the curves  $\alpha_1, \beta_1, \alpha_2, \beta_2, \ldots, \alpha_g, \beta_g, \delta_1, \ldots, \delta_n$  subject to the relation that the homology classes of the curves  $\delta_1, \ldots, \delta_n$  sum up to zero. By solving for one of these generators, we see that the first homology group is free abelian of rank 2g + n - 1. We note that the homology class of  $\delta_j$  is equal to the homology class of  $\partial_j$ , because these curves are freely homotopic.

The algebraic intersection number introduced in Paragraph 1.6 depends only on the homology classes of the curves involved. It therefore defines a symplectic form on the abelian group  $H_1(\Sigma_{g,n}, \mathbb{Z})$ . For the generators, we have on the one hand

$$i_A(\beta_i, \alpha_j) = -i_A(\alpha_j, \beta_i) = \delta_{i,j}$$
 and  $i_A(\alpha_i, \alpha_j) = 0 = i_A(\beta_i, \beta_j)$ 

(cf. [FM, Par. 6.1.2, p. 165]), while on the other hand we have  $i_A(\delta_j, \gamma) = 0$  for every closed curve  $\gamma$ , because its free homotopy class has a representative that does not intersect  $\delta_j$ .

Because the outer automorphism group of any group evidently acts on the corresponding commutator factor group, Hurewicz' theorem implies that the commutative square obtained in Paragraph 1.10 can be enlarged to the diagram

$$\begin{split} & \Gamma_{g,n}(x) \longrightarrow \operatorname{Aut}(\pi_1(\Sigma_{g,n}, x)) \\ & F_x \downarrow & \downarrow \\ & \Gamma_{g,n} \longrightarrow \operatorname{Out}(\pi_1(\Sigma_{g,n}, x)) \\ & \parallel & \downarrow \\ & \Gamma_{g,n} \longrightarrow \operatorname{Aut}(H_1(\Sigma_{g,n}, \mathbb{Z})) \end{split}$$

It is immediate from the construction that the action of  $\Gamma_{g,n}$  on  $H_1(\Sigma_{g,n})$  preserves the intersection form and therefore takes values in the symplectic group. The discussion in Paragraph 1.7 and Paragraph 1.8 also implies how the special mapping classes introduced there act on the first homology group. The mapping classes  $[\mathfrak{d}_j]$  and the mapping classes  $[\mathfrak{q}_{i,j}]$  act as the identity on the entire first homology group. From Proposition 1.6, we see that the mapping classes  $[\mathfrak{t}_i]$ ,  $[\mathfrak{r}_i]$ , and  $[\mathfrak{s}_i]$  preserve the two subgroups generated by the homology classes of the curves  $\alpha_1, \beta_1, \alpha_2, \beta_2, \ldots, \alpha_g, \beta_g$ on the one hand and the curves  $\delta_1, \ldots, \delta_n$  on the other hand, and act as the identity on the second subgroup. As an abelian group, the first homology group is the direct sum of these two subgroups, and the first subgroup is free of rank 2g on the given generating set. We can therefore represent the action on the first subgroup by matrices in  $\operatorname{Sp}(2g,\mathbb{Z})$ . It follows from Proposition 1.6 that these matrix representations are block-diagonal with g blocks of  $2 \times 2$ -matrices. For  $\mathfrak{t}_i$ ,  $\mathfrak{r}_i$ , and  $\mathfrak{s}_i$ , the blocks corresponding to the generators  $\alpha_i$  and  $\beta_i$  are

$$\mathfrak{t} := \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix} \qquad \mathfrak{r} := \begin{pmatrix} 1 & 0 \\ -1 & 1 \end{pmatrix} \qquad \mathfrak{s} := \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}$$

respectively, while the other blocks are  $2 \times 2$ -identity matrices. For  $i = 1, \ldots, g - 1$ , the matrix representation of  $\mathbf{n}_i$  is also block-diagonal, but now contains a  $4 \times 4$ -block

$$\mathfrak{n} := \begin{pmatrix} 1 & 1 & 0 & -1 \\ 0 & 1 & 0 & 0 \\ 0 & -1 & 1 & 1 \\ 0 & 0 & 0 & 1 \end{pmatrix}$$

corresponding to the homology classes of the curves  $\alpha_i$ ,  $\beta_i$ ,  $\alpha_{i+1}$ , and  $\beta_{i+1}$ .

#### 1.12 The mapping class group of the torus

The surface  $\Sigma_1$  is a torus. Because the defining relation of the fundamental group discussed in Paragraph 1.2 in this case states that the relative homotopy class of  $\alpha_1\beta_1\alpha_1^{-1}\beta_1^{-1}$  is trivial, the fundamental group is abelian here, and therefore Hurewicz' theorem mentioned in Paragraph 1.11 yields that it is isomorphic to the first singular homology group. According to the Dehn-Lickorish theorem 1.4, the Dehn twists  $[\mathfrak{t}_1]$  and  $[\mathfrak{r}_1]$ , or alternatively  $[\mathfrak{t}_1]$  and  $[\mathfrak{s}_1] = [\mathfrak{t}_1^{-1}\mathfrak{r}_1^{-1}\mathfrak{t}_1^{-1}]$ , generate the mapping class group  $\Gamma_1$ . As we have just seen, the actions of these generators on the first homology group are, respectively, represented by the matrices

$$\mathfrak{t} = \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix} \qquad \mathfrak{r} = \begin{pmatrix} 1 & 0 \\ -1 & 1 \end{pmatrix} \qquad \mathfrak{s} = \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}$$

with respect to the basis consisting of the singular homology classes of  $\alpha_1$  and  $\beta_1$ . Because  $\operatorname{Sp}(2g, \mathbb{Z}) = \operatorname{SL}(2, \mathbb{Z})$  if g = 1, the group homomorphism  $\Gamma_{g,n} \to \operatorname{Sp}(2g, \mathbb{Z})$ described in Paragraph 1.11 becomes in this situation a group homomorphism

$$\Gamma_1 \to \mathrm{SL}(2,\mathbb{Z})$$

that maps  $[\mathfrak{t}_1]$ ,  $[\mathfrak{r}_1]$ , and  $[\mathfrak{s}_1]$  to  $\mathfrak{t}$ ,  $\mathfrak{r}$ , and  $\mathfrak{s}$ , respectively. It is a classical fact that the modular group  $\mathrm{SL}(2,\mathbb{Z})$  is generated by the matrices  $\mathfrak{s}$  and  $\mathfrak{t}$  and that the relations  $\mathfrak{s}^4 = 1$  and  $\mathfrak{s}\mathfrak{t}\mathfrak{s} = \mathfrak{t}^{-1}\mathfrak{s}\mathfrak{t}^{-1}$  that they satisfy are defining (cf. [KT, Thm. A.2, p. 312]).

These relations imply that  $\mathfrak{s}^2$  is central, which can also easily be seen directly. Alternatively, the modular group is generated by the elements  $\mathfrak{r}$  and  $\mathfrak{t}$ , and the defining relations for  $\mathfrak{s}$  and  $\mathfrak{t}$  just stated translate into the defining relations  $\mathfrak{tr}\mathfrak{t} = \mathfrak{r}\mathfrak{t}\mathfrak{r}$  and  $(\mathfrak{r}\mathfrak{t})^6 = 1$  for the generators  $\mathfrak{r}$  and  $\mathfrak{t}$  (cf. [SZ, Prop. 1.1, p. 7]). This implies immediately that our group homomorphism  $\Gamma_1 \to \mathrm{SL}(2,\mathbb{Z})$  is surjective, a fact that is also a special case of a more general result for mapping class groups (cf. [FM, Par. 6.3.2, Thm. 6.4, p. 170]). In the case of the torus, the homomorphism is even bijective (cf. [FM, Par. 2.2.4, Thm. 2.5, p. 53]), so that  $\Gamma_1 \cong \mathrm{SL}(2,\mathbb{Z})$ .

To discuss the mapping class group  $\Gamma_{1,1}$ , we need to introduce the braid group  $B_3$ on three strands, which we define as the group generated by two generators r and tsubject to the one defining relation rtr = trt. We will refer to this relation as the braid relation. Geometrically, r can be interpreted as the interchange of the first two strands, while t can be interpreted as the interchange of the last two strands (cf. [FM, Sec. 9.2, p. 246f]). In view of the second presentation of the modular group given above, there is a surjective group homomorphism

$$B_3 \to \mathrm{SL}(2,\mathbb{Z})$$

that maps r to  $\mathfrak{r}$  and t to  $\mathfrak{t}$ . If we define  $s := (trt)^{-1}$ , then on the one hand s is mapped to  $\mathfrak{s}$  under our homomorphism, and on the other hand the braid relation is equivalent to the relation  $sts = t^{-1}st^{-1}$  for the generators s and t of  $B_3$ . The braid relation implies relatively easily that the element  $(rt)^3 = (rtr)(trt)$  is central in  $B_3$ . This yields, again in view of the second presentation of the modular group, not only that  $(rt)^6$  is contained in the kernel of this homomorphism, but that it even generates the kernel. We therefore have the short exact sequence

$$1 \longrightarrow \langle (rt)^6 \rangle \longrightarrow B_3 \longrightarrow \mathrm{SL}(2,\mathbb{Z}) \longrightarrow 1.$$

We want to compare this homomorphism from  $B_3$  to  $SL(2,\mathbb{Z})$  with the homomorphism  $D_1: \Gamma_{1,1} \to \Gamma_1$  that caps off the one boundary component with a full disk. As discussed in Paragraph 1.9,  $D_1$  is the composition of  $C_1: \Gamma_{1,1} \to \Gamma_{1,0}(y)$ , which caps off the boundary component with a punctured disk that contains a marked point y in its interior, and the forgetful map

$$F_y \colon \Gamma_{1,0}(y) \to \Gamma_{1,0} = \Gamma_1.$$

At this point, there is a small conceptual difficulty with the polygon model. In the polygon model, the capped-off boundary component is parametrized by  $\rho_1$ , and the

point y in its interior is different from the base point x that is the common image of all the vertices of the polygon that are different from the start point and the end point of the edge labeled by  $\rho_1$ . To account for this difference, we choose a diffeomorphism  $\phi: \Sigma_{1,0} \to \Sigma_{1,0}$  that is isotopic to the identity and satisfies  $\phi(x) = y$ . We then have the isomorphism

$$\Gamma_{1,0}(x) \to \Gamma_{1,0}(y), \ [\psi] \mapsto [\phi \circ \psi \circ \phi^{-1}]$$

that satisfies  $F_y([\phi \circ \psi \circ \phi^{-1}]) = F_x([\psi])$ . Moreover, the map

$$\pi_1(\Sigma_{1,0}, x) \to \pi_1(\Sigma_{1,0}, y), \ [\gamma] \mapsto [\phi \circ \gamma]$$

is an isomorphism of fundamental groups. Because the curves  $\gamma$  and  $\phi \circ \gamma$  are freely homotopic, they map to the same singular homology class under the respective Hurewicz maps. If  $\gamma$  is a smooth simple closed curve, we have

$$[\mathfrak{d}_{\phi\circ\gamma}] = [\phi][\mathfrak{d}_{\gamma}][\phi^{-1}] = [\mathfrak{d}_{\gamma}]$$

in  $\Gamma_1$ . These considerations show that the diagram

$$\begin{array}{ccc} \Gamma_{1,0}(y) & \longrightarrow \operatorname{Aut}(\pi_1(\Sigma_{1,0},y)) & \longrightarrow \operatorname{Aut}(H_1(\Sigma_{1,0},\mathbb{Z})) \cong \operatorname{SL}(2,\mathbb{Z}) \\ F_y & & & & \\ \Gamma_{1,0} & & & & \\ \end{array}$$

$$\begin{array}{cccc} F_y & & & & \\ \Gamma_{1,0} & & & & \\ \end{array}$$

$$\begin{array}{cccccc} \operatorname{Aut}(H_1(\Sigma_{1,0},\mathbb{Z})) \cong \operatorname{SL}(2,\mathbb{Z}) \end{array}$$

commutes. Although it does not prove this fact, this diagram suggests that the map  $\Gamma_{1,0}(y) \to \operatorname{Aut}(H_1(\Sigma_{1,0},\mathbb{Z}))$  in the top row might be an isomorphism, and this indeed turns out to be the case (cf. [FM, Par. 2.2.4, p. 54f]). Therefore the forgetful map  $F_y$  must be bijective in our present case. For the kernel of the homomorphism  $D_1 = F_y \circ C_1$ , we therefore have  $\operatorname{Ker}(D_1) = \operatorname{Ker}(C_1) = \langle \mathfrak{d}_1 \rangle$  by Proposition 1.8. For the Birman sequence

$$\pi_1(\Sigma_1, y) \xrightarrow{P_y} \Gamma_1(y) \xrightarrow{F_y} \Gamma_1 \longrightarrow 1$$

discussed in Paragraph 1.10, this implies that the pushing map  $P_y$  vanishes identically, and is in particular not injective. This does not contradict the claims made there, because the Euler characteristic  $\chi(\Sigma_1) = 0$  is not strictly negative.

If we compose the isomorphism  $\Gamma_{1,0}(y) \to \operatorname{Aut}(H_1(\Sigma_{1,0},\mathbb{Z}))$  just discussed with the capping map  $C_1: \Gamma_{1,1} \to \Gamma_{1,0}(y)$ , we obtain a homomorphism from  $\Gamma_{1,1}$  to Aut $(H_1(\Sigma_{1,0},\mathbb{Z}))$ . This map can also be constructed in another way. As discussed in Paragraph 1.2, the fundamental group  $\pi_1(\Sigma_{1,1}, x)$  of a torus with one boundary component is a free (nonabelian) group whose generators are the relative homotopy classes  $[\alpha_1]$  and  $[\beta_1]$ . But by Hurewicz' theorem, we then have that the first homology group  $H_1(\Sigma_{1,1},\mathbb{Z})$  is again free abelian with the homology classes of  $\alpha_1$  and  $\beta_1$  as generators. This shows that the inclusion map  $\Sigma_{1,1} \to \Sigma_1$  induces an isomorphism between the first homology groups, so that also  $H_1(\Sigma_{1,1},\mathbb{Z}) \cong SL(2,\mathbb{Z})$  with respect to these generators. In view of our considerations above, the Dehn twists  $\mathfrak{t}_1$  and  $\mathfrak{r}_1$ in  $\Gamma_{1,1}$  are then represented by the same matrices as the corresponding Dehn twists in  $\Gamma_1$ .

As we said at the beginning of Paragraph 1.7, the Dehn twists  $\mathfrak{r}_1$  and  $\mathfrak{t}_1$  satisfy the braid relation, so that we obtain a group homomorphism

$$B_3 \to \Gamma_{1,1}$$

that maps r to  $\mathfrak{r}_1$  and t to  $\mathfrak{t}_1$ . By the Dehn-Lickorish theorem 1.4, this homomorphism is surjective. Now the 2-chain relation (cf. [FM, Par. 4.4.1, p. 107f]) states that  $(\mathfrak{r}_1\mathfrak{t}_1)^6 = \mathfrak{d}_1$  (cf. [S, Par. 3.4.1, Fig. 141, p. 124] for an illustration). We therefore have the commutative diagram

in which the rightmost vertical map is the isomorphism described above. The fivelemma now implies that  $\Gamma_{1,1} \cong B_3$  (cf. [FM, Par. 3.6.4, p. 87f]).

#### 1.13 The mapping class group of the sphere

In the preceding paragraph, we have introduced the braid group on three strands. More generally, there is, for  $n \ge 2$ , a braid group  $B_n$  on n strands, which can be defined as the group generated by elements  $\sigma_1, \ldots, \sigma_{n-1}$  subject to the relations

$$\sigma_i \sigma_{i+1} \sigma_i = \sigma_{i+1} \sigma_i \sigma_{i+1}$$

for i = 1, ..., n-2 and  $\sigma_i \sigma_j = \sigma_j \sigma_i$  if j > i+1. Geometrically,  $\sigma_i$  can be interpreted as the interchange of the *i*-th and the i + 1-st strand (cf. [FM, Sec. 9.2, p. 246f]). In the case n = 3 considered in the preceding paragraph, we have  $r = \sigma_1$  and  $t = \sigma_2$ .

From this definition, we see that there is a group homomorphism from the braid group  $B_n$  to the symmetric group  $S_n$  that maps  $\sigma_i$  to the transposition of i and i + 1. Pulling the natural action of  $S_n$  back along this homomorphism, we get an action of  $B_n$  on the set  $\{1, 2, \ldots, n\}$  that we denote by  $i \mapsto \sigma.i$ . If we view  $\mathbb{Z}^n$  as the set of functions from  $\{1, 2, \ldots, n\}$  to  $\mathbb{Z}$ , we can form the corresponding wreath product, which we denote by  $\mathbb{Z}^n \rtimes B_n$ . If  $E_i = (e_i, 1_{B_n})$  denotes the *i*-th canonical basis vector, considered as an element of the wreath product, and if we identify  $\sigma \in B_n$ with  $(0, \sigma) \in \mathbb{Z}^n \rtimes B_n$ , we have the commutation relation  $\sigma E_i = E_{\sigma.i}\sigma$ . The wreath product is therefore generated by the elements  $\sigma_1, \ldots, \sigma_{n-1}$  together with  $E_1, \ldots, E_n$ . Besides the defining relations of the braid group stated above, these generators satisfy the relations  $E_i E_j = E_j E_i$  as well as

$$\sigma_i E_j = \begin{cases} E_{i+1} \sigma_i & : j = i \\ E_i \sigma_i & : j = i+1 \\ E_j \sigma_i & : j \neq i \text{ and } j \neq i+1 \end{cases}$$

It is not too complicated to show that these relations are defining.

It turns out that the mapping class group  $\Gamma_{0,n}$  is a quotient group of this wreath product. If we map  $\sigma_i$  to  $\mathfrak{b}_{i,i+1}$  and  $E_i$  to  $\mathfrak{d}_i$ , we obtain a surjective group homomorphism from  $\mathbb{Z}^n \rtimes B_n$  to  $\Gamma_{0,n}$ . However, the map is not bijective, because the generators  $\mathfrak{b}_{i,i+1}$  and  $\mathfrak{d}_i$  satisfy additional relations: Besides the relations

$$\mathfrak{b}_{i,i+1}\mathfrak{b}_{i+1,i+2}\mathfrak{b}_{i,i+1} = \mathfrak{b}_{i+1,i+2}\mathfrak{b}_{i,i+1}\mathfrak{b}_{i+1,i+2}$$

for i = 1, ..., n-2 and  $\mathfrak{b}_{i,i+1}\mathfrak{b}_{j,j+1} = \mathfrak{b}_{j,j+1}\mathfrak{b}_{i,i+1}$  if j > i+1 for the braidings as well as  $\mathfrak{d}_i\mathfrak{d}_j = \mathfrak{d}_j\mathfrak{d}_i$  and

$$\mathfrak{b}_{i,i+1}\mathfrak{d}_j = \begin{cases} \mathfrak{d}_{i+1}\mathfrak{b}_{i,i+1} & :j=i\\ \mathfrak{d}_i\mathfrak{b}_{i,i+1} & :j=i+1\\ \mathfrak{d}_j\mathfrak{b}_{i,i+1} & :j\neq i \text{ and } j\neq i+1 \end{cases}$$

for the Dehn twists and their interaction with the braidings, we have the additional relations

$$\mathfrak{b}_{1,2}\mathfrak{b}_{2,3}\cdots\mathfrak{b}_{n-1,n}^2\cdots\mathfrak{b}_{2,3}\mathfrak{b}_{1,2}=\mathfrak{d}_1^{-2}$$

and  $(\mathfrak{b}_{1,2}\mathfrak{b}_{2,3}\cdots\mathfrak{b}_{n-1,n})^n = \mathfrak{d}_1^{-1}\mathfrak{d}_2^{-1}\cdots\mathfrak{d}_n^{-1}$ . These relations all together are again defining (cf. [L1, Par. 4.2, p. 487], see also [L2, Par. 3.1, p. 322f]).

Clearly, we can embed the wreath product  $\mathbb{Z}^{n-1} \rtimes B_{n-1}$  into the wreath product  $\mathbb{Z}^n \rtimes B_n$  by sending  $\sigma_i$  to  $\sigma_i$  and  $E_i$  to  $E_i$ . By composing this injection with the surjection just described, we get a group homomorphism from  $\mathbb{Z}^{n-1} \rtimes B_{n-1}$  to  $\Gamma_{0,n}$ . It clearly takes values in the subgroup  $\Gamma_{0,n}(\rho_n)$  that fixes the last boundary component pointwise. The fundamental fact about this map is the following:

**Proposition 1.9.** The group homomorphism

$$\mathbb{Z}^{n-1} \rtimes B_{n-1} \to \Gamma_{0,n}(\rho_n)$$

that maps  $\sigma_i$  to  $\mathfrak{b}_{i,i+1}$  and  $E_i$  to  $\mathfrak{d}_i$  is an isomorphism.

This proposition follows from the fact that we can view a sphere with one boundary component as a closed disk, say of radius 1. The mapping class group  $\Gamma_{0,n}(\rho_n)$  is therefore isomorphic to the mapping class group of a disk with n-1-holes in which the diffeomorphisms are required to restrict to the identity on the boundary unit circle. One of the standard topological descriptions of the braid group yields that this mapping class group is isomorphic to  $\mathbb{Z}^{n-1} \rtimes B_{n-1}$  in the indicated way (cf. [PS, § 7, p. 64], see also [FM, Par. 9.1.3, p. 243f]).

### 2 Tensor categories

#### 2.1 Finiteness

Let K be a perfect field and  $\mathcal{C}$  be an essentially small abelian K-linear category. Here,  $\mathcal{C}$  is called essentially small if there is a set, not a class, of objects with the property that every object in  $\mathcal{C}$  is isomorphic to an object in this set. We assume that  $\mathcal{C}$  is finite in the sense of [EGNO, Def. 1.8.6, p. 9], i.e., that it has finitedimensional spaces of morphisms, that every object has finite length, that it has enough projectives, and that there are only finitely many isomorphism classes of simple objects. That it has enough projectives means that for every object X, there is an epimorphism  $f: P \to X$  from a projective object P (cf. [Mi, Sec. II.14, p. 70]). A standard, but not completely trivial argument shows that, under the assumption on finite length, the epimorphism f can be chosen to be essential, where an epimorphism  $f: P \to X$  is called essential if, given a morphism  $g: Y \to P$ , we can conclude that g is an epimorphism if  $f \circ g$  is an epimorphism. An essential epimorphism  $f: P \to X$  from a projective object P is usually called a projective cover of X; it follows from the assumption on finite length that this definition is equivalent to the one given in [EGNO, Def. 1.6.6, p. 6].

As explained in [DSS, Prop. 1.4, p. 3], an essentially small abelian K-linear category is finite if and only if it is equivalent to the category of finite-dimensional modules over a finite-dimensional K-algebra (cf. also [EGNO, p. 9f]). We note that finite categories are called bounded categories in [KL].

We will assume that C is a strict tensor category in the sense of [Ka, Def. XI.2.1, p. 282], so that we have in particular a tensor product functor  $\otimes$  from  $C \times C$  to Cand a unit object 1. We note that tensor categories are called monoidal categories in [EGNO, Def. 2.2.8, p. 25]; the term 'tensor category' is used there for a category that satisfies additional restrictions (cf. [EGNO, Def. 4.1.1, p. 65]), all of which will be satisfied in our situation, as we start to discuss now: First, we require the tensor product to be *K*-bilinear. Second, we require that the unit object 1 is simple and that End(1) is one-dimensional over *K*.

In addition, we require that every object X has a left dual  $X^*$  (cf. [EGNO, Def. 2.10.1, p. 40]; [Ka, Def. XIV.2.1, p. 342]). The corresponding evaluation and coevaluation morphisms will be denoted by

$$\operatorname{ev}_X \colon X^* \otimes X \to \mathbb{1}$$
 and  $\operatorname{coev}_X \colon \mathbb{1} \to X \otimes X^*$ .

We also assume that C is braided with braiding

$$c_{X,Y} \colon X \otimes Y \to Y \otimes X$$

(cf. [EGNO, Def. 8.1.1, p. 195]; [Ka, Def. XIII.1.1, p. 315]) and that it carries a ribbon structure

$$\theta_X \colon X \to X$$

(cf. [EGNO, Def. 8.10.1, p. 216]; [Ka, Def. XIV.3.2, p. 349]), which by definition satisfies the equations

$$\theta_{X\otimes Y} = (\theta_X \otimes \theta_Y) \circ c_{Y,X} \circ c_{X,Y}$$

and  $\theta_{X^*} = \theta_X^*$ . Under these assumptions, every object X does not only have a left dual  $X^*$ , but also a right dual  $^*X$ , which is defined by using the same object  $^*X := X^*$ , but introducing the evaluation morphism

$$\operatorname{ev}_X' := \operatorname{ev}_X \circ c_{X,X^*} \circ (\theta_X \otimes \operatorname{id}_{X^*}) \colon X \otimes {}^*X \to \mathbb{1}$$

and the coevaluation morphism

$$\operatorname{coev}_X' := (\operatorname{id}_{X^*} \otimes \theta_X) \circ c_{X,X^*} \circ \operatorname{coev}_X \colon \mathbb{1} \to {}^*X \otimes X$$

(cf. [Ka, Prop. XIV.3.5, p. 352]). The axioms of left and right duality imply that for every object X, the functor  $X \otimes -$  has the left adjoint  $X^* \otimes -$  and the right adjoint  $^*X \otimes -$ , so that

 $\operatorname{Hom}(X^* \otimes Y, Z) \cong \operatorname{Hom}(Y, X \otimes Z)$  and  $\operatorname{Hom}(X \otimes Y, Z) \cong \operatorname{Hom}(Y, {}^*X \otimes Z).$ 

Similarly, the functor  $-\otimes X$  has the right adjoint  $-\otimes X^*$  and the left adjoint  $-\otimes^* X$ , so that

$$\operatorname{Hom}(Y \otimes X, Z) \cong \operatorname{Hom}(Y, Z \otimes X^*)$$
 and  $\operatorname{Hom}(Y \otimes {}^*X, Z) \cong \operatorname{Hom}(Y, Z \otimes X)$ 

(cf. [EGNO, Prop. 2.10.8, p. 42]; [Ka, Prop. XIV.2.2, p. 343f and p. 346f]). This implies that the functors  $X \otimes -$  and  $- \otimes X$  are exact (cf. [HS, Chap. II, Thm. 7.7, p. 68]; [P, Par. 2.7, Satz 3, p. 72]). This in turn implies that  $X \otimes P$  and  $P \otimes X$  are projective if P is projective (cf. [HS, Chap. II, Prop. 10.2, p. 82]). We will also need the following related result on adjunctions:

**Proposition 2.1.** Suppose that  $F: \mathcal{C} \to \mathcal{D}$  is a K-linear functor between finite K-linear categories. Then F has a right adjoint if and only if it is right exact.

**Proof.** If F has a right adjoint, it is right exact (cf. [HS, Chap. II, Thm. 7.7, p. 68]). As pointed out above, we can assume for the converse that C and D are the categories of finite-dimensional left modules over finite-dimensional algebras A and B, respectively. Then the Eilenberg-Watts theorem (cf. [Ro2, Thm. 5.45, p. 261]; [I, Thm. 2.4, p. 677]) states that F is naturally equivalent to the functor  $M \otimes_A -$  for a B-A-bimodule M. But by the standard adjunction (cf. [Ro2, Thm. 2.76, p. 93]), this functor has the adjoint  $\operatorname{Hom}_B(M, -)$ .

A different proof of this proposition that connects it to the adjoint functor theorem is given in [DSS, Cor. 1.9, p. 6].

An object X is called transparent if  $c_{Y,X} \circ c_{X,Y} = \mathrm{id}_{X \otimes Y}$  for all objects  $Y \in \mathcal{C}$  (cf. [B, p. 224]). Transparent objects are called central in [Mue, Rem. 2.10, p. 296]. The full subcategory consisting of all transparent objects is called the Müger center of  $\mathcal{C}$ .

The unit object is transparent (cf. [EGNO, Exerc. 8.1.6, p. 196]; [Ka, Prop. XIII.1.2, p. 316]), and the direct sum of two transparent objects is transparent. Therefore, an
object that is isomorphic to a finite direct sum of copies of the unit object is contained in the Müger center. We say that C is modular if the Müger center does not contain any other objects. In this definition, we do not require that C be semisimple. In the semisimple case, this condition is equivalent to the invertibility of the modular S-matrix (cf. [EGNO, Prop. 8.20.12, p. 243]; [Mue, Cor. 2.16, p. 297]), which is in this case usually taken as the definition of modularity (cf. [EGNO, Def. 8.13.4, p. 224]).

#### 2.2 Factorizable Hopf algebras

A fundamental example for categories with the properties that we have just listed are the representation categories of factorizable ribbon Hopf algebras. Suppose that A is a Hopf algebra with coproduct  $\Delta$ , counit  $\varepsilon$ , and antipode S. For the coproduct, we will use the sigma notation of R. Heyneman and M. Sweedler in the modified form  $\Delta(a) = a_{(1)} \otimes a_{(2)}$ . If we take for C the category of finite-dimensional left A-modules, it is not only a finite K-linear abelian category, but also a tensor category, where the tensor product of two A-modules X and Y becomes an A-module via

$$a.(x \otimes y) := a_{(1)}.x \otimes a_{(2)}.y$$

for  $x \in X, y \in Y$ , and  $a \in A$ . The unit object  $\mathbb{1}$  is the base field K, endowed with the trivial A-module structure coming from the counit. The left dual  $X^*$  of X is the dual vector space  $X^* := \operatorname{Hom}_K(X, K)$ , endowed with structures described in [Ka, Examp. XIV.2.1, p. 347]. Although this category is not strict, our considerations below apply to this category, because we have the notion of the tensor product  $X_1 \otimes X_2 \otimes \cdots \otimes X_n$  of n vector spaces or A-modules, which is not build as an iteration of the tensor product of two vector spaces and therefore does not depend on the insertion of parentheses. The tensor products that appear below have to be understood in this way.

We assume that A is quasitriangular with R-matrix  $R = R^1 \otimes R^2 \in A \otimes A$ , where the notation follows the spirit of the notation for the coproduct above and writes this tensor, which is in general not decomposable, as if it were decomposable (cf. [Ka, Def. VIII.2.1, p. 173]). From the R-matrix, we obtain a braiding on C via the formula

$$c_{X,Y}(x\otimes y) := R^2.y \otimes R^1.x$$

for  $x \in X$  and  $y \in Y$  (cf. [Ka, Prop. XIII.1.2, p. 318]). From the R-matrix, we also obtain the monodromy matrix Q := R'R, where  $R' := R^2 \otimes R^1$ . As in the case of

the R-matrix, we write  $Q = Q^1 \otimes Q^2$ . A is called factorizable if the map

 $\Phi\colon A^*\to A, \ \varphi\mapsto (\mathrm{id}_A\otimes\varphi)(Q)$ 

is bijective (cf. [SZ, Par. 3.2, p. 26] and the references given there). Note that this condition implies that A is finite-dimensional. We will require that A is factorizable, which is equivalent to the requirement that the category C is modular, as we will explain in Paragraph 2.3.

Furthermore, we assume that A contains a ribbon element, i.e., a nonzero central element v that satisfies

$$\Delta(v) = Q(v \otimes v) \quad \text{and} \quad S(v) = v$$

(cf. [SZ, Par. 4.3, p. 37]; note the difference to [Ka, Def. XIV.6.1, p. 361]). Our ribbon element gives rise to a ribbon structure of C by defining

$$\theta_X \colon X \to X, \ x \mapsto v.x$$

(cf. [Ka, Prop. XIV.6.2, p. 361]).

#### 2.3 Coends

If  $\mathcal{C}$  and  $\mathcal{D}$  are categories and  $F: \mathcal{C}^{\text{op}} \times \mathcal{C} \to \mathcal{D}$  is a bifunctor, a coend of F is an object L of  $\mathcal{D}$  together with a morphism  $\iota_X: F(X, X) \to L$  for every object  $X \in \mathcal{C}$  which is dinatural in the sense that, for every morphism  $f: X \to Y$ , the diagram

commutes (cf. [ML2, Chap. IX, Sec. 6, p. 226f]). Moreover, this dinatural transformation is required to be universal, which means that for another dinatural transformation  $\kappa_X \colon F(X, X) \to Z$ , there is a unique morphism  $g \colon L \to Z$  that makes for all  $X \in \mathcal{C}$  the diagram

$$F(X,X) \xrightarrow{\iota_X} L$$

$$\kappa_X \xrightarrow{\downarrow_g} Z$$

commutative.

From this universality, we see that coends behave well with respect to natural transformations:

**Lemma 2.2.** Suppose that  $F': \mathcal{C}^{\text{op}} \times \mathcal{C} \to \mathcal{D}$  is a second bifunctor with coend  $\iota'_X: F'(X, X) \to L'$ , and that  $\eta_{X,Y}: F(X, Y) \to F'(X, Y)$  is a natural transformation. Then there is a unique morphism  $g: L \to L'$  such that the diagram

$$\begin{array}{ccc} F(X,X) & \stackrel{\iota_X}{\longrightarrow} & L \\ \eta_{X,X} & & & \downarrow^g \\ F'(X,X) & \stackrel{\iota_X'}{\longrightarrow} & L' \end{array}$$

commutes for all objects  $X \in \mathcal{C}$ .

**Proof.** This lemma is the version of [ML2, Chap. IX, Sec. 7, Prop. 1, p. 228] for coends instead of ends.  $\Box$ 

When we speak of a coend without further specifications, we think of the case where  $\mathcal{C} = \mathcal{D}$  is our base category satisfying the requirements stated in Paragraph 2.1, and the bifunctor F is given by  $F(X, Y) = X^* \otimes Y$ . It is shown in [KL, Par. 5.1.3, p. 266ff] that, under our assumptions on  $\mathcal{C}$ , a coend for this bifunctor F exists.

It is not too difficult to see that a functor that possesses a right adjoint preserves coends. This together with the Fubini theorem for coends (cf. [ML2, Chap. IX, Sec. 8, p. 230f]) implies that the dinatural transformation

$$\iota_X \otimes \iota_Y \colon X^* \otimes X \otimes Y^* \otimes Y \to L \otimes L$$

is a coend. Comparing this transformation with the dinatural transformation

$$(\operatorname{ev}_X \otimes \operatorname{ev}_Y) \circ (\operatorname{id}_{X^*} \otimes (c_{Y^*,X} \circ c_{X,Y^*}) \otimes \operatorname{id}_Y) \colon X^* \otimes X \otimes Y^* \otimes Y \to \mathbb{1},$$

the universal property yields a morphism  $\omega_L \colon L \otimes L \to \mathbb{1}$  that, via the adjunctions stated in Paragraph 2.1, determines homomorphisms  $\omega'_L \colon L \to L^*$  and  $\omega''_L \colon L \to {}^*L$ , which are duals of each other. It is shown in [Sh1, Thm. 1.1, p. 3] that the modularity of  $\mathcal{C}$  is equivalent to the property that  $\omega'_L$ , or alternatively  $\omega''_L$ , is an isomorphism. This property is used in [KL, Def. 5.2.7, p. 276] as the definition of modularity.

The same method can be used to introduce other morphisms that will be needed in the sequel. The easiest of these morphisms arises from the dinatural transformation  $\iota_X \circ (\operatorname{id}_{X^*} \otimes \theta_X) \colon X^* \otimes X \to L$ . Applying the universal property of the coend yields a morphism  $\mathfrak{T} \colon L \to L$  that makes all diagrams of the form

$$\begin{array}{ccc} X^* \otimes X & \stackrel{\iota_X}{\longrightarrow} & L \\ & & & \downarrow_{\mathcal{I}} \\ \iota_X \circ (\operatorname{id}_{X^*} \otimes \theta_X) & & \downarrow_{\mathcal{I}} \\ & & & L \end{array}$$

commutative. Similarly, the universal property of  $L \otimes L$  explained above can be used to obtain a morphism  $\mathfrak{N}': L \otimes L \to L \otimes L$  that makes all diagrams of the form

$$\begin{array}{cccc} X^* \otimes X \otimes Y^* \otimes Y & \xrightarrow{\iota_X \otimes \iota_Y} & L \otimes L \\ & & & & \downarrow \mathfrak{n}' \\ & & & & & \downarrow \mathfrak{n}' \\ & & & & & L \otimes L \end{array}$$

commutative. Both of these morphisms are used when defining  $\mathfrak{N} := \mathfrak{N}' \circ (\mathfrak{T} \otimes \mathfrak{T})$ . An in a sense hybrid form of the dinatural transformations used in the definition of  $\omega_L$  and of  $\mathfrak{N}'$  is used in the definition of  $\mathfrak{S}'$  from  $L \otimes L$  to L: It arises by applying the universal property of  $L \otimes L$  to a dinatural transformation in such a way that the diagram

$$(\operatorname{ev}_X \otimes \iota_Y) \circ (\operatorname{id}_{X^*} \otimes (c_{Y^*,X} \circ c_{X,Y^*}) \otimes \operatorname{id}_Y) \xrightarrow{\iota_X \otimes \iota_Y} L \otimes L$$

becomes commutative.

Given an object  $Z \in \mathcal{C}$ , we define another morphism that is similar to  $\mathfrak{N}'$ : Again, because a functor that possesses a right adjoint preserves coends, the dinatural transformation  $\operatorname{id}_Z \otimes \iota_X : Z \otimes X^* \otimes X \to Z \otimes L$  is a coend. Its universal property yields a morphism  $\mathfrak{N}_{Z,L}^l: Z \otimes L \to Z \otimes L$  that makes all diagrams of the form

$$\begin{array}{cccc} Z \otimes X^* \otimes X & \xrightarrow{\operatorname{id}_Z \otimes \iota_X} & Z \otimes L \\ (\operatorname{id}_Z \otimes \iota_X) \circ ((c_{X^*, Z} \circ c_{Z, X^*}) \otimes \operatorname{id}_X) & & \downarrow \mathfrak{N}_{Z, L}^l \\ & & Z \otimes L \end{array}$$

commutative.

The coend L is a Hopf algebra inside the category C. We will not need the arising product, coproduct, unit, counit, and antipode, which are constructed in a similar way as the morphisms  $\omega_L$ ,  $\mathfrak{T}$ ,  $\mathfrak{N}'$ , and  $\mathfrak{S}'$  by using the universal property of a coend and are described in [V, Par. 1.6, p. 478f] (cf. also [KL, Par. 5.2.2, p. 271ff]; note that the conventions are slightly different there). We will, however, need that, as a consequence of the Hopf algebra structure, there are two-sided integrals  $\Lambda_L \colon \mathbb{1} \to L$ and  $\lambda_L \colon L \to \mathbb{1}$ . Here, the assumption on modularity is used for two points, namely on the one hand for the fact that the unit object  $\mathbb{1}$  can indeed be used as the domain of  $\Lambda_L$  and the codomain of  $\lambda_L$ , and on the other hand for the fact that  $\lambda_L$  is two-sided (cf. [KL, Sec. 5.2, p. 270ff]). The integral  $\Lambda_L$  is used to define  $\mathfrak{S} \in \text{End}(L)$  as the composition

$$L \xrightarrow{\cong} L \otimes \mathbb{1} \xrightarrow{\operatorname{id}_L \otimes \Lambda_L} L \otimes L \xrightarrow{\mathfrak{S}'} L$$

(cf. [L1, Par. 1.3, p. 473]; the setting there is slightly more general as the modularity hypothesis is weakened).

## 2.4 Coends from Hopf algebras

In the case where C is the category of finite-dimensional left modules over a factorizable ribbon Hopf algebra A, the coend can be described explicitly: The dual vector space  $L := A^*$ , viewed as a left A-module via the left coadjoint action

$$(a.\varphi)(a') := \varphi(S(a_{(1)})a'a_{(2)})$$

for  $a, a' \in A$  and  $\varphi \in A^*$ , becomes a coend when endowed with the dinatural transformation

$$\iota_X(\xi \otimes x)(a) = \xi(a.x)$$

for a left A-module X and elements  $\xi \in X^*$ ,  $x \in X$ , and  $a \in A$  (cf. [V, Lem. 4.3, p. 498], see also [KL, Thm. 7.4.13, p. 331]). It is not difficult to find the explicit form of the morphisms introduced in Paragraph 2.3 in this model of the coend:

**Proposition 2.3.** Suppose that  $\varphi, \psi \in A^*$  and that  $a, a' \in A$ . Then we have

- 1.  $\mathfrak{T}(\varphi)(a) = \varphi(av)$
- 2.  $\mathfrak{N}'(\varphi \otimes \psi)(a \otimes a') = \varphi(aQ_1)\psi(S(Q_2)a')$
- 3.  $\mathfrak{S}'(\varphi \otimes \psi)(a) = \varphi(Q_1)\psi(S(Q_2)a)$

4. 
$$\omega_L(\varphi \otimes \psi) = \varphi(Q^1)\psi(S(Q^2))$$

**Proof.** We prove these assertions by showing that the respective right-hand side indeed has the required universal property. For the first assertion, this holds since we have for an A-module X and  $\xi \in X^*, x \in X$  that

$$((\iota_X \circ (\mathrm{id}_{X^*} \otimes \theta_X))(\xi \otimes x))(a) = \iota_X(\xi \otimes v.x)(a) = \xi(av.x) = \mathfrak{T}(\iota_X(\xi \otimes x))(a).$$

The proof of the second assertion is similar: If Y is another A-module and  $\zeta \in Y^*$ ,  $y \in X$  are elements, we have

$$([(\iota_X \otimes \iota_Y) \circ (\mathrm{id}_{X^*} \otimes (c_{Y^*,X} \circ c_{X,Y^*}) \otimes \mathrm{id}_Y)](\xi \otimes x \otimes \zeta \otimes y))(a \otimes a')$$
  
=  $((\iota_X \otimes \iota_Y)(\xi \otimes Q_1.x \otimes Q_2.\zeta \otimes y))(a \otimes a') = \xi(hQ_1.x)(Q_2.\zeta)(a'.y)$   
=  $\xi(hQ_1.x)\zeta(S(Q_2)a'.y) = \mathfrak{N}'(\iota_X(\xi \otimes x) \otimes \iota_Y(\zeta \otimes y))(a \otimes a').$ 

The proofs of the third and the fourth assertion are again very similar.

Because the antipode of a finite-dimensional Hopf algebra is bijective (cf. [Mo, Thm. 2.1.3, p. 18]), the fourth assertion shows that  $\omega_L$  is nondegenerate if and only if the map  $\Phi$  introduced in Paragraph 2.2 is bijective. This shows that the modularity of C is equivalent to the factorizability of A (cf. [KL, Par. 7.4.6, p. 332]).

We note that, because these maps are defined via the universal property of the coend, they must automatically be morphisms in C, i.e., they must be A-linear. We also note that in the second assertion, we have chosen a special identification of  $A^* \otimes A^*$ with  $(A \otimes A)^*$ , which is determined by the equation  $(\varphi \otimes \psi)(a \otimes a') = \varphi(a)\psi(a')$ .

It is important to stress that the Hopf algebra structure on L is not the usual Hopf algebra structure on  $A^*$ . In fact,  $A^*$  is a Hopf algebra in the category of vector spaces, whereas L is a Hopf algebra in the category C, which has a different braiding. The Hopf algebra structure on L is rather dual to the so-called transmuted Hopf algebra structure of A (cf. [Maj, Examp. 9.4.9, p. 504]; [V, Lem. 4.4, p. 499]). However, there is a very direct relation between the integrals of  $A^*$  and L: Because a factorizable Hopf algebra is unimodular (cf. [L1, Prop. 3.7.4, p. 482] and [Ra, Prop. 12.4.2, p. 405f]), there is a nonzero two-sided integral  $\Lambda_A \in A$ . This leads to the two-sided integral

$$\lambda_L \colon L \to \mathbb{1} = K, \ \varphi \mapsto \varphi(\Lambda_A).$$

On the other hand, a right integral  $\rho: A \to K$  is by definition contained in  $L = A^*$ , and it can be shown that the unique morphism  $\Lambda_L: \mathbb{1} = K \to L$  that maps  $\mathbb{1}_K$  to  $\rho$ is even a two-sided integral (cf. [BKLT, Prop. 6.6, p. 153]). As a consequence, the morphism  $\mathfrak{S}$  is given by the formula

$$\mathfrak{S}(\varphi)(a) = \varphi(Q_1)\rho(S(Q_2)a).$$

#### 2.5 The block spaces

In topological field theory, one associates to a modular category projective representations of mapping class groups of surfaces. There is a vast literature on this topic; our approach here is based on a construction given by V. Lyubashenko in his articles [L1] and [L2]. We will now briefly review some key aspects of his construction to the extent that we need them. Suppose that C is a category that satisfies the requirements stated in Paragraph 2.1. We assume that each of the *n* boundary components of the surface  $\Sigma_{g,n}$  described in Paragraph 1.1 is labeled by an object  $X_i$ from C. Associated with such a labeled surface is the space

$$\mathcal{Z}(\Sigma_{q,n}^{X_1,\ldots,X_n}) := \operatorname{Hom}_{\mathcal{C}}(X_n \otimes \cdots \otimes X_1, L^{\otimes g}),$$

where L denotes the coend defined in Paragraph 2.3. This space, which we call the space of chiral conformal blocks or briefly the block space, obviously depends functorially on the labels and can therefore be viewed as the value of a left exact contravariant functor  $\mathcal{Z}(\Sigma_{g,n})$  on the object  $(X_1, \ldots, X_n) \in \mathcal{C}^n$ , by which we mean that the functor is left exact in each argument (cf. [P, Par. 4.6, p. 130f]).

The construction of a projective action of the mapping class group then starts from an oriented net (cf. [L2, Sec. 6, p. 355] and [L1, Par. 4.1, p. 486f]). We consider the following oriented net that encodes the structure of our surface:



The block space arises from the oriented net by taking suitable coends over internal edges (cf. [L2, Par. 8.2, p. 374]). For our chosen net, the construction gives the space

$$T_{1} := \int^{A_{i},N_{j},Z_{l},Y_{m}} \operatorname{Hom}(A_{1} \otimes Y_{2},N_{1}) \otimes \operatorname{Hom}(Y_{3},Y_{2} \otimes N_{2}) \otimes \cdots \otimes \operatorname{Hom}(Y_{g},Y_{g-1} \otimes N_{g-1}) \\ \otimes \operatorname{Hom}(X_{n} \otimes Z_{n-1},Y_{g}) \otimes \operatorname{Hom}(X_{n-1} \otimes Z_{n-2},Z_{n-1}) \otimes \cdots \otimes \operatorname{Hom}(X_{2} \otimes Z_{1},Z_{2}) \\ \otimes \operatorname{Hom}(X_{1},Z_{1} \otimes A_{q}) \otimes \operatorname{Hom}(N_{q-1} \otimes A_{q},A_{q-1}) \otimes \cdots \otimes \operatorname{Hom}(N_{1} \otimes A_{2},A_{1})$$

where in the upper limit of the integral sign the abbreviation  $A_i$  has been used for  $A_1, A_2, \ldots, A_g$ . Similarly, the abbreviations  $N_j$ ,  $Z_l$ , and  $Y_m$  have been used for the objects with indices  $j = 1, \ldots, g - 1$ ,  $l = 1, \ldots, n - 1$ , and  $m = 2, \ldots, g$ . The coends are taken in the category of left exact functors: Obviously, the argument of the coend depends functorially on the objects, and the corresponding functor is left exact. A more detailed discussion of this aspect can be found in [L2, App. B, p. 398] and [FS, Sec. 3, p. 72ff].

In view of [L2, Lem. B.1, p. 398] and the functor adjunctions discussed in Paragraph 2.1, the space  $T_1$  is naturally isomorphic to

$$T_{2} := \int^{A_{i}, N_{j}, Y_{2}, Y_{g}, Z_{1}} \operatorname{Hom}(Y_{2}, {}^{*}A_{1} \otimes N_{1}) \otimes \operatorname{Hom}(Y_{g}, Y_{2} \otimes N_{2} \otimes \cdots \otimes N_{g-1})$$
  
$$\otimes \operatorname{Hom}(X_{n} \otimes \cdots \otimes X_{2} \otimes Z_{1}, Y_{g}) \otimes \operatorname{Hom}(X_{1} \otimes {}^{*}A_{g}, Z_{1})$$
  
$$\otimes \operatorname{Hom}(N_{g-1}, A_{g-1} \otimes A_{g}^{*}) \otimes \cdots \otimes \operatorname{Hom}(N_{1}, A_{1} \otimes A_{2}^{*})$$

Taking also the coend over  $Y_g$ , we obtain a natural isomorphism with the space

$$T_{3} := \int^{A_{i}, N_{j}, Z_{1}, Y_{2}} \operatorname{Hom}(Y_{2}, {}^{*}A_{1} \otimes N_{1})$$
  
 
$$\otimes \operatorname{Hom}(X_{n} \otimes \cdots \otimes X_{2} \otimes Z_{1}, Y_{2} \otimes N_{2} \otimes \cdots \otimes N_{g-1}) \otimes \operatorname{Hom}(X_{1} \otimes {}^{*}A_{g}, Z_{1})$$
  
 
$$\otimes \operatorname{Hom}(N_{g-1}, A_{g-1} \otimes A_{g}^{*}) \otimes \cdots \otimes \operatorname{Hom}(N_{1}, A_{1} \otimes A_{2}^{*})$$

Treating  $N_1, \ldots, N_{g-1}$  in the same way and moreover taking the coend over  $Z_1$ , we obtain the space

$$T_4 := \int^{A_i, Y_2} \operatorname{Hom}(Y_2, {}^*A_1 \otimes A_1 \otimes A_2^*) \\ \otimes \operatorname{Hom}(X_n \otimes \cdots \otimes X_1 \otimes {}^*A_g, Y_2 \otimes A_2 \otimes A_3^* \otimes \cdots \otimes A_{g-1} \otimes A_g^*)$$

Finally, if we also take the coend over  $Y_2$  and apply the functor adjunctions to  $A_g$ , we arrive at the space

$$T_5 := \int^{A_i} \operatorname{Hom}(X_n \otimes \cdots \otimes X_1, {}^*A_1 \otimes A_1 \otimes A_2^* \otimes A_2 \otimes \cdots \otimes A_{g-1} \otimes A_g^* \otimes A_g)$$

It is now important to realize that we have introduced right duals in Paragraph 2.1 in such a way that left and right duals have the same underlying object. They

differ only in their evaluation and coevaluation morphisms, which affect the functor adjunctions, but not the objects themselves. The space  $T_5$  is therefore also equal to

$$T_5 = \int^{A_i} \operatorname{Hom}(X_n \otimes \cdots \otimes X_1, A_1^* \otimes A_1 \otimes \cdots \otimes A_g^* \otimes A_g)$$

Because the coends are taken in the category of left exact functors, the space  $T_5$  is naturally isomorphic to our block space  $\operatorname{Hom}(X_n \otimes \cdots \otimes X_1, L^{\otimes g})$  defined above (cf. [FS, Prop. 3.4, p. 74]).

We have claimed that the specific oriented net given above encodes the structure of the surface introduced in Paragraph 1.1. We now explain this relation in the special case g = 3 and n = 2; the general case is not essentially different. The first step consists in the application of the fattening functor (cf. [L2, p. 341]), which turns the net into a so-called ribbon graph by replacing each trivalent vertex by a hexagon in which every second side is represented by a double line. These hexagons are arranged so that their double lines come to lie on the edges of the net, and if an edge connects two vertices, the corresponding hexagons are glued along their now common double line. If we apply this procedure to our net above, we obtain the ribbon graph



in which the colored lines indicate the glued double lines.

The second step consists in the application of the duplication functor (cf. [L2, Par. 2.2 and Par. 2.3, p. 315ff]). The duplication functor turns a ribbon graph into a surface by taking two copies of the graph and gluing them along the boundary components that do not arise from the double lines mentioned above. Applied to our ribbon graph, it yields a surface of genus 3 that has two boundary components, which arise from the two double lines still present in our ribbon graph and are labeled by  $X_1$ 

and  $X_2$ . If we deform the arising surface slightly and bring the first and the third handle closer together by pulling the middle handle to the bottom, we obtain the standard surface



described in Paragraph 1.1, in which the curves arise from the duplication of the lines in the ribbon graph that have the same color. It should be noted that the red curves are freely homotopic to the curves denoted by  $\alpha_1$ ,  $\alpha_2$ , and  $\alpha_3$  in Paragraph 1.1, and the orange curves are freely homotopic to the curves  $\mu_1$  and  $\mu_2$  from Paragraph 1.2, respectively.

It is possible to compare this discussion with the corresponding one in [L1, Par. 4.5, p. 494f]. To do that, it is necessary to deform the surface given in [L1, Fig. 6, p. 495] by pulling down the middle handles to the bottom in order to bring the first and the g-th handle together at the top, exactly as in our discussion of the relation between the net and the surface above. This leads to a surface in which the handles are labeled counterclockwise, but the boundary components are labeled clockwise, in contrast to the surface used in Paragraph 1.1, where the boundary components arose from the polygon and therefore were also labeled counterclockwise. As a consequence, the order of the labels  $X_1, \ldots, X_n$  is reversed. The following table explains the relation between our surface and the surface in [L1, Fig. 2, p. 490 and Fig. 6, p. 495]:

Curves	$\alpha_i$	$\beta_i$	$\mu_j$	$\zeta_l$	$\partial_k$	
Color	red	blue	orange	turquoise	purple	
Label in the net	$A_i$		$N_j$	$Z_l$	$X_k$	
Dehn twists	$\mathfrak{t}_i$	$\mathfrak{r}_i$	$\mathfrak{r}_i$ $\mathfrak{n}_j$		$\mathfrak{d}_k$	
[L1]	$e_i$	$\overline{b}_i$	$a_{j+1}$	$t_{n-l,g}$	$R_{g-k+1}$	

Here, the index ranges are  $i = 1, \ldots, g$ ,  $j = 1, \ldots, g - 1$ ,  $k = 1, \ldots, n$ , and  $l = 1, \ldots, n - 1$ . The light green curves in our surface correspond to the curves labeled  $d_j$  in [L1, loc. cit.].

# 2.6 Mapping class group representations

As we just explained, the theory associates with a surface a certain left exact functor. But in addition, the theory associates with a mapping class  $[\psi] \in \Gamma_{g,n}$  a projective class  $[\mathcal{Z}(\psi)] = P(\mathcal{Z}(\psi))$  of natural equivalences between two of these functors, namely the functor

$$(X_1,\ldots,X_n)\mapsto \mathcal{Z}(\Sigma_{q,n}^{X_1,\ldots,X_n})$$

on the one hand and the functor

$$(X_1,\ldots,X_n)\mapsto \mathcal{Z}(\Sigma_{g,n}^{X_{\tau(1)},\ldots,X_{\tau(n)}})$$

on the other hand, where  $\tau := p([\psi])^{-1} \in S_n$  is the inverse permutation of the marked points introduced in Paragraph 1.3. A representative  $\mathcal{Z}(\psi)$  of this projective class is a natural equivalence between these functors, i.e., a family of linear isomorphisms

$$\mathcal{Z}(\psi)\colon \mathcal{Z}(\Sigma_{g,n}^{X_1,\dots,X_n}) \to \mathcal{Z}(\Sigma_{g,n}^{X_{\tau(1)},\dots,X_{\tau(n)}})$$

that are natural in the following sense: If, for i = 1, ..., n, we are given morphisms  $f_i: X_i \to Y_i$ , then the diagram

commutes. Clearly, a nonzero scalar multiple of such a natural equivalence is again a natural equivalence, and the set of all its nonzero scalar multiples constitutes its projective class  $[\mathcal{Z}(\psi)] = P(\mathcal{Z}(\psi))$ .

This assignment is also compatible with composition in the sense that, for a second mapping class  $[\phi]$ , we have  $[\mathcal{Z}(\phi \circ \psi)] = [\mathcal{Z}(\phi) \circ \mathcal{Z}(\psi)]$ . However, this does not imply that the mapping class group  $\Gamma_{g,n}$  acts projectively on each block space  $\mathcal{Z}(\Sigma_{g,n}^{X_1,\ldots,X_n})$ , because the block spaces are not preserved by  $\mathcal{Z}(\psi)$  if  $[\psi]$  permutes two boundary components with different labels. In general, only the pure mapping class group  $\Gamma_{g,n}$ acts projectively on each block space. However, the entire mapping class group  $\Gamma_{g,n}$ acts projectively on the direct sum

$$\bigoplus_{\tau \in S_n} \mathcal{Z}(\Sigma_{g,n}^{X_{\tau(1)},\dots,X_{\tau(n)}})$$

of block spaces. In the case where all boundary components are labeled with the same object X, we also have an action of  $\Gamma_{g,n}$  on the space  $\operatorname{Hom}_{\mathcal{C}}(X^{\otimes n}, L^{\otimes g})$ . We can in fact realize the direct sum above as a subspace of this space by setting  $X := X_1 \oplus \cdots \oplus X_n$ .

Via a sophisticated system of rules laid out in [L1] and [L2], the description of a surface in terms of a net makes it possible to describe the actions of the elements of the mapping class group on the block spaces explicitly. According to this formalism, the elements of the mapping class group introduced in Paragraph 1.4, Paragraph 1.5, Paragraph 1.7, and Paragraph 1.8 act on the block spaces as follows:

(1) For j = 1, ..., n, the Dehn twist  $\mathfrak{d}_j$  acts by precomposition with the morphism  $\mathrm{id}_{X_n \otimes \cdots \otimes X_{j+1}} \otimes \theta_{X_j} \otimes \mathrm{id}_{X_{j-1} \otimes \cdots \otimes X_1}$ , i.e., via the map

$$\mathcal{Z}(\mathfrak{d}_j) \colon \operatorname{Hom}_{\mathcal{C}}(X_n \otimes \cdots \otimes X_1, L^{\otimes g}) \to \operatorname{Hom}_{\mathcal{C}}(X_n \otimes \cdots \otimes X_1, L^{\otimes g})$$
$$f \mapsto f \circ (\operatorname{id}_{X_n \otimes \cdots \otimes X_{j+1}} \otimes \theta_{X_j} \otimes \operatorname{id}_{X_{j-1} \otimes \cdots \otimes X_1}).$$

(2) For j = 1, ..., n - 1, the braiding  $\mathfrak{b}_{j,j+1}$  acts by precomposition with the morphism  $\mathrm{id}_{X_n \otimes \cdots \otimes X_{j+2}} \otimes c_{X_j, X_{j+1}} \otimes \mathrm{id}_{X_{j-1} \otimes \cdots \otimes X_1}$ , i.e., it induces a map

$$\mathcal{Z}(\mathfrak{b}_{j,j+1})\colon \operatorname{Hom}_{\mathcal{C}}(X_n \otimes \cdots \otimes X_{j+1} \otimes X_j \otimes \cdots \otimes X_1, L^{\otimes g}) \to \operatorname{Hom}_{\mathcal{C}}(X_n \otimes \cdots \otimes X_j \otimes X_{j+1} \otimes \cdots \otimes X_1, L^{\otimes g}) f \mapsto f \circ (\operatorname{id}_{X_n \otimes \cdots \otimes X_{j+2}} \otimes c_{X_j, X_{j+1}} \otimes \operatorname{id}_{X_{j-1} \otimes \cdots \otimes X_1}).$$

(3) For j = 1, ..., n-1, the element  $\mathfrak{d}_{j,j+1}$  acts by precomposition with the morphism  $\mathrm{id}_{X_n \otimes \cdots \otimes X_{j+2}} \otimes \theta_{X_{j+1} \otimes X_j} \otimes \mathrm{id}_{X_{j-1} \otimes \cdots \otimes X_1}$ , a map that we denote by  $\mathcal{Z}(\mathfrak{d}_{j,j+1})$ .

(4) For j = 1, ..., n-1, the element  $\mathfrak{q}_{j,j+1}$  acts by precomposition with the morphism  $\mathrm{id}_{X_n \otimes \cdots \otimes X_{j+2}} \otimes (c_{X_j, X_{j+1}} \circ c_{X_{j+1}, X_j}) \otimes \mathrm{id}_{X_{j-1} \otimes \cdots \otimes X_1}$ , a map that we denote by  $\mathcal{Z}(\mathfrak{q}_{j,j+1})$ .

(5) For  $i = 1, \ldots, g$ , the Dehn twist  $\mathfrak{t}_i$  acts by postcomposition with the morphism  $\mathrm{id}_{L^{\otimes (i-1)}} \otimes \mathfrak{T} \otimes \mathrm{id}_{L^{\otimes (g-i)}}$ , i.e., via the map

$$\mathcal{Z}(\mathfrak{t}_{i})\colon \operatorname{Hom}_{\mathcal{C}}(X_{n}\otimes\cdots\otimes X_{1},L^{\otimes g})\to \operatorname{Hom}_{\mathcal{C}}(X_{n}\otimes\cdots\otimes X_{1},L^{\otimes g})$$
$$f\mapsto (\operatorname{id}_{L^{\otimes (i-1)}}\otimes\mathfrak{T}\otimes \operatorname{id}_{L^{\otimes (g-i)}})\circ f.$$

(6) For  $i = 1, \ldots, g$ , the element  $\mathfrak{s}_i$  acts by postcomposition with the morphism  $\mathrm{id}_{L^{\otimes (i-1)}} \otimes \mathfrak{S} \otimes \mathrm{id}_{L^{\otimes (g-i)}}$ , a map that we denote by  $\mathcal{Z}(\mathfrak{s}_i)$ .

(7) For i = 1, ..., g-1, the Dehn twist  $\mathbf{n}_i$  acts by postcomposition with the morphism  $\mathrm{id}_{L^{\otimes (i-1)}} \otimes \mathfrak{N} \otimes \mathrm{id}_{L^{\otimes (g-i-1)}}$ , a map that we denote by  $\mathcal{Z}(\mathbf{n}_i)$ .

(8) For j = 1, ..., n - 1, the Dehn twist  $\mathfrak{z}_j$  acts via the map

$$\mathcal{Z}(\mathfrak{z}_j)$$
: Hom <sub>$\mathcal{C}$</sub>  $(X_n \otimes \cdots \otimes X_1, L^{\otimes g}) \to \operatorname{Hom}_{\mathcal{C}}(X_n \otimes \cdots \otimes X_1, L^{\otimes g})$ 

which is the unique linear map that makes the diagram

commutative. In this diagram, the horizontal arrows are the functor adjunctions from Paragraph 2.1, and the left vertical map is given by

$$h \mapsto (\theta_{*(X_n \otimes \dots \otimes X_{j+1}) \otimes L^{\otimes (g-1)}} \otimes \mathfrak{T}) \circ \mathfrak{N}_{*(X_n \otimes \dots \otimes X_{j+1}) \otimes L^{\otimes (g-1)}, L}^l \circ h.$$

We now explain how some of the formulas come about. For any edge of a trivalent net, there is an automorphism called the twist (cf. [L2, Par. 4.3, p. 350]) that corresponds to a Dehn twist around the corresponding curve on the surface (cf. [L2, Prop. 2.2, p. 319]). Recall that, as already mentioned in Paragraph 1.4, Dehn twists as defined here are inverse Dehn twists as defined in [L1] and [L2]. By [L2, Par. 8.1, No. (x), p. 374], the twist acts by applying the ribbon twist to the corresponding variable. Because the isomorphisms between the spaces  $T_1$  to  $T_5$  introduced in Paragraph 2.5 and our block space are natural, this implies Claim (1). Similarly, the generators  $\mathfrak{t}_i$ ,  $\mathfrak{n}_j$ , and  $\mathfrak{z}_l$  act by applying a ribbon twist to the internal variables  $A_i$ ,  $N_j$ , and  $Z_l$  (cf. [L1, p. 493]). We calculate the action of these generators explicitly in the case g = 3 and n = 2 to explain Claim (5), Claim (7), and Claim (8).

We first calculate the action of  $\mathfrak{t}_1$ : On the space  $T_1$ , it acts by postcomposition with the twist  $\theta_{A_1}$  on the last tensor factor, which corresponds to postcomposition with  $\theta_{A_1} \otimes \mathrm{id}_{A_2^*}$  on the last tensor factor on the spaces  $T_2$  and  $T_3$ . Under the isomorphism to the space  $T_4$ , this corresponds to postcomposition with  $\mathrm{id}_{*A_1} \otimes \theta_{A_1} \otimes \mathrm{id}_{A_2^*}$  on the first tensor factor. This in turn corresponds on the space  $T_5$  to postcomposition with  $\mathrm{id}_{*A_1} \otimes \theta_{A_1} \otimes \mathrm{id}_{A_2^* \otimes A_2 \otimes A_3^* \otimes A_3}$ . In view of the definition of  $\mathfrak{T}$  in Paragraph 2.3, this becomes postcomposition with  $\mathfrak{T} \otimes \mathrm{id}_{L \otimes L}$  on our block space  $\mathrm{Hom}(X_2 \otimes X_1, L \otimes L \otimes L)$ . Very similar calculations yield the claim for  $\mathfrak{t}_2$  and  $\mathfrak{t}_3$ .

Next, we calculate the action of  $\mathbf{n}_1$ : On the space  $T_1$ , it acts by postcomposition with the twist  $\theta_{N_1}$  on the first tensor factor. Under the isomorphism to the space  $T_2$ , this corresponds to postcomposition with  $\mathrm{id}_{*A_1} \otimes \theta_{N_1}$  on the first tensor factor, which does not change under the isomorphism to the space  $T_3$ . On the space  $T_4$ , it becomes postcomposition with  $\mathrm{id}_{*A_1} \otimes \theta_{A_1 \otimes A_2^*}$  on the first tensor factor. Under the isomorphism to the space  $T_5$ , this corresponds to postcomposition with  $\mathrm{id}_{*A_1} \otimes \theta_{A_1 \otimes A_2^*} \otimes \mathrm{id}_{A_2 \otimes A_3^*}$ . This is equal to postcomposition with  $\mathrm{id}_{*A_1} \otimes ((\theta_{A_1} \otimes \theta_{A_2^*}) \circ c_{A_2^*,A_1} \circ c_{A_1,A_2^*}) \otimes \mathrm{id}_{A_2 \otimes A_3^* \otimes A_3}$ , which corresponds on our block space  $\mathrm{Hom}(X_2 \otimes X_1, L \otimes L \otimes L)$  to postcomposition with  $\mathfrak{N} \otimes \mathrm{id}_L$ . The calculations for  $\mathbf{n}_2$  are similar.

For the action of  $\mathfrak{z}_1$ , we proceed differently: On the space  $T_1$ , it acts by postcomposition with  $\theta_{Z_1} \otimes \mathrm{id}_{A_3}$  on the tensor factor  $\mathrm{Hom}(X_1, Z_1 \otimes A_3)$ , which on the spaces  $T_2$ and  $T_3$  becomes postcomposition with  $\theta_{Z_1}$  on the fourth and the third tensor factor, respectively. Under the isomorphism to the space  $T_4$ , this corresponds to precomposition with  $\mathrm{id}_{X_2} \otimes \theta_{X_1 \otimes^* A_3}$ . This space is isomorphic to

$$\int^{A_i} \operatorname{Hom}(X_1 \otimes {}^*A_3, {}^*X_2 \otimes A_1^* \otimes A_1 \otimes A_2^* \otimes A_2 \otimes A_3^*),$$

where the action becomes precomposition with  $\theta_{X_1 \otimes^* A_3}$ , which equals postcomposition with  $\theta_{X_2 \otimes A_1^* \otimes A_1 \otimes A_2^* \otimes A_2 \otimes A_3^*}$ . Under the isomorphism to the space

$$\int^{A_i} \operatorname{Hom}(X_1, {}^*X_2 \otimes A_1^* \otimes A_1 \otimes A_2^* \otimes A_2 \otimes A_3^* \otimes A_3),$$

it becomes postcomposition with  $\theta_{X_2 \otimes A_1^* \otimes A_1 \otimes A_2^* \otimes A_2 \otimes A_3^*} \otimes id_{A_3}$ , which equals postcomposition with

$$\left(\left(\theta_{*X_2\otimes A_1^*\otimes A_1\otimes A_2^*\otimes A_2}\otimes \theta_{A_3^*}\right)\circ c_{A_3^*,*X_2\otimes A_1^*\otimes A_1\otimes A_2^*\otimes A_2}\circ c_{*X_2\otimes A_1^*\otimes A_1\otimes A_2^*\otimes A_2,A_3^*}\right)\otimes \mathrm{id}_{A_3}.$$

Under the isomorphism to the space Hom $(X_1, {}^*X_2 \otimes L^{\otimes 3})$ , this equals postcomposition with  $(\theta_{X_2 \otimes L \otimes L} \otimes \mathfrak{T}) \circ \mathfrak{N}^l_{X_2 \otimes L \otimes L, L}$ . But this means precisely that  $\mathcal{Z}(\mathfrak{z}_1)$  acts as asserted on our block space Hom $(X_2 \otimes X_1, L^{\otimes 3})$ .

Using a fusing morphism (cf. [L2, Eq. (6.8), p. 359]), the braiding of two neighboring boundary components corresponds to a braiding morphism of a trivalent net (cf. [L2, Par. 4.3, p. 350f]), which in turn corresponds to the braiding of the corresponding labels of the boundary components on our block space (cf. [L2, Par. 8.1, No. (viii) and (ix), p. 373f]). In this way, we arrive at Claim (2) and Claim (4). Claim (3) follows from Claim (1), because we have seen in Paragraph 1.8 that  $[\mathfrak{q}_{j,j+1}] = [\mathfrak{d}_{j,j+1}\mathfrak{d}_{j+1}^{-1}\mathfrak{d}_{j+1}^{-1}]$ , and in a ribbon category we have

$$c_{X_{j},X_{j+1}} \circ c_{X_{j+1},X_{j}} = (\theta_{X_{j+1}}^{-1} \otimes \theta_{X_{j}}^{-1}) \circ \theta_{X_{j+1} \otimes X_{j}} = \theta_{X_{j+1} \otimes X_{j}} \circ (\theta_{X_{j+1}}^{-1} \otimes \theta_{X_{j}}^{-1})$$

The element  $\mathfrak{s}_i$  corresponds to a switch (cf. [L2, Prop. 2.2, p. 319]) that acts by  $\mathfrak{S}$  on the corresponding tensor factor of  $L^{\otimes g}$  (cf. [L2, Prop. 8.8, p. 391]). This establishes Claim (6) and concludes our discussion about how the different elements of the mapping class group act.

It is important to note that, as we use a different net for the description of the surface, our formulas for the actions of the generators of the mapping class group on the block space are different from those in [L1, Par. 4.5, p. 494f]. Starting from the net in [L1, Fig. 5, p. 491] and using the internal variables called there  $E_i$  instead of those called there  $D_i$  to form the coends in the last step would give formulas similar to ours. To facilitate the comparison, we include a small table that relates our notation to the one used in [L1]:

Present notation	$\Gamma_{g,n}$	L	$\mathfrak{N}'$	$\mathfrak{N}^l_{X,L}$	$\mathfrak{S}$	T	$\mathfrak{b}_{l,l+1}$	$\mathfrak{q}_{l,l+1}$	$\mathfrak{s}_i$
[L1]	$M'_{g,n}$	f	Ω	$\Omega^l_{X,\mathbf{f}}$	S	T	$\omega_l$	$\omega_l^2$	$S_i$

It must be emphasized that the morphisms in the first row are only analogous, but not strictly equal, to the ones in the second row.

In the case where n = 1, i.e., when there is only one boundary component, the projective action on the block space can be obtained by postcomposition from a projective action of  $\Gamma_{g,1}$  on  $L^{\otimes g}$ : For a mapping class  $[\psi] \in \Gamma_{g,1}$ , a representative  $\mathcal{Z}(\psi)$  of the associated projective class  $[\mathcal{Z}(\psi)]$  is a natural equivalence from the contravariant Hom-functor

$$X \mapsto \mathcal{Z}(\Sigma_{q,1}^X) = \operatorname{Hom}_{\mathcal{C}}(X, L^{\otimes g})$$

to itself, which according to the Yoneda lemma (cf. [ML2, Chap. III, Sec. 2, p. 61]; [P, Par. 1.15, p. 37]) is given by postcomposition with an automorphism of  $L^{\otimes g}$ . For the generators of the mapping class group whose projective action is described in the list above, this can be seen explicitly: The only generator that appears in this case and is not already given by postcomposition is  $\mathfrak{d}_1$ , which according to Claim (1) is given by precomposition with  $\theta_X$ . However, by the naturality of the twist, this is equal to postcomposition with  $\theta_{L^{\otimes g}}$ .

#### 2.7 Modular functors

The various mapping class group representations just described are not unrelated, but rather together form a modular functor (cf. [L2, Par. 8.1, p. 372ff]). In particular, they are compatible with gluing of surfaces (cf. [L2, Par. 8.1, No. (xi), p. 374]). We will need only one very special instance of this general property: Consider the last two boundary components of the surface  $\Sigma_{g,n+2}$ . They carry an orientation that is induced from the orientation of the surface, which is opposite to the orientation of the curves  $\rho_{n+1}$  and  $\rho_{n+2}$  in Paragraph 1.1. Up to isotopy, there is a unique orientationreversing diffeomorphism between these two boundary components that maps the distinguished point on one boundary component to the distinguished point on the other. If we identify the two boundary components along this diffeomorphism, the arising quotient space is diffeomorphic to  $\Sigma_{g+1,n}$ , so that we have effectively attached a handle, which we consider as the first one. A diffeomorphism  $\varphi: \Sigma_{g,n+2} \to \Sigma_{g,n+2}$ that restricts to the identity on the last two boundary components then induces a diffeomorphism  $\psi: \Sigma_{g+1,n} \to \Sigma_{g+1,n}$ . For  $\tau := p([\varphi])^{-1} \in S_{n+2}$ , our hypothesis means that  $\tau$  fixes n + 1 and n + 2.

Suppose we are given n objects  $X_1, \ldots, X_n$  and another object X. From the adjunctions recalled in Paragraph 2.1, we get an isomorphism

$$\operatorname{Hom}_{\mathcal{C}}(X^* \otimes X^{**} \otimes X_n \otimes \cdots \otimes X_1, L^{\otimes g}) \to \operatorname{Hom}_{\mathcal{C}}(X_n \otimes \cdots \otimes X_1, X^* \otimes X \otimes L^{\otimes g}).$$

If we postcompose with  $\iota_X \otimes \mathrm{id}_{L^{\otimes g}}$ , we obtain a morphism

$$\operatorname{Hom}_{\mathcal{C}}(X^* \otimes X^{**} \otimes X_n \otimes \cdots \otimes X_1, L^{\otimes g}) \to \operatorname{Hom}_{\mathcal{C}}(X_n \otimes \cdots \otimes X_1, L^{\otimes (g+1)})$$

that we call the handle gluing homomorphism. By construction, this homomorphism is natural with respect to the objects  $X_1, \ldots, X_n$  of  $\mathcal{C}$  in the sense that for given morphisms  $f_i: X_i \to Y_i$ , the diagram

commutes.

The special case of the gluing property under consideration here then states that the diagram

$$\mathcal{Z}(\Sigma_{g,n+2}^{X_1,\dots,X_n,X^{**},X^*}) \longrightarrow \mathcal{Z}(\Sigma_{g+1,n}^{X_1,\dots,X_n}) \\
\downarrow^{\mathcal{Z}(\varphi)} \qquad \qquad \downarrow^{\mathcal{Z}(\psi)} \\
\mathcal{Z}(\Sigma_{g,n+2}^{X_{\tau(1)},\dots,X_{\tau(n)},X^{**},X^*}) \longrightarrow \mathcal{Z}(\Sigma_{g+1,n}^{X_{\tau(1)},\dots,X_{\tau(n)}})$$

commutes for suitably chosen representatives  $\mathcal{Z}(\varphi)$  and  $\mathcal{Z}(\psi)$  within their respective projective classes. As above,  $\mathcal{Z}(\varphi)$  and  $\mathcal{Z}(\psi)$  are viewed here as natural transformations of the corresponding functors, so that the nonzero scalar used in passing to a different representative does not depend on the spaces  $X_1, \ldots, X_n$ .

To illustrate this property, we consider the example  $\varphi = \mathfrak{d}_{n+1}$ , which also plays an important role later. Then the corresponding map on the glued surface is  $\psi = \mathfrak{t}_1$ . The actions of  $\varphi$  and  $\psi$  are described in the list in Paragraph 2.6. By its definition in Paragraph 2.3, we have  $\mathfrak{T} \circ \iota_X = \iota_X \circ (\mathrm{id}_{X^*} \otimes \theta_X) = \iota_X \circ (\theta_X^* \otimes \mathrm{id}_X)$ , where the second equality follows from the dinaturality of  $\iota$ . Therefore, the diagram

commutes.

Combining this with the naturality of the adjunction, we see that the gluing property holds in this case.

We note that the choice  $\varphi = \mathfrak{d}_{n+2}$  leads to the same map on the glued surface, namely  $\psi = \mathfrak{t}_1$ . A very similar reasoning shows that the gluing property also holds in this case.

#### 2.8 The case of the torus

In order to understand how the results of this article generalize the results of our previous one (cf. [LMSS1]), it will be important to consider the projective representation reviewed in Paragraph 2.6 in the case of the torus, where g = 1, and the case where C is the category of representations of a Hopf algebra A with the properties described in Paragraph 2.2. Important ingredients of this projective representation are the endomorphisms  $\mathfrak{S}$  and  $\mathfrak{T}$  of  $L = A^*$ , whose explicit form we have determined in Paragraph 2.4. In our previous article, we have used the same symbols for endomorphisms of A, which we will now denote by  $\hat{\mathfrak{S}}$  and  $\hat{\mathfrak{T}}$  and which are given by

$$\hat{\mathfrak{S}}(a) = \rho(aQ_1)S(Q_2)$$
 and  $\hat{\mathfrak{T}}(a) = va$ 

(cf. [LMSS1, Sec. 3, p. 410]). In order to understand how these endomorphisms are related, we use a variant of the Radford map

$$\iota \colon A \to A^*, \ a \mapsto \iota(a)$$

defined by  $\iota(a)(a') = \rho(aa')$ , which was also introduced on the page just cited. Here  $\rho$  is the right integral already considered in Paragraph 2.4. As recalled in [LMSS1], loc. cit.,  $\rho$  is contained in the space

$$\bar{C}(A) := \{ \varphi \in A^* \mid \varphi(aa') = \varphi(S^2(a')a) \text{ for all } a, a' \in A \},\$$

which can be viewed as the set of invariants  $(A^*)^A$  for the left coadjoint action discussed in Paragraph 2.4. The variant of the Radford map that we will need is the map  $\bar{\iota} := \iota \circ S^2$ , which then satisfies

$$\bar{\iota}(a)(a') = \iota(S^2(a))(a') = \rho(S^2(a)a') = \rho(a'a).$$

With the help of  $\bar{\iota}$ , we see that  $\hat{\mathfrak{S}}$  and  $\hat{\mathfrak{T}}$  are conjugate to  $\mathfrak{S}$  and  $\mathfrak{T}$ :

Lemma 2.4. The diagrams

$$\begin{array}{cccc} A^* & \stackrel{\mathfrak{S}}{\longrightarrow} & A^* & & & A^* & \stackrel{\mathfrak{T}}{\longrightarrow} & A^* \\ \overline{\iota} & & \uparrow_{\overline{\iota}} & & \text{and} & & \overline{\iota} & & \uparrow_{\overline{\iota}} \\ A & \stackrel{\mathfrak{S}}{\longrightarrow} & A & & & A & \stackrel{\mathfrak{T}}{\longrightarrow} & A \end{array}$$

commute. Moreover,  $\bar{\iota}$  is an isomorphism of A-modules, where  $A^*$  carries the left coadjoint action and A carries the left adjoint action of the coopposite Hopf algebra, given by  $a.a' := a_{(2)}a'S^{-1}(a_{(1)})$ .

**Proof.** To show the commutativity of the first diagram, we use that  $\rho \in \overline{C}(A)$  to compute

$$\bar{\iota}(\mathfrak{S}(a))(a') = \rho(aQ_1)\bar{\iota}(S(Q_2))(a') = \rho(aQ_1)\rho(a'S(Q_2)) = \rho(S^2(Q_1)a)\rho(S^3(Q_2)a') = \rho(Q_1a)\rho(S(Q_2)a') = \bar{\iota}(a)(Q_1)\rho(S(Q_2)a') = \mathfrak{S}(\bar{\iota}(a))(a').$$

The proof of the commutativity of the second diagram is even simpler: We have

$$\bar{\iota}(\hat{\mathfrak{T}}(a))(a') = \bar{\iota}(va)(a') = \rho(a'va) = \bar{\iota}(a)(a'v) = \mathfrak{T}(\bar{\iota}(a))(a')$$

by Proposition 2.3. Because  $\rho$  is a Frobenius homomorphism,  $\iota$  and  $\bar{\iota}$  are bijective. To see that  $\bar{\iota}$  is linear with respect to the specified A-actions, we compute

$$\bar{\iota}(a.a')(a'') = \rho(a''(a.a')) = \rho(a''a_{(2)}a'S^{-1}(a_{(1)})) = \rho(S(a_{(1)})a''a_{(2)}a')$$
$$= \bar{\iota}(a')(S(a_{(1)})a''a_{(2)}) = (a.(\bar{\iota}(a')))(a'')$$

and get the assertion.

By [SZ, Prop. 4.3, p. 37], we have  $\hat{\mathfrak{S}} \circ \hat{\mathfrak{T}} \circ \hat{\mathfrak{S}} = \rho(v) \, \hat{\mathfrak{T}}^{-1} \circ \hat{\mathfrak{S}} \circ \hat{\mathfrak{T}}^{-1}$ . The preceding lemma therefore implies that we also have  $\mathfrak{S} \circ \mathfrak{T} \circ \mathfrak{S} = \rho(v) \, \mathfrak{T}^{-1} \circ \mathfrak{S} \circ \mathfrak{T}^{-1}$ . It then follows from our discussion of the braid group  $B_3$  in Paragraph 1.12 that the assignment

$$s \mapsto \mathfrak{S}$$
 and  $t \mapsto \mathfrak{T}$ 

yields a projective representation of  $B_3$  in  $\operatorname{Aut}_A(L)$ . If X is an A-module, postcomposition with  $\mathfrak{S}$  and  $\mathfrak{T}$  therefore leads to a projective representation of  $B_3$  on the space  $\operatorname{Hom}_A(X, L)$ . On the other hand, we have seen in Paragraph 1.12 that the assignment  $r \mapsto \mathfrak{r}_1$  and  $t \mapsto \mathfrak{t}_1$  yields an isomorphism between the braid group  $B_3$ and the mapping class group  $\Gamma_{1,1}$ . By construction, this isomorphism maps s to  $\mathfrak{s}_1$ . Therefore, if we transport the projective action of  $B_3$  to  $\Gamma_{1,1}$  along this isomorphism, we obtain exactly the projective representation considered in Paragraph 2.6 in the case g = 1 and n = 1.

It is instructive to see why the 2-chain relation  $(\mathfrak{r}_1\mathfrak{t}_1)^6 = \mathfrak{d}_1$  in the mapping class group  $\Gamma_{1,1}$ , which we discussed in Paragraph 1.12, is satisfied in our situation. We have already used in Paragraph 1.12 that  $(\mathfrak{r}_1\mathfrak{t}_1)^6 = \mathfrak{s}_1^{-4}$ . Now suppose that  $f \in \operatorname{Hom}_A(X, L)$  is an A-linear map. For  $x \in X$ , we choose  $a \in A$  so that  $f(x) = \overline{\iota}(a)$ . For example from [LMSS1, Prop. 3.2, p. 411] combined with [LMSS1, Lem. 3.3, p. 412], we know that

$$\hat{\mathfrak{S}}^4(a) = ((\rho \otimes \rho)(Q))^2 S(v_{(1)}^{-1}) a v_{(2)}^{-1}.$$

From the lemma above and the fact that  $\rho \in \overline{C}(A)$ , we then get

$$\begin{split} ((\mathfrak{s}_{1}^{4}.f)(x))(a') &= \mathfrak{S}^{4}(f(x))(a') = \mathfrak{S}^{4}(\bar{\iota}(a))(a') = \bar{\iota}(\hat{\mathfrak{S}}^{4}(a))(a') = \rho(a'\hat{\mathfrak{S}}^{4}(a)) \\ &= ((\rho \otimes \rho)(Q))^{2} \ \rho(a'S(v_{(1)}^{-1})av_{(2)}^{-1}) \\ &= ((\rho \otimes \rho)(Q))^{2} \ \rho(S^{2}(v_{(2)}^{-1})a'S(v_{(1)}^{-1})a) \\ &= ((\rho \otimes \rho)(Q))^{2} \ f(x)(S^{2}(v_{(2)}^{-1})a'S(v_{(1)}^{-1})) \\ &= ((\rho \otimes \rho)(Q))^{2} \ (S(v^{-1}).f(x))(a') = ((\rho \otimes \rho)(Q))^{2} \ (v^{-1}.f(x))(a') \\ &= ((\rho \otimes \rho)(Q))^{2} \ f(v^{-1}.x)(a'). \end{split}$$

On the other hand, we have

$$((\mathfrak{d}_1^{-1}.f)(x))(a') = ((f \circ \theta_X^{-1})(x))(a') = f(v^{-1}.x)(a'),$$

so that the actions of  $\mathfrak{s}_1^4$  and  $\mathfrak{d}_1^{-1}$  agree up to a scalar, as required.

In the case g = 1, but n = 0, the definition in Paragraph 2.5 is to be understood in such a way that the action is on the block space  $\operatorname{Hom}_A(K, L)$ , because the base field is the unit object of the category, as explained in Paragraph 2.2. For any left *A*-module *Y*, the map  $\operatorname{Hom}_A(K, Y) \to Y^A$ ,  $f \mapsto f(1_K)$  yields a bijection with the space of invariants

$$Y^{A} = \{ y \in Y \mid a.y = \varepsilon(a)y \text{ for all } a \in A \}.$$

As we said above, in the case Y = L we have  $L^A = \overline{C}(A)$ , while in the case Y = A, endowed with the action described in Lemma 2.4, we have  $Y^A = Z(A)$ , the center of A. In view of its A-linearity,  $\overline{\iota}$  restricts to an isomorphism between Z(A) and  $\overline{C}(A)$ .

In the case X = K, we have that  $\theta_X$  is the identity map, because  $\varepsilon(v) = 1$ . This means that  $\mathfrak{d}_1$  acts trivially on  $\operatorname{Hom}_A(K, L)$ . By the 2-chain relation that we have just checked explicitly, this implies that  $\mathfrak{s}_1^4$  acts trivially. This in turn means, in view of the discussion in Paragraph 1.12, that the projective action of the braid group  $B_3$  restricts to a projective action of the modular group  $\operatorname{SL}(2,\mathbb{Z})$  on  $\overline{C}(A)$ , as it is required for the construction in Paragraph 2.6, because  $\Gamma_1 \cong \operatorname{SL}(2,\mathbb{Z})$ . Via  $\overline{\iota}$ , this projective action on  $\overline{C}(A)$  is isomorphic to an action of  $\operatorname{SL}(2,\mathbb{Z})$  on the center Z(A). This is the action that was considered in [LMSS1, Cor. 3.4, p. 412].

Let us mention that using  $\bar{\iota}$  is not the only way to relate the projective representations on  $\bar{C}(A)$  and Z(A). Other ways are discussed in [SZ, Par. 9.1, p. 87ff].

# **3** Derived functors

## 3.1 **Projective resolutions**

There are various approaches to the definition of the Ext-functors in abelian categories. We assume here that our category satisfies the assumptions listed in Paragraph 2.1; one of these assumptions was that it has enough projectives. In this case, we can use the approach via projective resolutions described in [Mi, Chap. VII, § 7, p. 182ff] and denoted there by  $\overline{\text{Ext}}$ , but here just denoted by  $\overline{\text{Ext}}$ . In this approach, the group  $\operatorname{Ext}^m(X, Y)$  for two objects X and Y of  $\mathcal{C}$  is defined by choosing a projective resolution

$$X \xleftarrow{\xi} P_0 \xleftarrow{d_1} P_1 \xleftarrow{d_2} P_2 \xleftarrow{d_3} \cdots$$

of X, which exists by the above hypothesis. The group  $\operatorname{Ext}^m(X,Y)$  is then defined as the *m*-cohomology group of the cochain complex

$$\operatorname{Hom}_{\mathcal{C}}(P_0, Y) \longrightarrow \operatorname{Hom}_{\mathcal{C}}(P_1, Y) \longrightarrow \operatorname{Hom}_{\mathcal{C}}(P_2, Y) \longrightarrow \cdots$$

of abelian groups. Although the notation does not reflect this, the definition depends on the chosen resolution; different resolutions lead to Ext-groups that are canonically isomorphic, but not equal (cf. [ML1, Chap. XII, § 9, p. 390]; [LMSS1, Sec. 1, p. 402]). Note that  $\text{Ext}^0(X, Y) \cong \text{Hom}_{\mathcal{C}}(X, Y)$ , because the Hom-functor is left exact.

The following lemma about the ribbon structure  $\theta$  will be important in the sequel:

Lemma 3.1. Suppose that

$$\mathbb{1} \xleftarrow{\xi} P_0 \xleftarrow{d_1} P_1 \xleftarrow{d_2} P_2 \xleftarrow{d_3} \cdots$$

is a projective resolution of the unit object. Then there exists, for every object  $X \in \mathcal{C}$ and every  $m \geq 0$ , a morphism  $h_m(X): P_m \otimes X \to P_{m+1} \otimes X$  such that

$$\theta_{P_m \otimes X} - \mathrm{id}_{P_m} \otimes \theta_X = (d_{m+1} \otimes \mathrm{id}_X) \circ h_m(X) + h_{m-1}(X) \circ (d_m \otimes \mathrm{id}_X)$$

for all  $m \ge 1$ , and  $\theta_{P_0 \otimes X} - \mathrm{id}_{P_0} \otimes \theta_X = (d_1 \otimes \mathrm{id}_X) \circ h_0(X)$ . These morphisms are natural in X in the sense that, for a morphism  $f: X \to Y$ , the diagram

$$\begin{array}{ccc} P_m \otimes X & \xrightarrow{\operatorname{id}_{P_m} \otimes f} & P_m \otimes Y \\ & & & \downarrow \\ h_m(X) \downarrow & & \downarrow \\ P_{m+1} \otimes X & \xrightarrow{} & I_{m+1} \otimes f \end{array} \\ P_{m+1} \otimes X & \xrightarrow{} & I_{m+1} \otimes Y \end{array}$$

commutes.

**Proof.** We denote by  $\operatorname{Rex}(\mathcal{C}, \mathcal{C})$  the category of K-linear right exact functors from  $\mathcal{C}$  to  $\mathcal{C}$ , whose morphisms are natural transformations. By [Sh2, Cor. 2.6, p. 466], a result that relies on our assumption that the base field is perfect, the functor  $X \mapsto P_m \otimes X$  is a projective object in the K-linear category  $\operatorname{Rex}(\mathcal{C}, \mathcal{C})$ , and therefore these functors form a projective resolution of the functor  $X \mapsto \mathbb{1} \otimes X \cong X$  in the category  $\operatorname{Rex}(\mathcal{C}, \mathcal{C})$ . The result now follows from the comparison theorem (cf. [Ro2, Thm. 6.16, p. 340]) applied to the category  $\operatorname{Rex}(\mathcal{C}, \mathcal{C})$ .

If  $\mathcal{C}$  is the category of finite-dimensional modules over a factorizable Hopf algebra discussed in Paragraph 2.2, the preceding proof can be given a more explicit form. By the Eilenberg-Watts theorem already mentioned in the proof of Proposition 2.1, right exact functors can in this case be represented by tensoring with bimodules. In particular, the functor  $X \mapsto P_m \otimes X$  is represented as  $X \mapsto (P_m \otimes A) \otimes_A X$ , where the space  $P_m \otimes A$  carries the A-bimodule structure

$$a.(p\otimes a').a'' = a_{(1)}.p\otimes a_{(2)}a'a''$$

for  $p \in P_m$  and  $a, a', a'' \in A$ . We then have two liftings of  $\theta_A$ , which appear in the diagram

$$A \xleftarrow{\xi \otimes \mathrm{id}_A} P_0 \otimes A \xleftarrow{d_1 \otimes \mathrm{id}_A} P_1 \otimes A \xleftarrow{d_2 \otimes \mathrm{id}_A} \cdots$$
$$\downarrow_{\theta_A} \qquad \theta_{P_0 \otimes A} \downarrow \downarrow_{\mathrm{id}_{P_0} \otimes \theta_A} \qquad \theta_{P_1 \otimes A} \downarrow \downarrow_{\mathrm{id}_{P_1} \otimes \theta_A}$$
$$A \xleftarrow{\xi \otimes \mathrm{id}_A} P_0 \otimes A \xleftarrow{d_1 \otimes \mathrm{id}_A} P_1 \otimes A \xleftarrow{d_2 \otimes \mathrm{id}_A} \cdots$$

and have the explicit form

$$\theta_{P_m \otimes A}(p \otimes a) = v_{(1)} \cdot p \otimes v_{(2)}a \quad \text{and} \quad (\mathrm{id}_{P_m} \otimes \theta_A)(p \otimes a) = p \otimes va$$

in terms of the ribbon element v.

Now we can apply the more standard comparison theorem for bimodules, considered as  $A \otimes A^{\text{op}}$ -modules, to see that these two liftings are chain-homotopic via a chain homotopy  $h'_m: P_m \otimes A \to P_{m+1} \otimes A$ . The mappings  $h_m(X)$  are then induced from the mappings  $h'_m \otimes_A \operatorname{id}_X$  via the isomorphism  $(P_m \otimes A) \otimes_A X \cong P_m \otimes X$ .

## 3.2 Derived block spaces

Our goal now is to extend the projective action of the pure mapping class group  $\mathrm{P}\Gamma_{g,n}$ on the block spaces

$$\mathcal{Z}(\Sigma_{g,n}^{X_1,\dots,X_n}) := \operatorname{Hom}_{\mathcal{C}}(X_n \otimes \dots \otimes X_1, L^{\otimes g})$$

to the derived block spaces

$$\mathcal{Z}^m(\Sigma_{g,n}^{X_1,\ldots,X_n}) := \operatorname{Ext}^m(X_n \otimes \cdots \otimes X_1, L^{\otimes g}).$$

To do this, we consider the surface  $\Sigma_{g,n+1}$  with one additional boundary component, and choose a projective resolution

$$\mathbb{1} \xleftarrow{\xi} P_0 \xleftarrow{d_1} P_1 \xleftarrow{d_2} P_2 \xleftarrow{d_3} \cdots$$

of the unit object. As we mentioned in Paragraph 2.1, tensoring is exact and preserves projectives, so that

$$\mathbb{1} \otimes X_n \otimes \cdots \otimes X_1 \longleftarrow P_0 \otimes X_n \otimes \cdots \otimes X_1 \longleftarrow P_1 \otimes X_n \otimes \cdots \otimes X_1 \longleftarrow \cdots$$

is a projective resolution of  $X_n \otimes \cdots \otimes X_1 \cong \mathbb{1} \otimes X_n \otimes \cdots \otimes X_1$ . If we use the abbreviation  $\Gamma_{g,n+1}(n+1)$  for the mapping class group  $\Gamma_{g,n+1}(\operatorname{Im}(\rho_{n+1}))$ , an element  $[\psi]$  of this group fixes the n+1-st boundary component. Then the permutation  $\tau := p([\psi])^{-1} \in S_{n+1}$  introduced in Paragraph 1.3 fixes n+1, and therefore each representative  $\mathcal{Z}(\psi)$  of the associated projective class  $[\mathcal{Z}(\psi)]$  yields by naturality a cochain homomorphism between the cochain complexes

$$\mathcal{Z}(\Sigma_{g,n+1}^{X_1,\dots,X_n,P_m}) = \operatorname{Hom}_{\mathcal{C}}(P_m \otimes X_n \otimes \dots \otimes X_1, L^{\otimes g})$$

and

$$\mathcal{Z}(\Sigma_{g,n+1}^{X_{\tau(1)},\dots,X_{\tau(n)},P_m}) = \operatorname{Hom}_{\mathcal{C}}(P_m \otimes X_{\tau(n)} \otimes \dots \otimes X_{\tau(1)}, L^{\otimes g})$$

and so induces a homomorphism

$$\mathcal{Z}^m(\psi) \colon \operatorname{Ext}^m(X_n \otimes \cdots \otimes X_1, L^{\otimes g}) \to \operatorname{Ext}^m(X_{\tau(n)} \otimes \cdots \otimes X_{\tau(1)}, L^{\otimes g})$$

between the derived block spaces. Choosing a different representative of the projective class clearly rescales  $\mathcal{Z}^m(\psi)$  by a nonzero scalar, so that the projective class  $[\mathcal{Z}^m(\psi)]$  is well-defined. However, we have associated this homomorphism between derived block spaces with a mapping class  $[\psi] \in \Gamma_{g,n+1}(n+1)$ , whereas the homomorphisms between the original block spaces were associated with a mapping class  $[\psi] \in \Gamma_{g,n}$ . As we will show now, we can also associate homomorphisms between the derived block spaces with a mapping class  $[\psi] \in \Gamma_{g,n}$ , namely by choosing a preimage in  $\Gamma_{g,n+1}(n+1)$  under the homomorphism  $D_{n+1}$  defined at the end of Paragraph 1.9. For this, we obviously need to show that the arising homomorphisms are independent of the chosen preimage. We begin with a few auxiliary results:

**Lemma 3.2.** For the Dehn twist  $\mathfrak{d}_{n+1} \in \Gamma_{g,n+1}(n+1)$ , we have  $[\mathcal{Z}^m(\mathfrak{d}_{n+1})] = [\mathrm{id}]$ .

**Proof.** According to Paragraph 2.6, the Dehn twist  $\mathfrak{d}_{n+1}$  acts by precomposition with the morphism  $\theta_{X_{n+1}} \otimes \operatorname{id}_{X_n \otimes \cdots \otimes X_1}$ . By [Ka, Lem. XIV.3.3, p. 350], we have  $\theta_{\mathbb{I}} = \operatorname{id}_{\mathbb{I}}$ . As we have discussed already at the end of Paragraph 3.1, the diagram

$$\mathbb{1} \xleftarrow{\xi} P_0 \xleftarrow{d_1} P_1 \xleftarrow{d_2} \cdots \\
\downarrow_{\theta_1} \qquad \downarrow_{\theta_{P_0}} \qquad \downarrow_{\theta_{P_1}} \\
\mathbb{1} \xleftarrow{\xi} P_0 \xleftarrow{d_1} P_1 \xleftarrow{d_2} \cdots$$

commutes by the naturality of the twist. On the other hand, it is obvious that the diagram

$$\mathbb{1} \xleftarrow{\xi} P_0 \xleftarrow{d_1} P_1 \xleftarrow{d_2} \cdots \\
\downarrow_{\mathrm{id}_1} \qquad \downarrow_{\mathrm{id}_{P_0}} \qquad \downarrow_{\mathrm{id}_{P_1}} \\
\mathbb{1} \xleftarrow{\xi} P_0 \xleftarrow{d_1} P_1 \xleftarrow{d_2} \cdots$$

commutes. Therefore, both the family  $(\theta_{P_m})$  and the family  $(id_{P_m})$  lift the morphism  $\theta_1 = id_1$  to the projective resolution. By the comparison theorem, the two lifts are chain-homotopic. This chain homotopy induces a cochain homotopy on the cochain complex  $\operatorname{Hom}_{\mathcal{C}}(P_m \otimes X_n \otimes \cdots \otimes X_1, L^{\otimes g})$ , which yields that the two maps induce the same map in cohomology, namely the identity.

We will need another lemma of a similar nature:

**Lemma 3.3.** For the two Dehn twists  $\mathfrak{d}_n$  and  $\mathfrak{d}_{n,n+1}$  in  $\Gamma_{g,n+1}(n+1)$ , we have  $[\mathcal{Z}^m(\mathfrak{d}_n)] = [\mathcal{Z}^m(\mathfrak{d}_{n,n+1})].$ 

**Proof.** From Lemma 3.1, we know that the chain maps  $\theta_{P_m \otimes X_n}$  and  $\mathrm{id}_{P_m} \otimes \theta_{X_n}$  are chain-homotopic. This implies that the chain maps

$$\theta_{P_m \otimes X_n} \otimes \operatorname{id}_{X_{n-1} \otimes \cdots \otimes X_1}$$
 and  $\operatorname{id}_{P_m} \otimes \theta_{X_n} \otimes \operatorname{id}_{X_{n-1} \otimes \cdots \otimes X_1}$ 

are chain-homotopic. Because  $\mathfrak{d}_{n,n+1}$  acts by precomposition with the first one and  $\mathfrak{d}_n$  acts by precomposition with the second one, this implies the assertion.

We now consider the curve  $\alpha_i$  introduced in Paragraph 1.1, which begins and ends in the base point x for the surface  $\Gamma_{g,n}$ . If we push it off the base point to the left and to the right as described in Paragraph 1.6, we can cut out a small disk centered at x without intersecting these two curves, which we consider as the n + 1-st boundary component. In this way, we obtain two curves  $\alpha'_i$  and  $\alpha''_i$  on the surface  $\Gamma_{g,n+1}$ :



We now label the new boundary component with the elements of our projective resolution; i.e., for a given m, we set  $X_{n+1} := P_m$ . We want to show that the two Dehn twists  $\mathfrak{t}'_i := \mathfrak{d}_{\alpha'_i}$  and  $\mathfrak{t}''_i := \mathfrak{d}_{\alpha''_i}$  induce the same map in cohomology. As we will see below, it is sufficient to consider the case i = 1:

**Lemma 3.4.** For the two Dehn twists  $\mathfrak{t}'_1 := \mathfrak{d}_{\alpha'_1}$  and  $\mathfrak{t}''_1 := \mathfrak{d}_{\alpha''_1}$  in  $\Gamma_{g,n+1}(n+1)$ , we have  $[\mathcal{Z}^m(\mathfrak{t}'_1)] = [\mathcal{Z}^m(\mathfrak{t}''_1)].$ 

**Proof.** (1) If we cut the surface along the curve that is denoted by  $\sigma$  in the picture above, we obtain a surface that is diffeomorphic to  $\Sigma_{g-1,n+3}$ :



Here, we consider the lower boundary component arising from the cut in the picture above as the n + 2-nd one and the upper boundary component as the n + 3-rd one.

The surface  $\Sigma_{g,n+1}$  can be reconstructed from the surface  $\Sigma_{g-1,n+3}$  by gluing in a handle, as described in Paragraph 2.7. Upon gluing, the Dehn twists along the curves denoted by  $\gamma'$  and  $\gamma''$  in the second picture become the Dehn twists along the curves denoted by  $\alpha'_1$  and  $\alpha''_1$  in the first picture, respectively. As we saw in Paragraph 2.7, the diagram

commutes for our choices of the representatives  $\mathcal{Z}(\mathfrak{d}_{\gamma'})$  and  $\mathcal{Z}(\mathfrak{t}'_1)$  of the projective classes, because upon appropriate labeling  $\mathfrak{d}_{\gamma'}$  is  $\mathfrak{d}_{n+2}$  and  $\mathfrak{t}'_1$  is  $\mathfrak{t}_1$  in the notation used there. From the discussion in Paragraph 2.7, we know that a similar diagram commutes for  $\mathfrak{d}_{\gamma''}$  and  $\mathfrak{t}''_1$ . Our goal is to show that  $\mathcal{Z}(\mathfrak{t}'_1)$  and  $\mathcal{Z}(\mathfrak{t}''_1)$  induce the same map in cohomology.

(2) To see this, we apply Lemma 3.1 to the so-called reversed category, in which tensor products are taken in the opposite order. In this way, we obtain for each

object  $X \in \mathcal{C}$  a chain homotopy  $h_m(X): X \otimes P_m \to X \otimes P_{m+1}$  between the chain maps  $(\theta_{X \otimes P_m})$  and  $(\theta_X \otimes \operatorname{id}_{P_m})$  that is natural in X. For two objects X and Y of  $\mathcal{C}$ , the Dehn twists  $\mathfrak{d}_{\gamma'}$  and  $\mathfrak{d}_{\gamma''}$  act on the space  $\operatorname{Hom}_{\mathcal{C}}(Y^* \otimes X^{**} \otimes P_m \otimes X_n \otimes \cdots \otimes X_1, L^{\otimes (g-1)})$ by precomposition with  $\operatorname{id}_{Y^*} \otimes \theta_{X^{**}} \otimes \operatorname{id}_{P_m \otimes Z}$  and  $\operatorname{id}_{Y^*} \otimes \theta_{X^{**} \otimes P_m} \otimes \operatorname{id}_Z$ , respectively, where, for brevity, we have used the notation  $Z := X_n \otimes \cdots \otimes X_1$ . By naturality, the adjunction isomorphism

$$\operatorname{Hom}_{\mathcal{C}}(Y^* \otimes X^{**} \otimes P_m \otimes Z, L^{\otimes (g-1)}) \to \operatorname{Hom}_{\mathcal{C}}(P_m \otimes Z, X^* \otimes Y \otimes L^{\otimes (g-1)})$$

is an isomorphism of cochain complexes, so that we obtain a cochain map  $(f_{X,Y}'^m)$  of the cochain complex on the right-hand side whose defining property is that the diagram

commutes. For  $\gamma''$ , there is a second cochain map  $(f_{X,Y}'')$  that makes a very similar diagram commutative. For  $\gamma'$ , the naturality of the adjunction isomorphism implies that  $f_{X,Y}''$  is given by postcomposition with  $\theta_{X^*} \otimes \operatorname{id}_Y \otimes \operatorname{id}_{L^{\otimes (g-1)}}$ . The chain homotopy  $(\operatorname{id}_{Y^*} \otimes h_m(X^{**}) \otimes \operatorname{id}_Z)$  induces a cochain homotopy

$$h'_m(X,Y)$$
: Hom <sub>$\mathcal{C}$</sub>  $(P_m \otimes Z, X^* \otimes Y \otimes L^{\otimes (g-1)}) \to \text{Hom}_{\mathcal{C}}(P_{m-1} \otimes Z, X^* \otimes Y \otimes L^{\otimes (g-1)})$ 

that is natural in X and Y and makes the diagram

commutative.

(3) Now the functor  $\operatorname{Hom}_{\mathcal{C}}(P_m \otimes Z, - \otimes L^{\otimes (g-1)})$  is an exact functor from the category  $\mathcal{C}$  to the category  $\mathcal{V}$  of finite-dimensional vector spaces. By Proposition 2.1, it has a right adjoint and therefore preserves coends, as already mentioned in Paragraph 2.3. Therefore, the family of morphisms

$$(\iota_X \otimes \mathrm{id}_{L^{\otimes (g-1)}}) \circ \colon \mathrm{Hom}_{\mathcal{C}}(P_m \otimes Z, X^* \otimes X \otimes L^{\otimes (g-1)}) \to \mathrm{Hom}_{\mathcal{C}}(P_m \otimes Z, L^{\otimes g})$$

is a coend for the bifunctor

$$\mathcal{C}^{\mathrm{op}} \times \mathcal{C} \to \mathcal{V}, \ (X, Y) \mapsto \mathrm{Hom}_{\mathcal{C}}(P_m \otimes Z, X^* \otimes Y \otimes L^{\otimes (g-1)}).$$

So we can apply Lemma 2.2 to this bifunctor and the corresponding bifunctor with m-1 instead of m to obtain a K-linear map

$$h'_m$$
: Hom <sub>$\mathcal{C}$</sub>  $(P_m \otimes Z, L^{\otimes g}) \to$  Hom <sub>$\mathcal{C}$</sub>  $(P_{m-1} \otimes Z, L^{\otimes g})$ 

that makes the diagram

$$\operatorname{Hom}_{\mathcal{C}}(P_m \otimes Z, X^* \otimes X \otimes L^{\otimes (g-1)}) \longrightarrow \operatorname{Hom}_{\mathcal{C}}(P_m \otimes Z, L^{\otimes g})$$
$$\downarrow^{h'_m}$$
$$\operatorname{Hom}_{\mathcal{C}}(P_{m-1} \otimes Z, X^* \otimes X \otimes L^{\otimes (g-1)}) \longrightarrow \operatorname{Hom}_{\mathcal{C}}(P_{m-1} \otimes Z, L^{\otimes g})$$

commutative, i.e., satisfies

$$h'_m((\iota_X \otimes \mathrm{id}_{L^{\otimes (g-1)}}) \circ k) = (\iota_X \otimes \mathrm{id}_{L^{\otimes (g-1)}}) \circ h'_m(X, X)(k)$$

for all morphisms  $k \colon P_m \otimes Z \to X^* \otimes X \otimes L^{\otimes (g-1)}$ .

(4) By the definition of  $h_m(X^{**})$ , we have

$$\theta_{X^{**}\otimes P_m} - \theta_{X^{**}} \otimes \operatorname{id}_{P_m} = (\operatorname{id}_{X^{**}} \otimes d_{m+1}) \circ h_m(X^{**}) + h_{m-1}(X^{**}) \circ (\operatorname{id}_{X^{**}} \otimes d_m)$$

For a morphism  $k' \colon Y^* \otimes X^{**} \otimes P_m \otimes Z \to L^{\otimes (g-1)}$ , this implies that

$$k' \circ [(\mathrm{id}_{Y^*} \otimes \theta_{X^{**} \otimes P_m} \otimes \mathrm{id}_Z) - (\mathrm{id}_{Y^*} \otimes \theta_{X^{**}} \otimes \mathrm{id}_{P_m \otimes Z})]$$
  
=  $k' \circ [(\mathrm{id}_{Y^* \otimes X^{**}} \otimes d_{m+1} \otimes \mathrm{id}_Z) \circ (\mathrm{id}_{Y^*} \otimes h_m(X^{**}) \otimes \mathrm{id}_Z)$   
+  $(\mathrm{id}_{Y^*} \otimes h_{m-1}(X^{**}) \otimes \mathrm{id}_Z) \circ (\mathrm{id}_{Y^* \otimes X^{**}} \otimes d_m \otimes \mathrm{id}_Z)]$ 

In view of the definition of  $h'_m(X, Y)$  and the naturality of the adjunction isomorphism, this yields

$$f_{X,Y}''^{m}(k) - f_{X,Y}'^{m}(k) = h_{m+1}'(X,Y)(k \circ (d_{m+1} \otimes \mathrm{id}_Z)) + h_m'(X,Y)(k) \circ (d_m \otimes \mathrm{id}_Z)$$

for all morphisms  $k: P_m \otimes Z \to X^* \otimes Y \otimes L^{\otimes (g-1)}$ . If we set X = Y and compose with  $\iota_X \otimes \operatorname{id}_{L^{\otimes (g-1)}}$ , this equation becomes

$$\begin{aligned} \mathcal{Z}(\mathfrak{t}_1'') \circ (\iota_X \otimes \mathrm{id}_{L^{\otimes (g-1)}}) \circ k &- \mathcal{Z}(\mathfrak{t}_1') \circ (\iota_X \otimes \mathrm{id}_{L^{\otimes (g-1)}}) \circ k \\ &= h_{m+1}'((\iota_X \otimes \mathrm{id}_{L^{\otimes (g-1)}}) \circ k \circ (d_{m+1} \otimes \mathrm{id}_Z)) + h_m'((\iota_X \otimes \mathrm{id}_{L^{\otimes (g-1)}}) \circ k) \circ (d_m \otimes \mathrm{id}_Z) \end{aligned}$$

Both sides of this equation define a dinatural transformation from the bifunctor

$$\mathcal{C}^{\mathrm{op}} \times \mathcal{C} \to \mathcal{V}, \ (X, Y) \mapsto \operatorname{Hom}_{\mathcal{C}}(P_m \otimes Z, X^* \otimes Y \otimes L^{\otimes (g-1)})$$

already considered above to its coend  $\operatorname{Hom}_{\mathcal{C}}(P_m \otimes Z, L^{\otimes g})$ . Our dinatural transformation factors over this coend, and the corresponding homomorphism, which is unique, can be read off directly from both the left and the right-hand side of the equation above. We get

$$\mathcal{Z}(\mathfrak{t}_1'') \circ k'' - \mathcal{Z}(\mathfrak{t}_1') \circ k'' = h'_{m+1}(k'' \circ (d_{m+1} \otimes \mathrm{id}_Z)) + h'_m(k'') \circ (d_m \otimes \mathrm{id}_Z)$$

for all  $k'' \in \operatorname{Hom}_{\mathcal{C}}(P_m \otimes Z, L^{\otimes g})$ . Therefore, the family  $(h'_m)$  constitutes a cochain homotopy between the cochain maps induced by  $\mathcal{Z}(\mathfrak{t}'_1)$  and  $\mathcal{Z}(\mathfrak{t}''_1)$ . When saying that, it should be noted that these maps are only determined up to a scalar; we have taken here for  $\mathcal{Z}(\mathfrak{t}'_1)$  the representative that corresponds to the dinatural transformation  $k \mapsto (\iota_X \otimes \operatorname{id}_{L^{\otimes (g-1)}}) \circ f'^m_{X,X}(k)$  via the universal property of the coend of our bifunctor, and have made a similar choice for the representative of  $\mathcal{Z}(\mathfrak{t}''_1)$  using  $f''^m_{X,X}$ .

By putting these auxiliary results together, we can now associate with a mapping class in  $\Gamma_{g,n}$  a projective class of morphisms not only between the original block spaces, but rather between the derived block spaces. As already stated above, we do this by choosing a preimage under the epimorphism  $D_{n+1}$  defined at the end of Paragraph 1.9. The key fact that we need to prove is therefore the following:

**Theorem 3.5.** Suppose that  $[\psi], [\psi'] \in \Gamma_{g,n+1}(n+1)$  satisfy  $D_{n+1}([\psi]) = D_{n+1}([\psi'])$ . Then we have  $[\mathcal{Z}^m(\psi)] = [\mathcal{Z}^m(\psi')]$ .

**Proof.** (1) Clearly,  $[\psi]$  and  $[\psi']$  differ by an element in the kernel of  $D_{n+1}$ , which is in particular an element in the pure mapping class group  $\Pr_{g,n+1}$ . It therefore suffices to show that  $[\mathcal{Z}^m(\psi)] = [\text{id}]$  for each mapping class  $[\psi]$  in the kernel of  $D_{n+1}$ . Recall that  $D_{n+1}$  is the composition

$$\Gamma_{g,n+1}(n+1) \xrightarrow{C_{n+1}} \Gamma_{g,n}(y) \xrightarrow{F_y} \Gamma_{g,n}(y)$$

of the capping homomorphism, which arises from gluing a punctured disk in place of the missing disk, and the forgetful map  $F_y$  that, depending on the perspective, either fills this puncture or 'forgets' that the point was marked. According to Proposition 1.8, the kernel of  $C_{n+1}$  is generated by the Dehn twist  $\mathfrak{d}_{n+1}$ . But by Lemma 3.2 above, we have  $[\mathcal{Z}^m(\mathfrak{d}_{n+1})] = [\text{id}]$ . This implies that the assignment  $[\psi] \mapsto [\mathcal{Z}^m(\psi)]$ is well-defined for  $[\psi] \in \Gamma_{g,n}(y)$ , not only for  $[\psi] \in \Gamma_{g,n+1}(n+1)$ . (2) To treat the second morphism  $F_y$  in the above composition, we use the Birman sequence from Paragraph 1.10. In order to do this, we proceed as in Paragraph 1.12 and first replace the additional puncture y by the base point x of the fundamental group that comes from the polygon model of the surface as the identification of all the vertices of the polygon, except for those that correspond to the marked points on the boundary components. The map that forgets the base point x instead of the puncture will be denoted by  $F_x$  instead of  $F_y$ . The Birman sequence then takes the form

$$\pi_1(\Sigma_{g,n}, x) \xrightarrow{P_x} \Gamma_{g,n}(x) \xrightarrow{F_x} \Gamma_{g,n} \longrightarrow 1,$$

where, as before,  $P_x$  denotes the pushing map. To complete the proof, we therefore have to show that  $[\mathcal{Z}^m(P_x([\gamma]))] = [\text{id}]$  for all homotopy classes  $[\gamma] \in \pi_1(\Sigma_{g,n}, x)$ . Because  $P_x$  is a group antihomomorphism, it suffices to prove this for all  $[\gamma]$  in a generating set of the fundamental group.

(3) As discussed in Paragraph 1.2, the fundamental group  $\pi_1(\Sigma_{g,n}, x)$  is generated by the homotopy classes of the simple closed curves  $\alpha_1, \beta_1, \ldots, \alpha_g, \beta_g$ , all of which are nonseparating, together with the curves  $\delta_1, \ldots, \delta_n$ , all of which are separating. From a formula stated at the end of Paragraph 1.10, we know that  $P_x([\alpha_1]) = [\mathfrak{d}_{\alpha'_1} \mathfrak{d}_{\alpha''_1}^{-1}]$ . Therefore Lemma 3.4 yields that  $[\mathcal{Z}^m(P_x([\alpha_1]))] = [\mathrm{id}]$ .

(4) For any of the other generators that correspond to nonseparating curves, say  $\beta_i$ , we use the change of coordinates principle (cf. [FM, Par. 1.3.1, p. 37]) to obtain a diffeomorphism  $\varphi \colon \Sigma_{g,n} \to \Sigma_{g,n}$  satisfying  $\varphi(\alpha_1) = \beta_i$ . The discussion in [FM, loc. cit.] shows that we can assume that  $\varphi$  is orientation-preserving and satisfies  $\varphi(x) = x$ . From Paragraph 1.10, we know that  $[P_x([\beta_i])] = [\varphi][P_x([\alpha_1])][\varphi^{-1}]$ . Now the compatibility with composition described in Paragraph 2.6 implies that

$$[\mathcal{Z}^m(P_x([\beta_i]))] = [\mathcal{Z}^m(\varphi)][\mathcal{Z}^m(P_x([\alpha_1]))][\mathcal{Z}^m(\varphi^{-1})] = [\mathcal{Z}^m(\varphi)][\mathcal{Z}^m(\varphi^{-1})] = [\mathrm{id}]$$

(5) For the separating curves  $\delta_1, \ldots, \delta_n$ , the formula already mentioned above yields that  $P_x([\delta_j]) = [\mathfrak{d}_{\delta'_j}\mathfrak{d}_{\delta''_j}^{-1}]$ . We have  $C_{n+1}([\mathfrak{d}_n]) = [\mathfrak{d}_{\delta'_n}]$  and  $C_{n+1}([\mathfrak{d}_{n,n+1}]) = [\mathfrak{d}_{\delta''_n}]$ . Therefore, Lemma 3.3 shows that  $[\mathcal{Z}^m(P_x([\delta_n]))] = [\mathrm{id}]$ . If  $j \neq n$ , we proceed as above and use the general change of coordinate principle (cf. [FM, loc. cit.]) to obtain a diffeomorphism  $\varphi \colon \Sigma_{g,n} \to \Sigma_{g,n}$  with  $\varphi(\delta_n) = \delta_j$  that is orientation-preserving and satisfies  $\varphi(x) = x$ . As above, we get  $[P_x([\delta_j])] = [\varphi][P_x([\delta_n])][\varphi^{-1}]$ , and a very similar computation as there then yields  $[\mathcal{Z}^m(P_x([\delta_j]))] = [\mathrm{id}]$ . By construction, a mapping class that permutes the boundary components does in general not carry a derived block space into itself. However, the pure mapping class group preserves these spaces:

**Corollary 3.6.** There is a projective action of  $P\Gamma_{g,n}$  on

$$\mathcal{Z}^m(\Sigma_{g,n}^{X_1,\dots,X_n}) = \operatorname{Ext}^m(X_n \otimes \dots \otimes X_1, L^{\otimes g})$$

that agrees with the original action on  $\mathcal{Z}(\Sigma_{g,n}^{X_1,\ldots,X_n}) = \operatorname{Hom}_{\mathcal{C}}(X_n \otimes \cdots \otimes X_1, L^{\otimes g})$ if m = 0.

As in Paragraph 2.6, the entire mapping class acts projectively on the direct sum

$$\bigoplus_{\tau \in S_n} \mathcal{Z}^m(\Sigma_{g,n}^{X_{\tau(1)},\dots,X_{\tau(n)}})$$

of derived block spaces. If all boundary components have the same label X, the mapping class group indeed acts projectively on a single derived block space:

**Corollary 3.7.** For  $X \in C$ , there is a projective action of  $\Gamma_{g,n}$  on  $\operatorname{Ext}^m(X^{\otimes n}, L^{\otimes g})$  that agrees with the original action on  $\operatorname{Hom}_{\mathcal{C}}(X^{\otimes n}, L^{\otimes g})$  if m = 0.

#### **3.3** The case of the sphere

The construction of the mapping class group representations in the preceding paragraph was not based on generators and relations for the mapping class group, as such a presentation of the mapping class group is notoriously difficult. In the case g = 0, however, we gave such a presentation in Paragraph 1.13. It is instructive to see why the defining relations are satisfied in this case. In order to verify these relations, we will need a couple of lemmas. As stated in Paragraph 2.1, we are assuming that our category is strict.

**Lemma 3.8.** Suppose that  $X_1, \ldots, X_n$  are objects of  $\mathcal{C}$ . Then we have

1. 
$$c_{X_1 \otimes \cdots \otimes X_{n-1}, X_n} = (c_{X_1, X_n} \otimes \operatorname{id}_{X_2 \otimes \cdots \otimes X_{n-1}}) \circ (\operatorname{id}_{X_1} \otimes c_{X_2, X_n} \otimes \operatorname{id}_{X_3 \otimes \cdots \otimes X_{n-1}})$$
  
 $\circ \cdots \circ (\operatorname{id}_{X_1 \otimes \cdots \otimes X_{n-2}} \otimes c_{X_{n-1}, X_n})$ 

2. 
$$c_{X_1,X_2\otimes\cdots\otimes X_n} = (\operatorname{id}_{X_2\otimes\cdots\otimes X_{n-1}}\otimes c_{X_1,X_n})\circ\cdots\circ(\operatorname{id}_{X_2}\otimes c_{X_1,X_3}\otimes\operatorname{id}_{X_4\otimes\cdots\otimes X_n})$$
  
 $\circ(c_{X_1,X_2}\otimes\operatorname{id}_{X_3\otimes\cdots\otimes X_n})$ 

3.  $c_{X_n,X_1\otimes\cdots\otimes X_{n-1}}\circ\cdots\circ c_{X_2,X_3\otimes\cdots\otimes X_n\otimes X_1}\circ c_{X_1,X_2\otimes\cdots\otimes X_n} = \theta_{X_1\otimes\cdots\otimes X_n}\circ(\theta_{X_1}^{-1}\otimes\cdots\otimes \theta_{X_n}^{-1})$ 4.  $c_{X_2\otimes\cdots\otimes X_n,X_1}\circ c_{X_3\otimes\cdots\otimes X_n\otimes X_1,X_2}\circ\cdots\circ c_{X_1\otimes\cdots\otimes X_{n-1},X_n} = \theta_{X_1\otimes\cdots\otimes X_n}\circ(\theta_{X_1}^{-1}\otimes\cdots\otimes \theta_{X_n}^{-1})$ 

**Proof.** (1) The first assertion is clearly correct for n = 2 and then follows inductively from the equation

$$c_{X_1 \otimes \cdots \otimes X_{n-1}, X_n} = (c_{X_1 \otimes \cdots \otimes X_{n-2}, X_n} \otimes \operatorname{id}_{X_{n-1}}) \circ (\operatorname{id}_{X_1 \otimes \cdots \otimes X_{n-2}} \otimes c_{X_{n-1}, X_n}).$$

(2) Associated with every braiding is another braiding in which  $c_{X,Y}$  is replaced by  $c_{Y,X}^{-1}$ . If we apply the first assertion to this braiding instead, take inverses, and permute  $X_1, \ldots, X_n$  cyclically, we obtain the second assertion.

(3) To prove the third assertion, we need the auxiliary statement that

 $c_{X_k,X_{k+1}\otimes\cdots\otimes X_n\otimes X_1\otimes\cdots\otimes X_{k-1}}\circ\cdots\circ c_{X_2,X_3\otimes\cdots\otimes X_n\otimes X_1}\circ c_{X_1,X_2\otimes\cdots\otimes X_n} = c_{X_1\otimes\cdots\otimes X_k}\circ (\theta_{X_1\otimes\cdots\otimes X_k}\otimes \operatorname{id}_{X_{k+1}\otimes\cdots\otimes X_n})\circ (\theta_{X_1}^{-1}\otimes\cdots\otimes \theta_{X_k}^{-1}\otimes \operatorname{id}_{X_{k+1}\otimes\cdots\otimes X_n})$ for  $k=1,\ldots,n-1$ . For k=1, this is obvious. For  $k\leq n-2$ , we have inductively

$$\begin{aligned} c_{X_{k+1},X_{k+2}\otimes\cdots\otimes X_n\otimes X_1\otimes\cdots\otimes X_k} &\circ c_{X_1\otimes\cdots\otimes X_k,X_{k+1}\otimes\cdots\otimes X_n} \\ &\circ \left(\theta_{X_1\otimes\cdots\otimes X_k}\otimes \operatorname{id}_{X_{k+1}\otimes\cdots\otimes X_n}\right) \circ \left(\theta_{X_1}^{-1}\otimes\cdots\otimes \theta_{X_k}^{-1}\otimes \operatorname{id}_{X_{k+1}\otimes\cdots\otimes X_n}\right) \\ &= \left(\operatorname{id}_{X_{k+2}\otimes\cdots\otimes X_n}\otimes c_{X_{k+1},X_1\otimes\cdots\otimes X_k}\right) \circ \left(c_{X_{k+1},X_{k+2}\otimes\cdots\otimes X_n}\otimes\operatorname{id}_{X_1\otimes\cdots\otimes X_k}\right) \\ &\circ \left(\operatorname{id}_{X_{k+1}}\otimes c_{X_1\otimes\cdots\otimes X_k,X_{k+2}\otimes\cdots\otimes X_n}\right) \circ \left(c_{X_1\otimes\cdots\otimes X_k,X_{k+1}}\otimes\operatorname{id}_{X_{k+2}\otimes\cdots\otimes X_n}\right) \\ &\circ \left(\theta_{X_1\otimes\cdots\otimes X_k}\otimes \theta_{X_{k+1}}\otimes\operatorname{id}_{X_{k+2}\otimes\cdots\otimes X_n}\right) \circ \left(\theta_{X_1}^{-1}\otimes\cdots\otimes \theta_{X_{k+1}}^{-1}\otimes\operatorname{id}_{X_{k+2}\otimes\cdots\otimes X_n}\right). \end{aligned}$$

By using the Yang-Baxter equation (cf. [Ka, Thm. XIII.1.3, p. 317]) on the first three terms, this expression becomes

$$(c_{X_1 \otimes \dots \otimes X_k, X_{k+2} \otimes \dots \otimes X_n} \otimes \operatorname{id}_{X_{k+1}}) \circ (\operatorname{id}_{X_1 \otimes \dots \otimes X_k} \otimes c_{X_{k+1}, X_{k+2} \otimes \dots \otimes X_n}) \\ \circ (c_{X_{k+1}, X_1 \otimes \dots \otimes X_k} \otimes \operatorname{id}_{X_{k+2} \otimes \dots \otimes X_n}) \circ (c_{X_1 \otimes \dots \otimes X_k, X_{k+1}} \otimes \operatorname{id}_{X_{k+2} \otimes \dots \otimes X_n}) \\ \circ (\theta_{X_1 \otimes \dots \otimes X_k} \otimes \theta_{X_{k+1}} \otimes \operatorname{id}_{X_{k+2} \otimes \dots \otimes X_n}) \circ (\theta_{X_1}^{-1} \otimes \dots \otimes \theta_{X_{k+1}}^{-1} \otimes \operatorname{id}_{X_{k+2} \otimes \dots \otimes X_n}),$$

which by the defining property of a ribbon structure is equal to

$$(c_{X_{1}\otimes\cdots\otimes X_{k},X_{k+2}\otimes\cdots\otimes X_{n}}\otimes \operatorname{id}_{X_{k+1}})\circ(\operatorname{id}_{X_{1}\otimes\cdots\otimes X_{k}}\otimes c_{X_{k+1},X_{k+2}\otimes\cdots\otimes X_{n}})$$
  

$$\circ(\theta_{X_{1}\otimes\cdots\otimes X_{k}\otimes X_{k+1}}\otimes \operatorname{id}_{X_{k+2}\otimes\cdots\otimes X_{n}})\circ(\theta_{X_{1}}^{-1}\otimes\cdots\otimes\theta_{X_{k+1}}^{-1}\otimes \operatorname{id}_{X_{k+2}\otimes\cdots\otimes X_{n}})$$
  

$$=c_{X_{1}\otimes\cdots\otimes X_{k+1},X_{k+2}\otimes\cdots\otimes X_{n}}\circ(\theta_{X_{1}\otimes\cdots\otimes X_{k+1}}\otimes \operatorname{id}_{X_{k+2}\otimes\cdots\otimes X_{n}})$$
  

$$\circ(\theta_{X_{1}}^{-1}\otimes\cdots\otimes\theta_{X_{k+1}}^{-1}\otimes \operatorname{id}_{X_{k+2}\otimes\cdots\otimes X_{n}}),$$

completing our inductive step.

(4) To derive the third assertion, we now use the case k = n - 1 of the auxiliary statement above and apply  $c_{X_n, X_1 \otimes \cdots \otimes X_{n-1}}$  to obtain

$$c_{X_n,X_1\otimes\cdots\otimes X_{n-1}}\circ c_{X_{n-1},X_n\otimes X_1\otimes\cdots\otimes X_{n-2}}\circ\cdots\circ c_{X_2,X_3\otimes\cdots\otimes X_n\otimes X_1}\circ c_{X_1,X_2\otimes\cdots\otimes X_n}$$

$$=c_{X_n,X_1\otimes\cdots\otimes X_{n-1}}\circ c_{X_1\otimes\cdots\otimes X_{n-1},X_n}\circ (\theta_{X_1\otimes\cdots\otimes X_{n-1}}\otimes \operatorname{id}_{X_n})$$

$$\circ (\theta_{X_1}^{-1}\otimes\cdots\otimes \theta_{X_{n-1}}^{-1}\otimes \operatorname{id}_{X_n})$$

$$=c_{X_n,X_1\otimes\cdots\otimes X_{n-1}}\circ c_{X_1\otimes\cdots\otimes X_{n-1},X_n}\circ (\theta_{X_1\otimes\cdots\otimes X_{n-1}}\otimes \theta_{X_n})\circ (\theta_{X_1}^{-1}\otimes\cdots\otimes \theta_{X_n}^{-1})$$

$$=\theta_{X_1\otimes\cdots\otimes X_n}\circ (\theta_{X_1}^{-1}\otimes\cdots\otimes \theta_{X_n}^{-1})$$

again by the defining property of a ribbon structure.

(5) A ribbon structure for the braiding  $c_{Y,X}^{-1}$  considered in the proof of the second assertion is given by the morphisms  $\theta_X^{-1} \colon X \to X$ . If we apply the third assertion to this braiding and this ribbon structure instead and then take inverses, we obtain the fourth assertion.

Another lemma that we will need concerns precomposition with the twist:

**Lemma 3.9.** For all  $f \in \text{Hom}_{\mathcal{C}}(X \otimes Y, \mathbb{1})$ , we have  $f \circ (\theta_X \otimes \text{id}_Y) = f \circ (\text{id}_X \otimes \theta_Y)$ . More generally, precomposition with  $\theta_X \otimes \text{id}_Y$  and precomposition with  $\text{id}_X \otimes \theta_Y$  induce the same homomorphism on  $\text{Ext}^m(X \otimes Y, \mathbb{1})$ .

**Proof.** (1) As we recorded in Paragraph 2.1, the functor  $-\otimes Y^*$  is the right adjoint of the functor  $-\otimes Y$ . The adjunction is given by

$$\eta \colon \operatorname{Hom}_{\mathcal{C}}(X \otimes Y, \mathbb{1}) \to \operatorname{Hom}_{\mathcal{C}}(X, Y^*), \ f \mapsto (f \otimes \operatorname{id}_{Y^*}) \circ (\operatorname{id}_X \otimes \operatorname{coev}_Y).$$

Therefore, the diagram

commutes. On the other hand, also the diagram

commutes, because

$$\eta(f \circ (\mathrm{id}_X \otimes \theta_Y)) = (f \otimes \mathrm{id}_{Y^*}) \circ (\mathrm{id}_X \otimes \theta_Y \otimes \mathrm{id}_{Y^*}) \circ (\mathrm{id}_X \otimes \mathrm{coev}_Y) = (f \otimes \mathrm{id}_{Y^*}) \circ (\mathrm{id}_X \otimes \mathrm{id}_Y \otimes \theta_Y^*) \circ (\mathrm{id}_X \otimes \mathrm{coev}_Y) = (f \otimes \theta_Y^*) \circ (\mathrm{id}_X \otimes \mathrm{coev}_Y) = \theta_Y^* \circ (f \otimes \mathrm{id}_{Y^*}) \circ (\mathrm{id}_X \otimes \mathrm{coev}_Y) = \theta_Y^* \circ \eta(f) = \theta_{Y^*} \circ \eta(f).$$

But by the naturality of the twist, we have  $\theta_{Y^*} \circ g = g \circ \theta_X$  for  $g \in \text{Hom}_{\mathcal{C}}(X, Y^*)$ , so that the bottom rows in the two diagrams are equal. By comparing the two diagrams, we obtain the first assertion.

(2) From our projective resolution

 $\mathbb{1} \longleftarrow P_0 \longleftarrow P_1 \longleftarrow P_2 \longleftarrow \cdots$ 

of the unit object, we get a projective resolution of X by tensoring with X, i.e., by defining  $Q_m := P_m \otimes X$ , and even a projective resolution of  $X \otimes Y$  by tensoring again with Y. Then

is a lift of  $id_X \otimes \theta_Y$  to this projective resolution. On the other hand, a lift of  $\theta_X \otimes id_Y$  is given by

So the action of  $\operatorname{id}_X \otimes \theta_Y$  on  $\operatorname{Ext}^m(X \otimes Y, \mathbb{1})$  is induced by precomposition with  $\operatorname{id}_{Q_m} \otimes \theta_Y$  on  $\operatorname{Hom}_{\mathcal{C}}(Q_m \otimes Y, \mathbb{1})$ , while the action of  $\theta_X \otimes \operatorname{id}_Y$  is induced by precomposition with  $\theta_{Q_m} \otimes \operatorname{id}_Y$  on this space. But by the preceding step, these two precomposition maps are equal.

The results of Paragraph 3.2 yield in the case g = 0 that the projective action of  $\Gamma_{0,n+1}(n+1)$  on the direct sum

$$\bigoplus_{\tau \in S_n} \operatorname{Ext}^m(X_{\tau(n)} \otimes \cdots \otimes X_{\tau(1)}, \mathbb{1})$$

descends to a projective action of  $\Gamma_{0,n}$ . In fact, the action is in this case not only projective, but rather an ordinary linear action. As we will explain now, we can see this explicitly from the presentations of these groups that we gave in Paragraph 1.13. According to Proposition 1.9, we have

$$\Gamma_{0,n+1}(n+1) \cong \mathbb{Z}^n \rtimes B_n$$

It is not difficult to check that the defining relations of  $\mathbb{Z}^n \rtimes B_n$  discussed there are strictly, not only projectively, satisfied on the direct sum

$$\bigoplus_{\tau \in S_n} \operatorname{Hom}_{\mathcal{C}}(P_m \otimes X_{\tau(n)} \otimes \cdots \otimes X_{\tau(1)}, \mathbb{1}).$$

To show that the action descends to an action of  $\Gamma_{0,n}$ , we have to show that the additional relations

$$\mathfrak{b}_{1,2}\mathfrak{b}_{2,3}\cdots\mathfrak{b}_{n-1,n}^2\cdots\mathfrak{b}_{2,3}\mathfrak{b}_{1,2}=\mathfrak{d}_1^{-2}$$

and  $(\mathfrak{b}_{1,2}\mathfrak{b}_{2,3}\cdots\mathfrak{b}_{n-1,n})^n = \mathfrak{d}_1^{-1}\mathfrak{d}_2^{-1}\cdots\mathfrak{d}_n^{-1}$  hold on this space. It is sufficient to verify the relations on a single summand of the direct sum:

**Proposition 3.10.** For  $f \in \text{Hom}_{\mathcal{C}}(P_m \otimes X_n \otimes \cdots \otimes X_1, \mathbb{1})$ , we have

$$\mathcal{Z}^{m}(\mathfrak{b}_{1,2}\mathfrak{b}_{2,3}\cdots\mathfrak{b}_{n-1,n}^{2}\cdots\mathfrak{b}_{2,3}\mathfrak{b}_{1,2})(\bar{f})=\mathcal{Z}^{m}(\mathfrak{d}_{1}^{-2})(\bar{f})$$

as well as

$$\mathcal{Z}^m((\mathfrak{b}_{1,2}\mathfrak{b}_{2,3}\cdots\mathfrak{b}_{n-1,n})^n)(\bar{f})=\mathcal{Z}^m(\mathfrak{d}_1^{-1}\mathfrak{d}_2^{-1}\cdots\mathfrak{d}_n^{-1})(\bar{f}).$$

**Proof.** (1) It follows from the second assertion in Lemma 3.8 that

$$\begin{aligned} \mathcal{Z}(\mathfrak{b}_{n-1,n}\cdots\mathfrak{b}_{2,3}\mathfrak{b}_{1,2})(f) \\ &= f \circ (\mathrm{id}_{P_m \otimes X_n \otimes \cdots \otimes X_3} \otimes c_{X_1,X_2}) \circ (\mathrm{id}_{P_m \otimes X_n \otimes \cdots \otimes X_4} \otimes c_{X_1,X_3} \otimes \mathrm{id}_{X_2}) \\ &\circ \cdots \circ (\mathrm{id}_{P_m} \otimes c_{X_1,X_n} \otimes \mathrm{id}_{X_{n-1} \otimes \cdots \otimes X_2}) \\ &= f \circ (\mathrm{id}_{P_m} \otimes c_{X_1,X_n \otimes \cdots \otimes X_2}) \end{aligned}$$

and then from the first assertion there that

$$\begin{aligned} \mathcal{Z}(\mathfrak{b}_{1,2}\mathfrak{b}_{2,3}\cdots\mathfrak{b}_{n-1,n}^{2}\cdots\mathfrak{b}_{2,3}\mathfrak{b}_{1,2})(f) \\ &= f \circ (\mathrm{id}_{P_{m}}\otimes c_{X_{1},X_{n}\otimes\cdots\otimes X_{2}}) \circ (\mathrm{id}_{P_{m}}\otimes c_{X_{n},X_{1}}\otimes\mathrm{id}_{X_{n-1}\otimes\cdots\otimes X_{2}}) \\ &\circ \cdots \circ (\mathrm{id}_{P_{m}\otimes X_{n}\otimes\cdots\otimes X_{4}}\otimes c_{X_{3},X_{1}}\otimes\mathrm{id}_{X_{2}}) \circ (\mathrm{id}_{P_{m}\otimes X_{n}\otimes\cdots\otimes X_{3}}\otimes c_{X_{2},X_{1}}) \\ &= f \circ (\mathrm{id}_{P_{m}}\otimes c_{X_{1},X_{n}\otimes\cdots\otimes X_{2}}) \circ (\mathrm{id}_{P_{m}}\otimes c_{X_{n}\otimes\cdots\otimes X_{2},X_{1}}) \\ &= f \circ (\mathrm{id}_{P_{m}}\otimes \theta_{X_{n}\otimes\cdots\otimes X_{1}}) \circ (\mathrm{id}_{P_{m}}\otimes \theta_{X_{n}\otimes\cdots\otimes X_{2}}^{-1}\otimes \theta_{X_{1}}^{-1}), \end{aligned}$$

while  $\mathcal{Z}(\mathfrak{d}_1^{-2})(f) = f \circ (\mathrm{id}_{P_m} \otimes \mathrm{id}_{X_n \otimes \cdots \otimes X_2} \otimes \theta_{X_1}^{-2})$ . So, if we introduce the abbreviations  $X := X_n \otimes \cdots \otimes X_2$  and  $Y := X_1$ , we have to show that the two cochain maps

$$f \mapsto f \circ (\mathrm{id}_{P_m} \otimes \theta_{X \otimes Y}) \circ (\mathrm{id}_{P_m} \otimes \theta_X^{-1} \otimes \theta_Y^{-1})$$

and  $f \mapsto f \circ (\mathrm{id}_{P_m} \otimes \mathrm{id}_X \otimes \theta_Y^{-2})$  of the cochain complex  $(\mathrm{Hom}_{\mathcal{C}}(P_m \otimes X \otimes Y, \mathbb{1}))$  induce the same homomorphism in cohomology.

By the first assertion in Lemma 3.9, the cochain map  $f \mapsto f \circ (\mathrm{id}_{P_m} \otimes \theta_{X \otimes Y})$  is equal to the cochain map  $f \mapsto f \circ (\theta_{P_m} \otimes \mathrm{id}_{X \otimes Y})$ , and we saw in Lemma 3.2 that this cochain map is homotopic to the identity. This shows that our assertion will follow if we can show that the two cochain maps  $f \mapsto f \circ (\mathrm{id}_{P_m} \otimes \theta_X^{-1} \otimes \mathrm{id}_Y)$  and  $f \mapsto f \circ (\mathrm{id}_{P_m} \otimes \mathrm{id}_X \otimes \theta_Y^{-1})$  induce the same map in cohomology. But this follows directly from the second assertion in Lemma 3.9 by taking inverses.

(2) To prove the second relation, we note that the first assertion in Lemma 3.8 gives

$$\begin{aligned} \mathcal{Z}(\mathfrak{b}_{1,2}\mathfrak{b}_{2,3}\cdots\mathfrak{b}_{n-1,n})(f) \\ &= f \circ (\mathrm{id}_{P_m} \otimes c_{X_{n-1},X_n} \otimes \mathrm{id}_{X_{n-2}\otimes\cdots\otimes X_1}) \circ \cdots \\ &\circ (\mathrm{id}_{P_m} \otimes \mathrm{id}_{X_{n-1}\otimes\cdots\otimes X_3} \otimes c_{X_2,X_n} \otimes \mathrm{id}_{X_1}) \circ (\mathrm{id}_{P_m} \otimes \mathrm{id}_{X_{n-1}\otimes\cdots\otimes X_2} \otimes c_{X_1,X_n}) \\ &= f \circ (\mathrm{id}_{P_m} \otimes c_{X_{n-1}\otimes\cdots\otimes X_1,X_n}) \end{aligned}$$

and therefore the fourth assertion of this lemma yields

$$\begin{aligned} \mathcal{Z}((\mathfrak{b}_{1,2}\mathfrak{b}_{2,3}\cdots\mathfrak{b}_{n-1,n})^n)(f) \\ &= f \circ (\mathrm{id}_{P_m} \otimes c_{X_{n-1}} \otimes \cdots \otimes X_1, X_n) \circ (\mathrm{id}_{P_m} \otimes c_{X_{n-2}} \otimes \cdots \otimes X_1 \otimes X_n, X_{n-1}) \\ &\circ \cdots \circ (\mathrm{id}_{P_m} \otimes c_{X_n} \otimes \cdots \otimes X_2, X_1) \\ &= f \circ (\mathrm{id}_{P_m} \otimes \theta_{X_n} \otimes \cdots \otimes X_1) \circ (\mathrm{id}_{P_m} \otimes \theta_{X_n}^{-1} \otimes \cdots \otimes \theta_{X_1}^{-1}) \end{aligned}$$

As in the case of the first relation, the cochain map  $f \mapsto f \circ (\mathrm{id}_{P_m} \otimes \theta_{X_n \otimes \cdots \otimes X_1})$  is equal to the cochain map  $f \mapsto f \circ (\theta_{P_m} \otimes \mathrm{id}_{X_n \otimes \cdots \otimes X_1})$  by the first assertion in Lemma 3.9,
which is homotopic to the identity by Lemma 3.2. Therefore, our cochain map is homotopic to the cochain map

$$\mathcal{Z}(\mathfrak{d}_1^{-1}\mathfrak{d}_2^{-1}\cdots\mathfrak{d}_n^{-1})(f)=f\circ(\mathrm{id}_{P_m}\otimes\theta_{X_n}^{-1}\otimes\cdots\otimes\theta_{X_1}^{-1})$$

as required.

## 3.4 Hochschild cohomology

In order to explain why Corollary 3.7 generalizes the main result of our previous article (cf. [LMSS1]), we now specialize the situation to the case where C is the category of left modules over the factorizable ribbon Hopf algebra A described in Paragraph 2.2, and furthermore to the case where g = 1 and n = 0. From Paragraph 1.12, we know that  $\Gamma_{1,0} \cong SL(2,\mathbb{Z})$  and  $\Gamma_{1,1} \cong B_3$ . From Corollary 3.7, we then obtain a projective action of  $\Gamma_{1,0} \cong SL(2,\mathbb{Z})$  on  $\operatorname{Ext}_A^m(K,L)$ , which arises as a quotient of the projective action of  $\Gamma_{1,1} \cong B_3$  on  $\operatorname{Hom}_A(P_m,L)$  for a projective resolution

$$K \longleftarrow P_0 \longleftarrow P_1 \longleftarrow P_2 \longleftarrow \cdots$$

of the base field K, considered as a trivial A-module, as explained in Paragraph 2.2. That the projective action of  $\Gamma_{1,1}$  indeed descends to a projective action of  $\Gamma_{1,0}$ on the cohomology groups can be seen quite explicitly in this case: According to Paragraph 2.6, the generators  $\mathfrak{s}_1$  and  $\mathfrak{t}_1$  of  $\Gamma_{1,1}$  act by postcomposition with  $\mathfrak{S}$ and  $\mathfrak{T}$ , respectively, and we have seen in Paragraph 2.8 that they satisfy the identity  $\mathfrak{S} \circ \mathfrak{T} \circ \mathfrak{S} = \rho(v) \mathfrak{T}^{-1} \circ \mathfrak{S} \circ \mathfrak{T}^{-1}$ , which means that they satisfy the defining identity of  $\Gamma_{1,1} \cong B_3$  projectively. We have also seen there that the 2-chain relation is satisfied projectively, i.e., that the actions of  $\mathfrak{s}_1^4$  and  $\mathfrak{d}_1^{-1}$  agree up to a scalar. It therefore follows from Lemma 3.2 that  $\mathfrak{s}_1^4$  acts on the cohomology groups as a scalar multiple of the identity, so that the defining relations of the modular group are satisfied projectively.

In order to relate this result to Hochschild cohomology, we choose our projective resolution of the base field in a special way. We first choose a projective resolution

$$A \longleftarrow Q_0 \longleftarrow Q_1 \longleftarrow Q_2 \longleftarrow \cdots$$

of the algebra A in the category of left  $A \otimes A^{\text{op}}$ -modules, or equivalently the category of A-bimodules, where we require, as in [CE, Chap. IX, § 3, p. 167], that the left and the right action of A on a bimodule become equal when restricted to K. By [CE, Chap. X, Thm. 2.1, p. 185], we can then set  $P_m := Q_m \otimes_A K$  to obtain a projective resolution of  $A \otimes_A K \cong K$ .

Now the left A-module L can be considered as an A-bimodule via the trivial right A-action, i.e., the action  $\varphi.a := \varepsilon(a)\varphi$  for  $\varphi \in L = A^*$ . We denote L by  $L_{\varepsilon}$  if it is considered as an A-bimodule in this way. With the help of the cochain map

 $\operatorname{Hom}_{A\otimes A^{\operatorname{op}}}(Q_m, L_{\varepsilon}) \to \operatorname{Hom}_A(Q_m \otimes_A K, L), \ f \mapsto (q \otimes \lambda \mapsto \lambda f(q))$ 

with inverse

$$\operatorname{Hom}_A(Q_m \otimes_A K, L) \to \operatorname{Hom}_{A \otimes A^{\operatorname{op}}}(Q_m, L_{\varepsilon}), \ g \mapsto (q \mapsto g(q \otimes 1_K))$$

we see that the cochain complexes  $\operatorname{Hom}_A(Q_m \otimes_A K, L)$  and  $\operatorname{Hom}_{A \otimes A^{\operatorname{op}}}(Q_m, L_{\varepsilon})$  are isomorphic, so that also their cohomology groups  $\operatorname{Ext}_A^m(K, L)$  and  $\operatorname{Ext}_{A \otimes A^{\operatorname{op}}}^m(A, L_{\varepsilon})$ are isomorphic. But the latter cohomology groups are, by definition, the Hochschild cohomology groups  $HH^m(A, L_{\varepsilon})$ .

Via this isomorphism of cochain complexes, we can transfer the action of  $\Gamma_{1,1}$  to the Hochschild cochain groups of the bimodule  $L_{\varepsilon}$ . The generators  $\mathfrak{s}_1$  and  $\mathfrak{t}_1$  of  $\Gamma_{1,1}$  then clearly still act by postcomposition with the bimodule homomorphisms  $\mathfrak{S}$  and  $\mathfrak{T}$ , while the Dehn twist  $\mathfrak{d}_1$  acts by precomposition with the endomorphism  $q \mapsto v.q$ of  $Q_m$ , or equivalently by postcomposition with the endomorphism  $\varphi \mapsto v.\varphi$  of  $L_{\varepsilon}$ . From this correspondence, we see that our Lemma 3.2 can be considered as an analogue of Proposition 1.2 in our previous treatment [LMSS1].

While the two results are analogous, it is not yet clear that the projective representations of the modular group constructed here and in [LMSS1, Cor. 5.6, p. 419] are isomorphic; in fact, the action constructed above is on the space  $HH^m(A, L_{\varepsilon})$ , while the action constructed in [LMSS1] is on the space  $HH^m(A, A)$ . However, it turns out that the two actions are indeed isomorphic:

**Proposition 3.11.** The projective actions of  $SL(2,\mathbb{Z})$  on the spaces  $HH^m(A, L_{\varepsilon})$ and  $HH^m(A, A)$  are isomorphic.

**Proof.** (1) In general, for an A-bimodule N, we can modify the bimodule structure by defining the new left action as

$$a.n := n.S^{-1}(a)$$

and similarly the new right action as  $n.a := S^{-1}(a).n$ . We denote N by N' if we consider it endowed with this bimodule structure. From the point of view of bimodules as  $A \otimes A^{\text{op}}$ -modules, this operation is the pullback along the ring homomorphism

$$A \otimes A^{\mathrm{op}} \to A \otimes A^{\mathrm{op}}, \ a \otimes b \mapsto S^{-1}(b) \otimes S^{-1}(a).$$

(2) Now

$$A' \longleftarrow Q'_0 \longleftarrow Q'_1 \longleftarrow Q'_2 \longleftarrow \cdots$$

is a projective resolution of A', and the cochain complex  $\operatorname{Hom}_{A\otimes A^{\operatorname{op}}}(Q'_m, N')$  is not only isomorphic to the cochain complex  $\operatorname{Hom}_{A\otimes A^{\operatorname{op}}}(Q_m, N)$ , but even set-theoretically equal. We therefore have that  $\operatorname{Ext}^m_{A\otimes A^{\operatorname{op}}}(A', N') = \operatorname{Ext}^m_{A\otimes A^{\operatorname{op}}}(A, N)$ .

(3) The antipode yields a bimodule isomorphism  $S: A' \to A$ , which by the comparison theorem lifts to a chain map

By pulling back along the chain map  $(S_m)$ , we obtain an isomorphism

$$\operatorname{Ext}_{A\otimes A^{\operatorname{op}}}^{m}(A, N') \to \operatorname{Ext}_{A\otimes A^{\operatorname{op}}}^{m}(A', N') = \operatorname{Ext}_{A\otimes A^{\operatorname{op}}}^{m}(A, N)$$

In other words, we obtain an isomorphism between  $HH^m(A, N')$  and  $HH^m(A, N)$ .

(4) We know from [LMSS1, Lem. 5.2, p. 417] and the subsequent discussion that the projective action of  $SL(2,\mathbb{Z})$  on  $HH^m(A, A)$  is isomorphic to the one on the cohomology groups  $HH^m(A, \varepsilon A_{ad})$ , where the generators  $\mathfrak{s}$  and  $\mathfrak{t}$  act by postcomposition with  $\hat{\mathfrak{S}}$  and  $\hat{\mathfrak{T}}$ , respectively. Here,  $\varepsilon A_{ad}$  is the A-bimodule with A as underlying vector space, trivial left action  $a.a' := \varepsilon(a)a'$ , and right action  $a(a \otimes a') = S(a'_{(1)})aa'_{(2)}$  (cf. [LMSS1, Sec. 2, p. 406]). We note that the preceding discussion shows that  $\varepsilon A_{ad}$  itself is not an SL(2, Z)-module, but only a  $\Gamma_{1,1}$ -module; however, the induced projective  $\Gamma_{1,1}$ -action on  $HH^m(A, \varepsilon A_{ad})$  descends to a projective SL(2, Z)-action.

If we set  $N = {}_{\varepsilon}A_{ad}$ , we find that N' is the bimodule with underlying vector space A, left action given by  $a.a' = a_{(2)}a'S^{-1}(a_{(1)})$ , and trivial right action. From the way how we constructed the isomorphism between  $HH^m(A, N')$  and  $HH^m(A, N)$  in the preceding step, we see that it becomes equivariant if we let the generators  $\mathfrak{s}$  and  $\mathfrak{t}$  also act on  $HH^m(A, N')$  by postcomposition with  $\hat{\mathfrak{S}}$  and  $\hat{\mathfrak{T}}$ , respectively. But now Lemma 2.4 shows that  $\bar{\iota} \colon N' \to L_{\varepsilon}$  is a bimodule isomorphism that is also equivariant under the projective  $\Gamma_{1,1}$ -action. Therefore postcomposition with  $\bar{\iota}$  yields an isomorphism between  $HH^m(A, N')$  and  $HH^m(A, L_{\varepsilon})$  that is  $SL(2, \mathbb{Z})$ -equivariant. Combining this with our isomorphism between  $HH^m(A, \varepsilon A_{ad})$  and  $HH^m(A, N')$  already obtained, our assertion follows. In view of this proposition, we can indeed say that the mapping class group actions obtained here in Corollary 3.7 generalize the projective  $SL(2,\mathbb{Z})$ -representation on the Hochschild cohomology groups  $HH^m(A, A)$  obtained in [LMSS1, Cor. 5.6, p. 419].

## References

- [AS] L. V. Ahlfors, L. Sario, *Riemann surfaces*, Princeton Math. Ser., Vol. 26, Princeton Univ. Press, Princeton, 1960
- [A] R. F. Arens, Topologies for homeomorphism groups, Am. J. Math. 68 (1946), 593–610
- [BKLT] Y. Bespalov, T. Kerler, V. Lyubashenko, V. Turaev, Integrals for braided Hopf algebras, J. Pure Appl. Algebra 148 (2000), 113–164
- [B] A. Bruguières, Catégories prémodulaires, modularisations et invariants des variétés de dimension 3, Math. Ann. 316 (2000), 215–236
- [CE] H. Cartan, S. Eilenberg, *Homological algebra*, Princeton Math. Ser., Vol. 19, Princeton Univ. Press, Princeton, 1956
- [D] A. Dold, Lectures on algebraic topology, 2nd ed., Grundlehren Math. Wiss., Vol. 200, Springer, Berlin, 1980
- [DSS] C. L. Douglas, C. Schommer-Pries, N. Snyder, *The balanced tensor product* of module categories, arXiv:1406.4204
- [EGNO] P. Etingof, S. Gelaki, D. Nikshych, V. Ostrik, *Tensor categories*, Math. Surv. Monogr., Vol. 205, Am. Math. Soc., Providence, 2015
- [FM] B. Farb, D. Margalit, A primer on mapping class groups, Princeton Math. Ser., Vol. 49, Princeton Univ. Press, Princeton, 2012
- [FS] J. Fuchs, C. Schweigert, Coends in conformal field theory, in: K. Barron (ed.) et al., Lie algebras, vertex operator algebras, and related topics, Contemp. Math., Vol. 695, Am. Math. Soc., Providence, 2017, 65–81
- [FS2] J. Fuchs, C. Schweigert, Full logarithmic conformal field theory—an attempt at a status report, Fortschr. Phys. 67 (2019), 1910018

- [F] W. Fulton, Algebraic topology, Grad. Texts Math., Vol. 153, Springer, Berlin, 1995
- [HS] P. J. Hilton, U. Stammbach, A course in homological algebra, 2nd ed., Grad. Texts Math., Vol. 4, Springer, Berlin, 1997
- [H] M. W. Hirsch, *Differential topology*, Grad. Texts Math., Vol. 33, Springer, Berlin, 1991
- S. O. Ivanov, Nakayama functors and Eilenberg-Watts theorems, J. Math. Sci., New York 183 (2012), 675–680
- [J] N. Jacobson, Lectures in abstract algebra, III. Theory of fields and Galois theory, Grad. Texts Math., Vol. 32, Springer, Berlin, 1975
- [Ka] C. Kassel, Quantum groups, Grad. Texts Math., Vol. 155, Springer, Berlin, 1995
- [KT] C. Kassel, V. G. Turaev, Braid groups, Grad. Texts Math., Vol. 247, Springer, Berlin, 2008
- [KL] T. Kerler, V. Lyubashenko, Non-semisimple topological quantum field theories for 3-manifolds with corners, Lect. Notes Math., Vol. 1765, Springer, Berlin, 2001
- [Ko] M. Korkmaz, Low-dimensional homology groups of mapping class groups: a survey, Turk. J. Math. 26 (2002), 101–114
- [LMSS1] S. Lentner, S. N. Mierach, C. Schweigert, Y. Sommerhäuser, Hochschild cohomology and the modular group, J. Algebra 507 (2018), 400–420
- [LMSS2] S. Lentner, S. N. Mierach, C. Schweigert, Y. Sommerhäuser, *Hochschild cohomology, modular tensor categories, and mapping class groups, II. Examples,* in preparation
- [L1] V. Lyubashenko, Invariants of 3-manifolds and projective representations of mapping class groups via quantum groups at roots of unity, Commun. Math. Phys. 172 (1995), 467–516
- [L2] V. Lyubashenko, Ribbon abelian categories as modular categories, J. Knot Theory Ramifications 5 (1996), 311–403

- [ML1] S. Mac Lane, *Homology*, 3rd revised printing, Grundlehren Math. Wiss., Vol. 114, Springer, Berlin, 1975
- [ML2] S. Mac Lane, *Categories for the working mathematician*, 2nd ed., Grad. Texts Math., Vol. 5, Springer, Berlin, 1998
- [Maj] S. Majid, *Foundations of quantum group theory*, 2nd ed., Camb. Univ. Press, Cambridge, 2000
- [Mas] W. S. Massey, Algebraic topology: An introduction, Grad. Texts Math., Vol. 56, Springer, Berlin, 1977
- [Mi] B. Mitchell, *Theory of categories*, Pure Appl. Math., Vol. 17, Academic Press, New York, 1965
- [Mo] S. Montgomery, *Hopf algebras and their actions on rings*, 2nd revised printing, Reg. Conf. Ser. Math., Vol. 82, Am. Math. Soc., Providence, 1997
- [Mue] M. Müger, On the structure of modular categories, Proc. Lond. Math. Soc., III. Ser., 87 (2003), 291–308
- [P] B. Pareigis, Kategorien und Funktoren, Math. Leitfäden, Teubner, Stuttgart, 1969
- [PS] V. V. Prasolov, A. B. Sossinsky, Knots, links, braids, and 3-manifolds, Transl. Math. Monogr., Vol. 154, Am. Math. Soc., Providence, 1997
- [Q] B. v. Querenburg, *Mengentheoretische Topologie*, 2. Aufl., Hochschultext, Springer, Berlin, 1979
- [Ra] D. E. Radford, *Hopf algebras*, Ser. Knots Everything, Vol. 49, World Scientific, Singapore, 2012
- [Ro1] J. J. Rotman, An introduction to algebraic topology, Grad. Texts Math., Vol. 119, Springer, Berlin, 1988
- [Ro2] J. J. Rotman, An introduction to homological algebra, 2nd ed., Universitext, Springer, Berlin, 2009
- [ST] H. Seifert, W. Threlfall, *Lehrbuch der Topologie*, Teubner, Leipzig, 1934
- [Sh1] K. Shimizu, Non-degeneracy conditions for braided finite tensor categories, Adv. Math. 355 (2019), 106778

- [Sh2] K. Shimizu, Integrals for finite tensor categories, Algebr. Represent. Theory 22 (2019), 459–493
- [SZ] Y. Sommerhäuser, Y. Zhu, Hopf algebras and congruence subgroups, Mem. Am. Math. Soc., Vol. 219, No. 1028, Am. Math. Soc., Providence, 2012
- [S] J. Stillwell, Classical topology and combinatorial group theory, 2nd ed., Grad. Texts Math., Vol. 72, Springer, Berlin, 1993
- [TUY] A. Tsuchiya, K. Ueno, Y. Yamada, Conformal field theory on universal family of stable curves with gauge symmetry, in: M. Jimbo (ed.) et al., Integrable systems in quantum field theory and statistical mechanics, Adv. Stud. Pure Math., Vol. 19, Academic Press, Boston, 1989, 459–566
- [T] V. G. Turaev, Quantum invariants of knots and 3-manifolds, De Gruyter Stud. Math., Vol. 18, de Gruyter, Berlin, 1994
- [V] A. Virelizier, Kirby elements and quantum invariants, Proc. Lond. Math. Soc., III. Ser., 93 (2006), 474–514